SECURITIES AND EXCHANGE COMMISSION

FORM 6-K

Current report of foreign issuer pursuant to Rules 13a-16 and 15d-16 Amendments

Filing Date: **2021-10-04** | Period of Report: **2021-10-01** SEC Accession No. 0001193125-21-290038

(HTML Version on secdatabase.com)

FILER

Wallbox N.V.

CIK:1866501| IRS No.: 0000000000 | State of Incorp.:P7 | Fiscal Year End: 1231

Type: 6-K | Act: 34 | File No.: 001-40865 | Film No.: 211300714

SIC: 3790 Miscellaneous transportation equipment

Mailing Address CARRER DEL FOC, 68 BARCELONA U3 08038 Business Address CARRER DEL FOC, 68 BARCELONA U3 08038 34 93.220.43.08

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

SECURITIES AND EXCHANGE COMMINISSION Washington, D.C. 20549
FORM 6-K
REPORT OF FOREIGN PRIVATE ISSUER PURSUANT TO SECTION 13A-16 OR 15D-16 UNDER THE SECURITIES EXCHANGE ACT OF 1934
For the month of October 2021
Commission File Number: 001-40865
Wallbox N.V. (Exact Name of Registrant as Specified in Its Charter)
Carrer del Foc, 68 Barcelona, Spain 08038 Tel: +34 930 181 668 (Address of principal executive offices)
Indicate by check mark whether the registrant files or will file annual reports under cover of Form 20-F or Form 40-F.
Form 20-F ✓ Form 40-F □
Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(1): □
Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(7): □

CONTENTS

Business Combination

As previously announced, on June 9, 2021, Kensington Capital Acquisition Corp. II, a Delaware corporation ("Kensington"), Wallbox B.V., a private company with limited liability incorporated under the Laws of the Netherlands (besloten vennootschap met beperkte aansprakelijkheid and which was converted into a public limited liability company (naamloze vennootschap)) ("Holdco"), Orion Merger Sub Corp., a Delaware corporation and wholly owned subsidiary of Holdco ("Merger Sub"), and Wall Box Chargers, S.L., a Spanish limited liability company (sociedad limitada) (the "Company"), entered into a business combination agreement (the "Business Combination Agreement"), pursuant to which, among other things, Kensington and the Company agreed to enter into a business combination. Terms not otherwise defined herein have the meanings assigned to them in the Business Combination Agreement, a copy of which is filed as Exhibit 2.1 to Holdco's registration statement on Form F-4.

On October 1, 2021 (the "Closing Date"), Holdco consummated the previously announced business combination pursuant to the Business Combination Agreement.

On the Closing Date, the following transactions occurred pursuant to the terms of the Business Combination Agreement (collectively, the "Business Combination"):

Holdco (i) changed its legal form from a private company with limited liability (besloten vennootschap met beperkte aansprakelijkheid) to a public limited liability company (naamloze vennootschap) and (ii) entered into the Deed of Conversion containing the Articles of Association of Wallbox N.V., copies of which are filed as Exhibit 3.1 and 3.2 to this Report of Foreign Private Issuer on Form 6-K and are incorporated by reference herein;

Prior to the Closing Date, each holder of Company Convertible Notes converted its Company Convertible Notes into Company Ordinary Shares and on the Closing Date, each holder of Company Ordinary Shares contributed its Company Ordinary Shares to Holdco in exchange for Holdco Ordinary Shares and the Company became a wholly-owned subsidiary of Holdco;

Each issued Company Ordinary Share was contributed in kind to Holdco and Holdco issued, in aggregate, to holders of Company Ordinary Shares (including the Company Ordinary Shares issued upon the conversion of the Company Convertible Notes) 106,778,437 Holdco Ordinary A Shares and 23,250,793 Holdco Ordinary B Shares, as applicable, and each holder of Company Ordinary Shares (including the Company Ordinary Shares issued upon the conversion of the Company Convertible Notes) ceased to be the holder of such Company Ordinary Shares;

Merger Sub merged with and into Kensington (the "Merger") with Kensington as the surviving corporation, and Kensington became a wholly-owned subsidiary of Holdco; as a result of the Merger, all of the Kensington Common Stock of the Surviving Corporation (other than Excluded Shares) were converted into the right to receive New Kensington Common Stock;

Each Company Option that was outstanding immediately prior to the Exchange Effective Time, whether vested or unvested, was converted at the Exchange Effective Time into an option to purchase a number of shares of the applicable class of Holdco Ordinary Shares as determined pursuant to the Business Combination Agreement:

19,861,318 shares of Kensington Common Stock issued and outstanding immediately prior to the Merger Effective Time (other than the Excluded Shares) were converted into and became shares of New Kensington Common Stock, and each such share of New Kensington Common Stock was immediately thereafter be exchanged for Holdco Shares, and Holdco issued one Holdco Ordinary A Share for each share of New Kensington Common Stock exchanged;

14,683,333 Kensington Warrants that were outstanding immediately prior to the Merger Effective Time ceased to represent a right to acquire the number of shares of Kensington Class A Common Stock and were converted into a right to acquire one Holdco Ordinary A Share and thereupon were assumed by Holdco pursuant to Warrant Assignment, Assumption and Amended & Restated Warrant Agreement which is filed as Exhibit 4.1 to this Report of Foreign Private Issuer on Form 6-K and is incorporated by reference herein; and

An aggregate of 11,100,000 Holdco Ordinary A Shares (inclusive of the 1,100,000 shares described under "PIPE Investment" below) were issued for a purchase price of \$10.00 per share and an aggregate purchase price of \$111 million, in a private placement pursuant to the subscription agreements entered into in connection with the Business Combination.

The Wallbox Class A ordinary shares will begin trading on the New York Stock Exchange on October 4, 2021 under the symbol "WBX."

On the Closing Date, Kensington issued a press release announcing the closing of the Business Combination. A copy of the press release is filed as Exhibit 99.1 to this Report of Foreign Private Issuer on Form 6-K and is incorporated by reference herein.

PIPE Investment

Effective as of September 29, 2021, Holdco and Kensington entered into separate subscription agreements (each, a "Subscription Agreement") with a number of investors (each a "Subscriber"), pursuant to which the Subscribers agreed to purchase, and Holdco agreed to issue to the Subscribers, an aggregate of 1,100,000 Holdco Ordinary A Shares (the "PIPE Shares"), for a purchase price of \$10.00 per share and an aggregate purchase price of \$11.0 million, in a private placement (the "New PIPE"). Certain officers and directors of Kensington and their affiliates have subscribed for approximately 600,000 PIPE Shares pursuant to the Subscription Agreements.

Pursuant to the Subscription Agreements, Holdco agreed, among other things, that, within 30 calendar days after the consummation of the Transactions (the "Filing Deadline"), Holdco will file with the SEC (at Holdco's sole cost and expense) a registration statement registering the resale of the PIPE Shares (the "Resale Registration Statement"), and Holdco will use its commercially reasonable efforts to have the Resale Registration Statement declared effective as soon as practicable after the filing thereof.

The foregoing description of the Subscription Agreements is qualified in its entirety by reference to the full text of the form of the Subscription Agreement, a copy of which is filed as Exhibit 10.2 to this Report of Foreign Private Issuer on Form 6-K and is incorporated herein by reference.

EXHIBIT INDEX

Exhibit No.	Description
3.1	Articles of Association of Wallbox N.V.
3.2	Deed of Conversion to Wallbox N.V., dated as of October 1, 2021.
4.1	Warrant Assignment, Assumption and Amended & Restated Warrant Agreement, dated October 1, 2021.
10.1	Registration Rights Agreement and Lock-up Agreement Agreement, dated October 1, 2021.
10.2	Form of Subscription Agreement dated September 29, 2021.
10.3	Wallbox N.V. 2021 Equity Incentive Plan.
10.4	Wallbox N.V. 2021 Employee Stock Purchase Plan.
99.1	Press release, dated October 1, 2021.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Wallbox N.V.

Date: October 1, 2021 By: /s/ Enric Asunción Escorsa

Enric Asunción Escorsa Chief Executive Officer

ARTICLES OF ASSOCIATION

CHAPTER I - DEFINITIONS AND INTERPRETATION

1 Definitions

1.1 In these articles of association the following words shall have the following meanings:

Annual Accounts: the Company's annual accounts as referred to in Section 2:361 DCC;

Auditor: an auditor as referred to in Section 2:393 DCC, or an organization in which such auditors work together;

Board: the Company's board of directors;

Board Regulations: the written rules and regulations adopted by the Board as referred to in article 21.4;

Chair & CEO: the Executive Director designated by the Board as chair and chief executive officer (CEO);

Chairman of the Board: the Non-Executive Director who serves as chairman of the Board as referred to under Dutch law;

Class A Share: an ordinary class A share in the Company's share capital;

Class B Share: an ordinary class B share in the Company's share capital;

Company: the Company to which these articles of association pertain;

Closing Date: the date on which the Company issues Class B Shares for the first time;

Conversion Date: the date that the Non-Executive Directors determine, in their sole discretion, that a Conversion Event has occurred, provided that the Conversion Date shall be no later than the date of such determination;

Conversion Event means the occurrence of:

- (a) any Transfer of a Class B Share that is not a Permitted Transfer;
- (b) the Final Conversion Event;
- (c) with respect to a Class B Share held by an Excluded Holder or such Excluded Holder's Permitted Entity, the death or Disability of such Excluded Holder; or
- (d) with respect to a Class B Share held by a Permitted Entity or Permitted Transferee, an event as a consequence of which such Permitted Entity or Permitted Transferee (as applicable);

Conversion Request shall have the meaning given thereto in article 6A;

Conversion Share: an ordinary conversion share in the Company's share capital;

Conversion Share Distribution: a distribution on each Conversion Share for an amount equal to one percent (1%) of the nominal value of Conversion Shares:

DCC: the Dutch Civil Code;

Director: an Executive Director or Non-Executive Director;

Disability or **Disabled**: the permanent and total disability such that an Excluded Holder is unable to engage in any substantial gainful activity by reason of any medically determinable mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than twelve (12) months as determined by a licensed medical practitioner. In the event of a dispute whether an Excluded Holder has suffered a Disability, no Disability of the Excluded Holder shall be deemed to have occurred unless and until an affirmative ruling regarding such Disability has been made by a court of competent jurisdiction, and such ruling has become final and non-appealable, whereby the start date of the Disability will be either the date of determination of the Disability by a licensed medical practitioner or the date the ruling by a court of competent jurisdiction has become final and non-appealable;

Excluded Holder: each of Enric Asunción Escorsa, born on the third day of April nineteen hundred eighty-five, and Eduard Castañeda, born on the twelfth day of August nineteen hundred eighty-five;

Executive Director: a member of the Board appointed as executive director;

Final Conversion Event: the conversion event occurring at the earliest of:

- (a) the date set by the Board that is no less than sixty-one (61) days and no more than one hundred eighty (180) days after the date on which the aggregate number of issued and outstanding Class B Shares held (jointly) by the Excluded Holders and their Permitted Entities represents less than twenty percent (20%) of the aggregate number of issued and outstanding Class B Shares held by the Initial Holders on the Closing Date; or
- (b) the date set by the meeting of holders of Class B Shares;

General Meeting: the corporate body of the Company consisting of Shareholders and all other Persons with Meeting Right or a meeting of Shareholders and other Persons with Meeting Right, as the case may be;

Group Company: a group company of the Company as referred to in Section 2:24b DCC;

in writing: by letter, by telecopier, by e-mail, or by a legible and reproducible message otherwise electronically sent, provided that the identity of the sender can be sufficiently established;

Initial Holder means, in relation to any Class B Share, the person or entity holding such Class B Share on the Closing Date;

Management Report: the Company's management report as referred to in Section 2:391 DCC;

Meeting Right: the right, either in person or by proxy authorized in writing, to attend and address the General Meeting;

Non-Executive Director: a member of the Board appointed as non-executive director;

Permitted Entity: means any entity (including a corporation, (limited liability) company, partnership, foundation and trust) so long as an Excluded Holder has exclusive Voting Control with respect to the Class B Shares held by such entity;

Permitted Transfer means:

(a) a Transfer between (i) Initial Holders, (ii) an Initial Holder and an Excluded Holder, and (iii) an Initial Holder and a Permitted Entity;

- (b) a Transfer between (i) Excluded Holders, and (ii) an Excluded Holder and a Permitted Entity;
- (c) a Transfer between Permitted Entities;

Permitted Transferee: a transferee of Class B Shares, or rights or interests therein, received in a Transfer that constitutes a Permitted Transfer;

Persons with Meeting Right: Shareholders, holders of a usufruct with Meeting Right and holders of a right of pledge with Meeting Right;

Persons with Voting Rights: Shareholders with voting rights, holders of a usufruct with voting rights and holders of a right of pledge with voting rights in the General Meeting;

Record Date: the twenty-eighth day prior to the date of a General Meeting, or such other day as prescribed by law;

Share: a share in the Company's share capital, unless the contrary is expressed this shall include each Class A Share, each Class B Share and each Conversion Share;

Shareholder: a holder of one or more Shares;

Subsidiary: a subsidiary of the Company as referred to in Section 2:24a DCC;

Transfer: any sale, assignment, transfer under general or specific title, conveyance, grant of any form of security interest (other than as explicitly provided in this definition), or other transfer or disposition of a Class B Share or any legal or beneficial interest in such Class B Share, whether or not for value and whether voluntary or involuntary or by operation of law, whether directly or indirectly, including by merger, consolidation or otherwise. A "Transfer" shall also include, without limitation, the transfer of, or entering into a binding agreement with respect to, Voting Control over a Class B Share by proxy or otherwise; provided, however, that the following shall not be considered a "Transfer":

- (a) the granting of a power of attorney (i) in connection with actions to be taken at a specific General Meeting, and/or (ii) pursuant to article 6A.1 (c), as well as the exercise of any such power of attorney by the respective attorney;
- (b) the creation of a right of pledge on Class B Shares that creates a mere security interest in such Class B Shares pursuant to a bona fide loan or indebtedness transaction so long as the holder of Class B Shares continues to exercise Voting Control over such pledged Class B Shares, provided, however, that an enforcement of a right of pledge on such Class B Shares or other similar action by the pledgee shall constitute a "Transfer" of a Class B Share;
- (c) the entering into of a voting trust, agreement or arrangement (with or without granting a proxy), or consummating the actions or transactions contemplated therein, between Initial Holders, Excluded Holders and/or Permitted Entities, that (i) is disclosed either in an applicable schedule filed with the United States securities and exchange commission or in writing to the Company, (ii) either has a term not exceeding one year or is terminable by the holder of Class B Shares at any time and (iii) does not involve any payment of cash, securities, property or other consideration to the holders of Class B Shares other than the mutual promise to vote shares in a designated manner;

(d) the entering into of any agreement with respect to supporting or voting in favor of or tendering Shares with respect to any transaction proposed to be supported by the Company, or supporting the actions or transactions contemplated therein (including, without limitation, tendering Class B Shares or voting such shares in favor of such transaction or in opposition to other proposals that may be expected to delay or impair the ability to support such transaction), if the entry into such support or voting agreement is approved by the Board;

Transferor shall have the meaning given thereto in article 6B; and

Voting Control means with respect to a Class B Share, the power to vote or direct the voting of such Class B Share, including by proxy, voting agreement or otherwise.

2 Construction

- 2.1 References to articles shall be deemed to refer to articles of these articles of association, unless the contrary is apparent.
- 2.2 Any reference to a gender includes all genders.

CHAPTER II - NAME, CORPORATE SEAT AND OBJECTS

3 Name and corporate seat

- 3.1 The Company's name is Wallbox N.V.
- 3.2 The corporate seat of the Company is in Amsterdam, the Netherlands.

4 Objects

The objects of the Company are:

- (a) to design, manufacture and distribute charging solutions for residential, business, and public use, and to provide installation services;
- (b) to incorporate, to participate in any way whatsoever in, to manage, to supervise businesses and companies;
- (c) to finance businesses and companies;
- (d) to borrow, to lend and to raise funds, including the issue of bonds, promissory notes or other securities or evidence of indebtedness as well as to enter into agreements in connection with aforementioned activities;
- (e) to render advice and services to businesses and companies with which the Company forms a group and to third parties;
- (f) to grant guarantees, to bind the Company and to pledge its assets for obligations of the Company, its group companies and/or third parties;
- (g) to acquire, alienate, manage and exploit registered property and items of property in general;
- (h) to trade in currencies, securities and items of property in general;
- (i) to develop and trade in patents, trade marks, licenses, know-how and other intellectual and industrial property rights; and
- (j) to perform any and all activities of an industrial, financial or commercial nature,

and to do all that is connected therewith or may be conducive thereto, all to be interpreted in the broadest sense.

CHAPTER III - SHARE CAPITAL

5 Authorized capital, share premium reserve

- 5.1 The authorized capital of the Company equals one hundred eight million two euro and sixteen eurocent (EUR 108,000,002.16).
- 5.2 The authorized capital of the Company is divided into four hundred million (400,000,000) Class A Shares, with a nominal value of twelve eurocent (EUR 0.12) each, fifty million (50,000,000) Class B Shares, with a nominal value of one euro and twenty eurocent (EUR 1.20) each, and two (2) Conversion Shares, with a nominal value of one euro and eight eurocent (EUR 1.08) each.
- 5.3 Upon the conversion of one or more Class B Shares into Class A Shares and Conversion Shares referred to in article 6, the authorized capital shall decrease with the number of Class B Shares so converted and shall increase with the number of Class A Shares and Conversion Shares into which such Class B Shares were converted.
- 5.4 Within eight days after a conversion of one or more Class B Shares into Class A Shares and Conversion Shares referred to in article 6, the Board shall (i) file a notification thereof with the Dutch trade register, which notification must at least include the authorized capital following the conversion, and (ii) register the conversion in the register of Shareholders as referred to in article 13.
- 5.5 No share certificates shall be issued.
- 5.6 The Company shall maintain a general share premium reserve for the benefit of the Shareholders.

6 Conversion of Shares

6.1 Each Class B Share can be converted into one Class A Share and one Conversion Share, subject to the provisions of this article 6. Class A Shares and Conversion Shares cannot be converted into other classes of Shares.

6A Voluntary conversion of Class B Shares

- 6A.1 Each holder of one or more Class B Shares may request the conversion of all or part of his Class B Shares into Class A Shares and Conversion Shares in the ratio set out in article 6.1 by means of a written request addressed to the Board (Conversion Request). The Conversion Request must:
 - (a) indicate the number of Class B Shares to which the Conversion Request pertains;
 - (b) include a representation by the requesting Shareholder that no depositary receipts have been issued with respect to the Class B Shares to which the Conversion Request pertains nor that these Class B Shares are encumbered with any usufruct, right of pledge, attachment or other encumbrance;
 - (c) provide for an irrevocable and unconditional power of attorney from the requesting Shareholder to the Company, with full power of substitution and governed by Dutch law, to perform the acts described in article 6A.2 on behalf of such Shareholder.

6A.2 Subject to article 6A.3:

- (a) upon receipt of a Conversion Request satisfactory to the Board, the Board shall, as soon as reasonably possible, resolve to convert the number of Class B Shares to which the Conversion Request pertains into Class A Shares and Conversion Shares in the ratio set out in article 6.1; and
- (b) promptly following such conversion, the requesting Shareholder shall be obliged to offer and transfer the Conversion Shares to the Company for no consideration and the Company shall accept such Conversion Shares.
- 6A.3 To the extent the Company would not be permitted under applicable law to acquire Conversion Shares in accordance with article 6A.2 paragraph (b), the Board shall, as soon as reasonably possible, convene a General Meeting in accordance with article 35, at which meeting it shall be proposed to the General Meeting to cancel such number of Shares (held in treasury) to allow again for the acquisition of Conversion Shares as referred to in article 6A.2 paragraph (b).

6B Mandatory conversion of Class B Shares

- 6B.1 A Class B Share shall automatically convert into Class A Shares and Conversion Shares in the ratio set out in article 6.1 upon the occurrence of a Conversion Event in respect of such Class B Share, subject to the provisions of this article 6B and with effect as of the Conversion Date. Upon the occurrence of a Conversion Event, the Shareholder concerned shall be obliged to notify the Board thereof by means of a written notice addressed to the Board.
- 6B.2 If at any time a Conversion Share is held by anyone other than the Company, such holder of Conversion Shares (**Transferor**) shall be obliged to offer and transfer such Conversion Shares to the Company unencumbered (without any usufruct, right of pledge, attachment or other encumbrance and without depositary receipts issued for such Conversion Shares) and for no consideration. If and for as long as the Transferor fails to offer and transfer the relevant Conversion Shares to the Company, the voting rights, Meeting Right and rights to receive distributions attached to the relevant Conversion Shares are suspended.
- 6B.3 If the Transferor fails to offer and transfer the relevant Conversion Shares to the Company within:
 - (a) sixty (60) days after the Conversion Date for Conversion Shares acquired as a result of a Conversion Event (c); or
 - (b) thirty (30) days after the Conversion Date for Conversion Shares acquired as a result of a Conversion Event other than (c),

the Company is irrevocably empowered and authorized to offer and transfer the relevant Conversion Shares to the Company until such transaction occurs.

6B.4 The Board may, from time to time, establish such policies and procedures relating to the general administration of the share capital structure as it may deem necessary or advisable, and may request that holders of Class B Shares furnish affidavits or other proof to the Board as it deems necessary to verify the legal and beneficial ownership of Class B Shares, and to confirm that a Conversion Event has not occurred.

6B.5 To the extent the Company would not be permitted under applicable law to acquire Conversion Shares in accordance with article 6B.2, the Board shall convene a General Meeting in accordance with article 35, at which meeting it shall be proposed to the General Meeting to cancel such number of Shares (held in treasury) to allow again for the acquisition of Conversion Shares as referred to in article 6B.2.

7 Issuance of Shares

- 7.1 Shares shall be issued pursuant to a resolution of the Board if the Board has been designated thereto by the General Meeting for a specific period and with due observance of applicable statutory provisions. Such designation by the General Meeting must state the number of Shares that may be issued.
 - The designation may be extended by specific consecutive periods with due observance of applicable statutory provisions. Unless otherwise stipulated at its grant, the designation may not be withdrawn.
- 7.2 If and insofar as the Board is not designated by the General Meeting, Shares shall be issued pursuant to a resolution of the General Meeting. The General Meeting shall, in addition to the Board, remain authorized to issue Shares if such is specifically stipulated in the resolution authorizing the Board to issue Shares as described in article 7.1.
- 7.3 The articles 7.1 and 7.2 shall apply by analogy to the granting of rights to subscribe for Shares, but shall not apply to an issue of Shares to a person exercising a previously granted right to subscribe for Shares.
- 7.4 A resolution of the General Meeting as referred to in this article 7 can only be adopted at the proposal of the Board.
- 7.5 If the resolution of the General Meeting to issue Shares or to designate the authority to issue Shares to the Board as referred to in article 7.1 is detrimental to the rights of holders of a specific class of Shares, the validity of such resolution of the General Meeting requires a prior or simultaneous approval by the group of holders of such class of Shares.

8 Pre-emptive rights

- 8.1 Each Shareholder shall have a pre-emptive right on any issuance of Class A Shares and Class B Shares in proportion to the aggregate amount of its Class A Shares and Class B Shares. No pre-emptive rights shall apply in respect of any issuance of Conversion Shares.
- 8.2 This pre-emptive right on any issuance of Class A Shares and Class B Shares does not apply to:
 - (a) Shares issued to employees of the Company or a Group Company;
 - (b) Shares that are issued against payment other than in cash; and
 - (c) Shares issued to a person exercising a previously granted right to subscribe for Shares.
- 8.3 Pre-emptive rights may be limited or excluded by a resolution of the General Meeting. Pre-emptive rights may also be limited or excluded by a resolution of the Board if the Board has been designated thereto by the General Meeting for a specific period and with due observance of applicable statutory provisions, and the Board has also been designated to issue Shares in accordance with article 7.1. The designation may be extended by specific consecutive periods with due observance of applicable statutory provisions. Unless otherwise stipulated at its grant, the designation may not be withdrawn.

- 8.4 A resolution of the General Meeting to limit or exclude pre-emptive rights and a resolution to designate the Board thereto, can only be adopted at the proposal of the Board and shall require a majority of at least two-thirds of the votes cast if less than half of the issued capital of the Company is represented at the General Meeting.
- 8.5 When adopting a resolution to issue Shares, the General Meeting or the Board shall determine how and during which period these pre-emptive rights may be exercised, subject to Section 2:96a DCC.
- 8.6 Article 8 shall apply by analogy to the granting of rights to subscribe for Shares.

9 Payment on Shares

- 9.1 Shares may only be issued against payment in full of the amount at which such Shares are issued and with due observance of the provisions of the Sections 2:80, 2:80a and 2:80b DCC.
- 9.2 Payment on Shares must be made in cash if no alternative contribution has been agreed. Payment other than in cash must be made in accordance with the provisions in Section 2:94b DCC. Payment in a currency other than euro may only be made with the consent of the Company and with due observance of the provisions of Section 2:93a DCC.
- 9.3 Shares issued to (i) current or former employees of the Company or a Group Company, (ii) current or former Directors under an equity compensation plan of the Company and (iii) holders of a right to subscribe for Shares granted in accordance with article 7.3 may be paid-up at the expense of the reserves of the Company, notwithstanding the provisions of article 34.
- 9.4 The Board may perform legal acts as referred to in Section 2:94 DCC without the prior approval of the General Meeting.

CHAPTER IV - OWN SHARES AND CAPITAL REDUCTION

10 Share repurchase and disposal of shares

- 10.1 The Company may repurchase fully paid-up Shares:
 - (a) for no consideration; or
 - (b) against consideration if and insofar as the Board has been authorized thereto by the General Meeting for a specific period and with due observance of applicable statutory provisions. The General Meeting shall determine in its authorization how many Shares the Company may repurchase, in what manner and at what price range.
- 10.2 The authorization by the General Meeting is not required if the Company repurchases fully paid-up Shares for the purpose of transferring these Shares to employees of the Company or a Group Company under any applicable equity compensation plan, provided that those Shares are quoted on an official list of a stock exchange.
- 10.3 Any disposal of Shares by the Company shall require a resolution of the Board. Such resolution shall also stipulate any conditions of the disposal.

11 Capital reduction

- 11.1 The General Meeting may only at the proposal of the Board resolve to reduce the Company's issued capital by (i) reducing the nominal value of Shares through amendment of these articles of association or (ii) cancelling Shares held by the Company itself.
- 11.2 A resolution of the General Meeting to reduce the Company's issued capital, shall require a majority of at least two-thirds of the votes cast if less than half of the issued capital of the Company is represented at the General Meeting.
- 11.3 If the resolution of the General Meeting to reduce the Company's issued capital by reducing the nominal value of Shares through amendment of these articles of association, as referred to above, is detrimental to the rights of holders of a specific class of Shares, the validity of such resolution of the General Meeting requires a prior or simultaneous approval by the group of holders of such class of Shares.

CHAPTER V - TRANSFER OF SHARES

12 Transfer of shares

- 12.1 The transfer of Shares shall require a deed executed for that purpose and, save in the event the Company itself is a party to such legal act, written acknowledgement by the Company of the transfer. Service of notice of such deed to the Company or of a true copy or extract of such deed will be the equivalent of such acknowledgement.
- 12.2 Article 12.1 shall apply *mutatis mutandis* to the creation of a right of pledge or usufruct on a Share, provided that a right of pledge may also be created without acknowledgement by or service of notice upon the Company, in which case Section 3:239 DCC applies and the acknowledgement by or service of notice upon the Company shall replace the announcement as referred to in Section 3:239(3) DCC.
- 12.3 If and as long as one or more Class A Shares are admitted to trading on the New York Stock Exchange, or if it may reasonably be expected that one or more Class A Shares shall shortly be admitted to trading on the New York Stock Exchange, the Board may resolve that the laws of the State of New York, United States of America, shall apply to the property law aspects of the Class A Shares, as a result of which the articles 12.1 and 12.2 shall not apply to the Class A Shares. Such resolution and the revocation thereof shall be made available for inspection on the Company's website and at the Dutch trade register.

The property law aspects of the Class A Shares (including the legal rules on ownership, legal title and transfer) in book-entry form, as included in the part of the register of shareholders kept by the relevant transfer agent, shall be governed by the laws of the State of New York, United States of America, in accordance with the applicable law on International Private laws as referred to in Title 10 of Book 10 DCC (*Boek 10 Internationaal privaatrecht*), especially Section 10:141 DCC.

CHAPTER VI - SHAREHOLDERS' REGISTER AND LIMITED RIGHTS TO SHARES

13 Shareholders' register

- 13.1 The Board must keep a shareholders' register; the Board may appoint a registrar to keep the register on its behalf. The register must be regularly updated. The shareholders' register may be kept in several copies and in several places. Part of the register may be kept outside the Netherlands to comply with applicable local law or pursuant to stock exchange rules.
- 13.2 Each Shareholder's name, address and further information as required by law or considered appropriate by the Board are recorded in the shareholders' register. Shareholders shall provide the Board with the necessary particulars in a timely fashion. Any consequences of not, or incorrectly, notifying such particulars will be the responsibility of the Shareholder concerned.
- 13.3 If a Shareholder so requests, the Board shall provide the Shareholder, free of charge, with written evidence of the information in the shareholders' register concerning the Shares registered in the Shareholder's name.
- 13.4 The articles 13.2 and 13.3 shall apply by analogy to holders of a usufruct or right of pledge on one or more Shares, with the exception of a holder of a right of pledge created without acknowledgement by or service of notice upon the Company.

14 Right of pledge on Shares

- 14.1 Class A Shares and Class B Shares can be pledged. Conversion Shares cannot be pledged.
- 14.2 Subject to article 12.3 (if applicable), if a Share is encumbered with a right of pledge, the voting rights attached to that Share shall vest in the Shareholder, unless at the creation of the right of pledge the voting rights were granted to the pledgee. A pledgee with voting rights shall have Meeting Right.
- 14.3 A Shareholder who as a result of a right of pledge does not have voting rights, shall have Meeting Right. A pledgee without voting rights shall not have Meeting Right.

15 Usufruct on Shares

- 15.1 Subject to article 12.3 (if applicable), if a usufruct is created on a Share, the voting rights attached to that Share shall vest in the Shareholder, unless at the creation of the usufruct the voting rights were granted to the usufructuary. A usufructuary with voting rights shall have Meeting Right.
- 15.2 A Shareholder who as a result of a usufruct does not have voting rights, shall have Meeting Right. A usufructuary without voting rights shall not have Meeting Right.

16 Depositary receipts

The Company shall not cooperate with the issuance of depositary receipts for Shares.

CHAPTER VII - MANAGEMENT AND SUPERVISION

17 Board: composition, appointment, suspension and dismissal

17.1 The Company shall be managed by the Board. The Board shall consist of one or more Executive Directors and one or more Non-Executive Directors. The number of Executive Directors and the number of Non-Executive Directors shall be determined by the Board. Only individuals may be Directors.

- 17.2 The Executive Directors and Non-Executive Directors shall be appointed as such by the General Meeting at the nomination of the Board. A nomination by the Board shall state whether a person is nominated for appointment as Executive Director or Non-Executive Director.
- 17.3 A Director shall be appointed for a term of approximately one year, which term of office shall lapse immediately after the close of the annual General Meeting held in the year after his appointment. A Director may be reappointed with due observance of the preceding sentence. At the proposal of the Board, the General Meeting may resolve to deviate from the term of office of approximately one year. A Non-Executive Director may be in office for a period not exceeding twelve (12) years, which period may or may not be interrupted, unless at the proposal of the Board the General Meeting resolves otherwise.
- 17.4 The General Meeting may at all times suspend or dismiss any Director. A resolution of the General Meeting to suspend or dismiss a Director other than at the proposal of the Board requires a two-thirds majority of the votes cast, representing more than one half of the issued capital of the Company. The Board may at all times suspend an Executive Director.
- 17.5 A suspension may be extended one or more times but may not last longer than three (3) months in aggregate. If at the end of that period, no decision has been taken on termination of the suspension or on dismissal, the suspension shall end. A suspension can be terminated by the General Meeting at any time.

18 Board: vacancy or inability

- 18.1 If the seat of an Executive Director is vacant or upon the inability of an Executive Director, the remaining Executive Directors shall temporarily be entrusted with the executive management of the Company, provided that the Board may provide for a temporary replacement. If the seats of all Executive Directors are vacant or upon the inability of all Executive Directors, the executive management of the Company shall temporarily be entrusted to the Non-Executive Directors, provided that the Board may provide for one or more temporary replacements.
- 18.2 If the seat of a Non-Executive Director is vacant or upon inability of a Non-Executive Director, the remaining Non-Executive Directors shall temporarily be entrusted with the performance of the duties and the exercise of the authorities of that Non-Executive Director, provided that the Board may provide for a temporary replacement. If the seats of all Non-Executive Directors are vacant or upon inability of all Non-Executive Directors, the General Meeting shall be authorized to temporarily entrust the performance of the duties and the exercise of the authorities of the Non-Executive Directors to one or more other individuals.
- 18.3 A Director shall in any event be unable to act within the meaning of the articles 18.1 and 18.2:
 - (a) during the period for which the Director has claimed inability in writing;
 - (b) during the Director's suspension; or
 - (c) during periods when the Company has not been able to contact the Director (including as a result of illness), provided that such period lasted longer than five consecutive days (or such other period as reasonably determined by the Board).

19 Board: Chairman of the Board, Chair & CEO and other titles

- 19.1 The Board may grant titles to Directors.
- 19.2 The Board shall designate a Non-Executive Director as Chairman of the Board. The Board may designate a Non-Executive Director as vice-chairman. If the Chairman of the Board is absent or unable to act, a vice-chairman, or another Non-Executive Director designated by the Board, is entrusted with the duties of the Chairman of the Board.
- 19.3 The Board shall designate an Executive Director as Chair & CEO.

20 Board: remuneration

- 20.1 The Company has a policy in respect of the remuneration of Executive Directors and Non-Executive Directors. The remuneration policy is adopted by the General Meeting at the proposal of the Board.
- 20.2 The remuneration of the Executive Directors shall be determined by the Board with observance of the remuneration policy adopted by the General Meeting. The Executive Directors shall not participate in the deliberations and decision-making regarding the determination of the remuneration of the Executive Directors.
- 20.3 The remuneration of the Non-Executive Directors shall be determined by the Board with observance of the remuneration policy adopted by the General Meeting.
- 20.4 A proposal with respect to remuneration schemes in the form of Shares or rights to subscribe for Shares shall be submitted by the Board to the General Meeting for its approval. Such proposal shall state at least the maximum number of Shares or rights to subscribe for Shares that may be granted to Directors and the criteria for making or amending such grants.

21 Board: tasks and duties

- 21.1 The Board shall be entrusted with the management of the Company and shall for such purpose have all the powers within the limits of the law that are not granted by these articles of association to others. In the performance of their tasks, the Directors shall be guided by the interests of the Company and the enterprise connected with it.
- 21.2 The Executive Directors are primarily responsible for all day-to-day operations of the Company. The Non-Executive Directors supervise (i) the Executive Directors' policy and performance of duties and (ii) the Company's general affairs and its business, and render advice and direction to the Executive Directors. The Executive Directors shall timely provide the Non-Executive Directors with the information they need to carry out their duties.
- 21.3 The Directors furthermore perform any duties allocated to them under or pursuant to the law or these articles of association.
- 21.4 With due observance of these articles of association, the Board shall adopt Board Regulations dealing with its internal organization, the manner in which decisions are taken, any quorum requirements, the composition, duties and organization of committees and any other matters concerning the Board, the Executive Directors, the Non-Executive Directors and committees established by the Board.

- 21.5 The Board may allocate its duties and powers among the Directors pursuant to the Board Regulations or otherwise in writing, provided that the following duties and powers may not be allocated to the Executive Directors:
 - (a) supervising the performance of the Executive Directors;
 - (b) making a nomination for the appointment of Directors pursuant to article 17.2;
 - (c) determining an Executive Director's remuneration pursuant to article 20.2; and
 - (d) instructing an auditor pursuant to article 33.1.
- 21.6 Without prejudice to any other provisions of these articles of association, the Board shall require the approval of the General Meeting for resolutions regarding a significant change in the identity or nature of the Company or the enterprise connected with it, including in any event:
 - (a) the transfer of the business enterprise, or practically the entire business enterprise, to a third party;
 - (b) concluding or cancelling any long-lasting cooperation of the Company or a Subsidiary with any other legal person or company or as a fully-liable general partner in a partnership, provided that such cooperation or cancellation thereof is of material significance to the Company; and
 - (c) acquiring or disposing of a participating interest in the share capital of a company with a value of at least one-third of the Company's assets, as shown in the consolidated balance sheet with explanatory notes thereto according to the last adopted Annual Accounts, by the Company or a Subsidiary.

22 Board: decision-making

- 22.1 Resolutions of the Board shall be adopted by a simple majority of the votes cast, unless the Board Regulations provide for a qualified majority. Each Director shall have one vote. If there is a tie vote, the proposal shall be rejected.
- 22.2 At a meeting of the Board, a Director may only be represented by another Director holding a proxy in writing.
- 22.3 Meetings of the Board may be held by means of an assembly of Directors in person or by telephone, video conference or any other means of electronic communication, provided that all Directors participating in such meeting are able to communicate with each other simultaneously. Participation in a meeting held in any of the foregoing ways shall constitute presence at such meeting.
- 22.4 A document stating that one or more resolutions have been adopted by the Board and signed by the Chairman of the Board or by the chairperson and secretary of the particular meeting constitutes valid proof of those resolutions.
- 22.5 The Board may also adopt resolutions outside of a meeting, provided that such resolutions are recorded in writing or otherwise and that none of the Directors entitled to vote objects to this manner of decision-making.

- 22.6 A Director shall not participate in deliberations and the decision-making process in the event of a direct or indirect personal conflict of interest between that Director and the Company and the enterprise connected with it. If the Board is unable to adopt a resolution as a result of all Directors being unable to participate in the deliberations and decision-making process due to such a conflict of interest, the decision shall nevertheless be taken by the Board, but the Board shall record in writing the reasons for the resolution.
- 22.7 The Board may determine pursuant to the Board Regulations or otherwise in writing that one or more Directors can lawfully adopt resolutions concerning matters belonging to their duties within the meaning of Section 2:129a(3) DCC.

23 Board: indemnification

- 23.1 The Company shall indemnify each current or former Director in any anticipated or pending action, suit, proceeding or investigation for any claim against that Director that such Director may derive from exercising his respective duties as a Director for any and all:
 - (a) costs and expenses, including but not limited to substantiated attorneys' fees, reasonably incurred in relation to that Director's defences in the relevant action, suit, proceeding or investigation or a settlement thereof;
 - (b) liabilities, losses, damages, fines, penalties and other claims and/or financial effects of judgements against that Director, excluding any reputational damages and (other) immaterial damages; and
 - (c) payments by that Director and/or any other financial effects resulting from a settlement of such action, suit, proceeding or investigation, excluding any reputational damages and (other) immaterial damages, subject to prior written approval of such settlement by the Company (such approval not to be unreasonably withheld),

provided he acted in good faith and in a manner he reasonably believed to be in, or not opposed to, the best interests of the Company or out of his mandate, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his conduct was unlawful.

- 23.2 Any indemnification by the Company referred to in article 23.1 shall be made only upon a determination by the Board that indemnification of the Director is proper under the circumstances because he had met the applicable standard of conduct set forth in article 23.1.
- 23.3 Indemnified amounts referred to in article 23.1 under (a) until (c) inclusive may be paid by the Company in advance of the final disposition of the relevant anticipated or pending action, suit or proceeding against that Director, upon a resolution of the Board with respect to the specific case
- 23.4 A Director, current or former, shall not be entitled to any indemnification as mentioned in this article 23, if and to the extent:
 - (a) a competent court, a judicial tribunal or, in case of an arbitration, an arbitrator or arbitral panel has established by final judgement that is not open to challenge or appeal, that the acts or omissions of the current or former Director can be considered intentional, willfully reckless or seriously culpable, unless this would in the given circumstances be unacceptable according to the standards of reasonableness and fairness;
 - (b) the costs or the decrease in assets of the current or former Director are/is covered by an insurance and the insurer started payment of the costs or the decrease in assets; or

(c) the Company and/or a company in the group brought the procedure in question up before the relevant court, judicial tribunal or, in case of an arbitration, arbitrator or arbitral panel,

in which event he shall immediately repay any amount paid to him (in advance, as the case may be) by the Company under this article 23.

24 Representation

- 24.1 The Company shall be represented by the Board. Any Executive Director shall also be authorized to represent the Company.
- 24.2 The Board may appoint officers with general or limited power to represent the Company. Each officer shall be competent to represent the Company, subject to the restrictions imposed on him. The Board shall determine each officer's title. Such officers may be registered with the Dutch trade register, indicating the scope of their power to represent the Company.

CHAPTER VIII - GENERAL MEETING AND CLASS MEETINGS

25 General Meeting

- 25.1 General Meetings can be held in Amsterdam or Haarlemmermeer (including Schiphol Airport).
- 25.2 The annual General Meeting shall be held each year within six months after the end of the Company's financial year.
- 25.3 Other General Meetings shall be held as often as the Board or the Chair & CEO deems necessary.

26 General Meeting: convocation

- 26.1 General Meetings are convened by the Board or the Chair & CEO.
- 26.2 One or more Shareholders and/or other Persons with Meeting Right who individually or jointly represent at least the part of the Company's issued capital prescribed by law for this purpose, may request the Board in writing to convene a General Meeting setting out in detail the matters to be discussed. If the Board has not taken the steps necessary to ensure that the General Meeting could be held within the relevant statutory period after the request, the requesting Shareholders and/or other Persons with Meeting Right may at their request be authorized by the preliminary relief judge of the district court to convene a General Meeting.

27 General Meeting: notice and agenda

- 27.1 The notice of a General Meeting shall be given by the Board by means of an announcement with due observance of the statutory notice period and in accordance with the law.
- 27.2 The notice of a General Meeting shall state the items to be dealt with, the items to be discussed and which items to be voted on, the place and time of the meeting, the procedure for participating at the meeting whether or not by written proxy-holder, the address of the website of the Company and, if applicable, the procedure for participating at the meeting and exercising one's right to vote by electronic means of communication as referred to in article 28.4, with due observance of the relevant provisions of the law.

- 27.3 The notice of a General Meeting shall also state the Record Date and the manner in which the Persons with Meeting Right may procure their registration and exercise their rights.
- A subject for discussion which has been requested in writing by one or more Shareholders and/or other Persons with Meeting Right who individually or jointly represent at least the part of the Company's issued capital prescribed by law for this purpose, shall be included in the notice of the General Meeting or shall be notified in the same manner as the other subjects for discussion, provided the Company has received the request (including the reasons for such request) not later than sixty days before the day of the meeting. Such written requests must comply with the conditions stipulated by the Board as posted on the Company's website.

28 General Meeting: admittance

- 28.1 Those Persons with Meeting Right and those Persons with Voting Rights who are listed on the Record Date for a General Meeting as such in a register designated for that purpose by the Board, are deemed Persons with Meeting Right or Persons with Voting Rights, respectively, for that General Meeting, regardless of who is entitled to the Shares at the date of the General Meeting.
- 28.2 In order for a person to be able to exercise Meeting Right and the right to vote in a General Meeting, that person must notify the Company in writing of his intention to do so no later than on the date and in the manner mentioned in the notice convening the General Meeting.
- 28.3 The Board may determine in the notice of a General Meeting that any vote cast prior to the meeting by means of electronic communication or by means of a letter, shall be deemed to be a vote cast in the meeting. Such a vote may not be cast prior to the Record Date for the General Meeting.
- 28.4 The Board may determine that each Person with Meeting Right has the right, in person or represented by a written proxy, to take part in, address and, to the extent applicable, to vote at the General Meeting by means of electronic communication, provided that such person can be identified via the same electronic means and is able to directly observe the proceedings and, to the extent applicable, to vote at the meeting. The Board may attach conditions to the use of the electronic communication, provided that these conditions are reasonable and necessary for the identification of the Person with Meeting Right and for the reliability and security of the communication. The conditions must be included in the notice of a General Meeting and be published on the Company's website.
- 28.5 The Directors are authorized to attend the General Meeting and shall, as such, have an advisory vote at the General Meeting.
- 28.6 The chairperson of the General Meeting decides on all matters relating to admission to the General Meeting. The chairperson of the General Meeting may admit third parties to the General Meeting.
- 28.7 The Company may direct that any person, before being admitted to a General Meeting, identifies himself by means of a valid passport or other means of identification and/or should be submitted to such security arrangements as the Company may consider to be appropriate under the given circumstances.

28.8 The General Meeting may be conducted in Dutch or English as determined by the chairperson of the General Meeting.

29 General Meeting: chairperson, secretary and minutes

- 29.1 The General Meeting shall be presided over by the Chairman of the Board or another Director designated for that purpose by the Board. If the Chairman of the Board is not present at the meeting and no other Director has been designated by the Board to preside over the General Meeting, the General Meeting itself shall appoint a chairperson.
- 29.2 The chairperson of the General Meeting shall appoint a secretary of the General Meeting.
- 29.3 Minutes of the proceedings at a General Meeting shall be kept by the secretary, unless a notarial record of the General Meeting is prepared at the request of the Board. The minutes shall be adopted by the chairperson and the secretary of the General Meeting and shall be signed by them as evidence thereof. A document stating that one or more resolutions have been adopted by the General Meeting and signed by the chairperson and secretary of the particular meeting constitutes valid proof of those resolutions.

30 General Meeting: decision-making

- 30.1 Each Class A Share confers the right to cast one vote at the General Meeting and each Class B Share confers the right to cast ten votes at the General Meeting. If and to the extent voting rights are not suspended, each Conversion Share confers the right to cast nine votes at the General Meeting.
- 30.2 To the extent the law or these articles of association do not require a qualified majority, all resolutions of the General Meeting shall be adopted by a simple majority of the votes cast.
- 30.3 The chairperson of the General Meeting shall decide on the method of voting.
- 30.4 Abstentions, blank votes and invalid votes shall not be counted as votes.
- 30.5 The ruling by the chairperson of the General Meeting on the outcome of a vote shall be decisive.
- 30.6 All disputes concerning voting for which neither the law nor these articles of association provide a solution are decided by the chairperson of the General Meeting.
- 30.7 No votes may be cast at the General Meeting for a Share held by the Company or a Subsidiary, nor for any Share for which the Company or a Subsidiary holds the depositary receipts. The Company or a Subsidiary may not cast a vote in respect of a Share on which it holds a right of pledge or a usufruct. However, holders of a right of pledge or a usufruct on Shares held by the Company or a Subsidiary are not excluded from voting, if the right of pledge or the usufruct was created before the Share belonged to the Company or a Subsidiary.
- 30.8 When determining how many votes are cast by Shareholders, how many Shareholders are present or represented, or which part of the Company's issued capital is represented at the General Meeting, no account shall be taken of Shares for which, pursuant to the law or these articles of association, no vote can be cast.

31 Class meetings

- 31.1 The provisions of these articles of association with respect to General Meetings, save for article 25.2, shall apply *mutatis mutandis* to a meeting of holders of Class A Shares and other persons entitled to attend such meeting.
- 31.2 The provisions of these articles of association with respect to General Meetings, save for article 25.2 and article 28.5, shall apply *mutatis mutandis* to a meeting of holders of Class B Shares or a meeting of holders of Conversion Shares (as applicable) and other persons entitled to attend such meeting, provided that the applicable meeting shall appoint its own chairman, and furthermore provided that for as long as Class B Shares or Conversion Shares (as applicable) are not admitted to listing and trading on a stock exchange with the cooperation of the Company:
 - (a) notice of a meeting as referred to in this article shall be given no later than on the fifteenth day before the date of the meeting and the convocation notice shall be sent to the addresses as included in the shareholders' register;
 - (b) resolutions may be adopted in writing without holding a meeting as referred to in this article, provided such resolutions are adopted by the unanimous vote of all holders of Class B Shares entitled to vote or Conversion Shares entitled to vote (as applicable); and
 - (c) valid resolutions may be adopted if the formalities for convening and holding of meetings as referred to in this article have not been complied with, if adopted by unanimous vote in a meeting at which all issued and outstanding Class B Shares or Conversion Shares (as applicable) are represented.

CHAPTER IX - FINANCIAL YEAR, ANNUAL ACCOUNTS AND AUDITOR

32 Financial year and annual accounts

- 32.1 The Company's financial year shall be the calendar year.
- 32.2 Annually, within the term set by law, the Board shall prepare the Annual Accounts. The Annual Accounts must be accompanied by an auditor's statement as referred to in article 33.3, the Management Report, and the additional information to the extent that this information is required.
- 32.3 The Annual Accounts shall be signed by the Directors; if one or more of their signatures is lacking, this shall be stated, giving the reasons therefor.
- 32.4 The Annual Accounts shall be adopted by the General Meeting.

33 Auditor

- 33.1 The General Meeting shall instruct an Auditor to audit the Annual Accounts. If the General Meeting fails to issue the instructions to an Auditor, the Board shall be authorized to do so. The Executive Directors shall not participate in the deliberations and decision-making regarding instructing an Auditor to audit the Annual Accounts.
- 33.2 The instructions issued to the Auditor may only be revoked by the General Meeting and, if the Board issued the instructions, by the Board, for valid reasons and in accordance with Section 2:393(2) DCC.
- 33.3 The Auditor shall report the findings of the audit to the Board and present the results of the audit in a statement on the true and fair view provided by the Annual Accounts.

CHAPTER X - RESULT AND DISTRIBUTIONS

34 Profits and distributions

- 34.1 Distribution of profits shall be made after adoption of the Annual Accounts from which it appears that the same is permitted.
- 34.2 Distributions may be made only to the extent the Company's equity exceeds the sum of its paid up and called up part of its issued capital and the reserves which must be maintained pursuant to the law.
- 34.3 The Board may determine which part of the profits shall be reserved, with due observance of the Company's policy on reserves and dividends. The Company may have a policy on reserves and dividends to be determined and amended by the Board.
- 34.4 The General Meeting may resolve to distribute any part of the profits remaining after reservation in accordance with article 34.3. If the General Meeting does not resolve to distribute these profits in whole or in part, such profits (or any profits remaining after distribution) shall also be reserved.
- 34.5 The General Meeting may only resolve to distribute to the Shareholders a dividend in kind or in the form of Shares at the proposal of the Board.
- 34.6 The Board, or the General Meeting at the proposal of the Board, may resolve to make distributions from the share premium reserve or other distributable reserves maintained by the Company.
- 34.7 The Board may resolve to make interim distributions on Shares, provided that an interim statement of assets and liabilities drawn up in accordance with the statutory requirements shows that the requirement of article 34.2 has been fulfilled, and with observance of (other) applicable statutory provisions.
- 34.8 The Board, or the General Meeting at the proposal of the Board, may resolve that a distribution on Shares shall not be paid in whole or in part in cash but in kind or in the form of Shares, or decide that Shareholders shall be given the option to receive the distribution in cash or in kind and/or in the form of Shares (and with due observance of articles 7 and 8), and may determine the conditions under which such option can be given to the Shareholders.
- 34.9 In calculating the amount of any distribution on Shares, Shares held by the Company shall be disregarded, unless such Shares are encumbered with a usufruct or right of pledge.
- 34.10 Any and all distributions on the Class A Shares and Class B Shares shall be made in such a way that on each Class A Share and Class B Share an equal amount or value will be distributed, provided that and with observance of the following order of priority:
 - (a) in the event of a distribution of profits in respect of a financial year, the Conversion Share Distribution shall first be distributed on each issued and outstanding Conversion Share; and
 - (b) following the Conversion Share Distribution, no further distribution shall be made on Conversion Shares in respect of such financial year.

35 Notices and payments

- 35.1 The date on which dividends and other distributions shall be made payable shall be announced in accordance with the law and published on the Company's website.
- 35.2 Distributions shall be payable on the date determined by the Board.
- 35.3 The persons entitled to a distribution shall be the relevant Shareholders, holders of a usufruct on Shares and holders of a right of pledge on Shares, at a date to be determined by the Board for that purpose. This date shall not be earlier than the date on which the distribution was announced.
- 35.4 Distributions which have not been claimed upon the expiry of five years and one day after the date when they became payable will be forfeited to the Company and will be carried to the reserves.
- 35.5 The Board may determine that distributions on Shares will be made payable either in euro or in another currency.

CHAPTER XI - AMENDMENT OF THE ARTICLES OF ASSOCIATION, DISSOLUTION AND LIQUIDATION

36 Amendment of the articles of association

- 36.1 The General Meeting may resolve to amend these articles of association at the proposal of the Board.
- 36.2 If a proposal to amend these articles of association is to be submitted to the General Meeting, the notice of such meeting must state so and a copy of the proposal, including the verbatim text thereof, shall be deposited and kept available at the Company's office for inspection by, and must be made available free of charge to, Shareholders and other Persons with Meeting Right, until the conclusion of the meeting. An amendment of these articles of association shall be laid down in a notarial deed.

37 Dissolution and liquidation

- 37.1 The General Meeting may resolve to dissolve the Company at the proposal of the Board.
- 37.2 If the Company is dissolved pursuant to a resolution of the General Meeting, the Directors shall become liquidators of the dissolved Company's property. The General Meeting may decide to appoint other persons as liquidators.
- 37.3 During liquidation, to the extent possible these articles of association shall continue to apply.
- 37.4 The balance remaining after payment of the debts of the dissolved Company shall be transferred to the Shareholders as follows and in the following order of priority:
 - (a) an amount equal to the nominal value of Conversion Shares shall first be transferred on each Conversion Share to the holders of the Conversion Shares; and
 - (b) the balance remaining thereafter shall be transferred *pro rata* in proportion to the number of Class A Shares and Class B Shares held by each Shareholder.

38 Federal forum provision

The sole and exclusive forum for any complaint asserting a cause of action arising under the United States Securities Act of 1933, as amended, to the fullest extent permitted by applicable law, shall be the federal district courts of the United States of America. This shall not apply to causes of action arising under the United States Securities Exchange Act of 1934, as amended.

39 First financial year

The first financial year of the Company shall end on the thirty-first day of December two thousand twenty-one. This article and its heading shall cease to exist after the end of the first financial year.

40 Transitory provisions authorized capital

- 40.1 The provisions of articles 5.1 and 5.2 shall only come into effect if and when Shares are issued to parties holding shares in Wall Box Chargers, S.L. prior to the Closing Date (**Authorized Capital Condition Precedent**), which shall be evidenced by the filing by the Company with the Dutch trade register that states that the Authorized Capital Condition Precedent has been fulfilled.
- 40.2 Upon these articles of association taking effect and prior to the fulfilment of the Authorized Capital Condition Precedent, the articles 5.1 and 5.2 shall read as follows:
 - "5.1 The authorized capital of the Company equals sixty thousand one hundred twenty-two euro and sixteen eurocent (EUR 60,122.16).
 - 5.2 The authorized capital of the Company is divided into five hundred thousand (500,000) Class A Shares, with a nominal value of twelve eurocent (EUR 0.12) each, one hundred (100) Class B Shares, with a nominal value of one euro and twenty eurocent (EUR 1.20) each, and two (2) Conversion Shares, with a nominal value of one euro and eight eurocent (EUR 1.08) each."
- 40.3 This article and its heading shall cease to exist upon the fulfilment of the Authorized Capital Condition Precedent.

* * * * *

NOTE ABOUT TRANSLATION:

This document is an English translation of a deed (to be) executed in the Dutch language. In preparing this document, an attempt has been made to translate as literally as possible without jeopardising the overall continuity of the text. The definitions in article 1.1 of this document are listed in the English alphabetical order which may differ from the Dutch alphabetical order. Inevitably, however, differences may occur in translation and if they do, the Dutch text will govern by law. In this translation, Dutch legal concepts are expressed in English terms and not in their original Dutch terms. The concepts concerned may not be identical to concepts described by the English terms as such terms may be understood under the laws of other jurisdictions.

AMENDMENT AND CONVERSION

(new name: Wallbox N.V.)

This first day of October two thousand twenty-one, there appeared before me, Bastiaan Cornelis Cornelisse, civil law notary officiating in Amsterdam, the Netherlands:

•, born in •, on •, with office address at Parnassusweg 300, 1081 LC Amsterdam, the Netherlands.

The person appearing declared the following:

on the first day of October two thousand twenty-one the general meeting of <u>Wallbox B.V.</u>, a private limited liability company under Dutch law, having its official seat in Amsterdam, the Netherlands, and with address at Carrer del Foc 68, 08038 Barcelona, Spain, registered with the Dutch trade register under number 83012559 (**Company**), resolved to convert the Company into a public limited liability company under Dutch law, to amend and completely readopt the articles of association of the Company, as well as to authorise the person appearing to have this deed executed. The adoption of such resolutions is evidenced by a copy of a shareholder's resolution, which shall be attached to this deed (<u>Annex I</u>).

The articles of association of the Company were established at the incorporation of the Company, by a notarial deed, executed on the seventh day of June two thousand twenty-one before the undersigned civil law notary. The articles of association have not been amended since.

In implementing the aforementioned resolution, the Company is converted into a public limited liability company under Dutch law as of today and the articles of association of the Company are amended and completely readopted as follows.

ARTICLES OF ASSOCIATION

CHAPTER I - DEFINITIONS AND INTERPRETATION

1 Definitions

1.1 In these articles of association the following words shall have the following meanings:

Annual Accounts: the Company's annual accounts as referred to in Section 2:361 DCC;

Auditor: an auditor as referred to in Section 2:393 DCC, or an organization in which such auditors work together;

Board: the Company's board of directors;

Board Regulations: the written rules and regulations adopted by the Board as referred to in article 21.4;

Chair & CEO: the Executive Director designated by the Board as chair and chief executive officer (CEO);

Chairman of the Board: the Non-Executive Director who serves as chairman of the Board as referred to under Dutch law;

Class A Share: an ordinary class A share in the Company's share capital;

Class B Share: an ordinary class B share in the Company's share capital;

Company: the Company to which these articles of association pertain;

Closing Date: the date on which the Company issues Class B Shares for the first time;

Conversion Date: the date that the Non-Executive Directors determine, in their sole discretion, that a Conversion Event has occurred, provided that the Conversion Date shall be no later than the date of such determination;

Conversion Event means the occurrence of:

- (a) any Transfer of a Class B Share that is not a Permitted Transfer;
- (b) the Final Conversion Event;
- (c) with respect to a Class B Share held by an Excluded Holder or such Excluded Holder's Permitted Entity, the death or Disability of such Excluded Holder; or
- (d) with respect to a Class B Share held by a Permitted Entity or Permitted Transferee, an event as a consequence of which such Permitted Entity or Permitted Transferee (as applicable);

Conversion Request shall have the meaning given thereto in article 6A;

Conversion Share: an ordinary conversion share in the Company's share capital;

Conversion Share Distribution: a distribution on each Conversion Share for an amount equal to one percent (1%) of the nominal value of Conversion Shares;

DCC: the Dutch Civil Code;

Director: an Executive Director or Non-Executive Director;

Disability or **Disabled**: the permanent and total disability such that an Excluded Holder is unable to engage in any substantial gainful activity by reason of any medically determinable mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than twelve (12) months as determined by a licensed medical practitioner. In the event of a dispute whether an Excluded Holder has suffered a Disability, no Disability of the Excluded Holder shall be deemed to have occurred unless and until an affirmative ruling regarding such Disability has been made by a court of competent jurisdiction, and such ruling has become final and non-appealable, whereby the start date of the Disability will be either the date of determination of the Disability by a licensed medical practitioner or the date the ruling by a court of competent jurisdiction has become final and non-appealable;

Excluded Holder: each of Enric Asunción Escorsa, born on the third day of April nineteen hundred eighty-five, and Eduard Castañeda, born on the twelfth day of August nineteen hundred eighty-five;

Executive Director: a member of the Board appointed as executive director;

Final Conversion Event: the conversion event occurring at the earliest of:

- (a) the date set by the Board that is no less than sixty-one (61) days and no more than one hundred eighty (180) days after the date on which the aggregate number of issued and outstanding Class B Shares held (jointly) by the Excluded Holders and their Permitted Entities represents less than twenty percent (20%) of the aggregate number of issued and outstanding Class B Shares held by the Initial Holders on the Closing Date; or
- (b) the date set by the meeting of holders of Class B Shares;

General Meeting: the corporate body of the Company consisting of Shareholders and all other Persons with Meeting Right or a meeting of Shareholders and other Persons with Meeting Right, as the case may be;

Group Company: a group company of the Company as referred to in Section 2:24b DCC;

in writing: by letter, by telecopier, by e-mail, or by a legible and reproducible message otherwise electronically sent, provided that the identity of the sender can be sufficiently established;

Initial Holder means, in relation to any Class B Share, the person or entity holding such Class B Share on the Closing Date;

Management Report: the Company's management report as referred to in Section 2:391 DCC;

Meeting Right: the right, either in person or by proxy authorized in writing, to attend and address the General Meeting;

Non-Executive Director: a member of the Board appointed as non-executive director;

Permitted Entity: means any entity (including a corporation, (limited liability) company, partnership, foundation and trust) so long as an Excluded Holder has exclusive Voting Control with respect to the Class B Shares held by such entity;

Permitted Transfer means:

(a) a Transfer between (i) Initial Holders, (ii) an Initial Holder and an Excluded Holder, and (iii) an Initial Holder and a Permitted Entity;

- (b) a Transfer between (i) Excluded Holders, and (ii) an Excluded Holder and a Permitted Entity;
- (c) a Transfer between Permitted Entities;

Permitted Transferee: a transferee of Class B Shares, or rights or interests therein, received in a Transfer that constitutes a Permitted Transfer;

Persons with Meeting Right: Shareholders, holders of a usufruct with Meeting Right and holders of a right of pledge with Meeting Right;

Persons with Voting Rights: Shareholders with voting rights, holders of a usufruct with voting rights and holders of a right of pledge with voting rights in the General Meeting;

Record Date: the twenty-eighth day prior to the date of a General Meeting, or such other day as prescribed by law;

Share: a share in the Company's share capital, unless the contrary is expressed this shall include each Class A Share, each Class B Share and each Conversion Share;

Shareholder: a holder of one or more Shares;

Subsidiary: a subsidiary of the Company as referred to in Section 2:24a DCC;

Transfer: any sale, assignment, transfer under general or specific title, conveyance, grant of any form of security interest (other than as explicitly provided in this definition), or other transfer or disposition of a Class B Share or any legal or beneficial interest in such Class B Share, whether or not for value and whether voluntary or involuntary or by operation of law, whether directly or indirectly, including by merger, consolidation or otherwise. A "Transfer" shall also include, without limitation, the transfer of, or entering into a binding agreement with respect to, Voting Control over a Class B Share by proxy or otherwise; provided, however, that the following shall not be considered a "Transfer":

- (a) the granting of a power of attorney (i) in connection with actions to be taken at a specific General Meeting, and/or (ii) pursuant to article 6A.1 (c), as well as the exercise of any such power of attorney by the respective attorney;
- (b) the creation of a right of pledge on Class B Shares that creates a mere security interest in such Class B Shares pursuant to a bona fide loan or indebtedness transaction so long as the holder of Class B Shares continues to exercise Voting Control over such pledged Class B Shares, provided, however, that an enforcement of a right of pledge on such Class B Shares or other similar action by the pledgee shall constitute a "Transfer" of a Class B Share;
- (c) the entering into of a voting trust, agreement or arrangement (with or without granting a proxy), or consummating the actions or transactions contemplated therein, between Initial Holders, Excluded Holders and/or Permitted Entities, that (i) is disclosed either in an applicable schedule filed with the United States securities and exchange commission or in writing to the Company, (ii) either has a term not exceeding one year or is terminable by the holder of Class B Shares at any time and (iii) does not involve any payment of cash, securities, property or other consideration to the holders of Class B Shares other than the mutual promise to vote shares in a designated manner;

(d) the entering into of any agreement with respect to supporting or voting in favor of or tendering Shares with respect to any transaction proposed to be supported by the Company, or supporting the actions or transactions contemplated therein (including, without limitation, tendering Class B Shares or voting such shares in favor of such transaction or in opposition to other proposals that may be expected to delay or impair the ability to support such transaction), if the entry into such support or voting agreement is approved by the Board;

Transferor shall have the meaning given thereto in article 6B; and

Voting Control means with respect to a Class B Share, the power to vote or direct the voting of such Class B Share, including by proxy, voting agreement or otherwise.

2 Construction

- 2.1 References to articles shall be deemed to refer to articles of these articles of association, unless the contrary is apparent.
- 2.2 Any reference to a gender includes all genders.

CHAPTER II - NAME, CORPORATE SEAT AND OBJECTS

3 Name and corporate seat

- 3.1 The Company's name is Wallbox N.V.
- 3.2 The corporate seat of the Company is in Amsterdam, the Netherlands.

4 Objects

The objects of the Company are:

- (a) to design, manufacture and distribute charging solutions for residential, business, and public use, and to provide installation services;
- (b) to incorporate, to participate in any way whatsoever in, to manage, to supervise businesses and companies;
- (c) to finance businesses and companies;
- (d) to borrow, to lend and to raise funds, including the issue of bonds, promissory notes or other securities or evidence of indebtedness as well as to enter into agreements in connection with aforementioned activities;
- (e) to render advice and services to businesses and companies with which the Company forms a group and to third parties;
- (f) to grant guarantees, to bind the Company and to pledge its assets for obligations of the Company, its group companies and/or third parties;
- (g) to acquire, alienate, manage and exploit registered property and items of property in general;
- (h) to trade in currencies, securities and items of property in general;
- (i) to develop and trade in patents, trade marks, licenses, know-how and other intellectual and industrial property rights; and
- (j) to perform any and all activities of an industrial, financial or commercial nature,

and to do all that is connected therewith or may be conducive thereto, all to be interpreted in the broadest sense.

CHAPTER III - SHARE CAPITAL

5 Authorized capital, share premium reserve

- 5.1 The authorized capital of the Company equals one hundred eight million two euro and sixteen eurocent (EUR 108,000,002.16).
- 5.2 The authorized capital of the Company is divided into four hundred million (400,000,000) Class A Shares, with a nominal value of twelve eurocent (EUR 0.12) each, fifty million (50,000,000) Class B Shares, with a nominal value of one euro and twenty eurocent (EUR 1.20) each, and two (2) Conversion Shares, with a nominal value of one euro and eight eurocent (EUR 1.08) each.
- 5.3 Upon the conversion of one or more Class B Shares into Class A Shares and Conversion Shares referred to in article 6, the authorized capital shall decrease with the number of Class B Shares so converted and shall increase with the number of Class A Shares and Conversion Shares into which such Class B Shares were converted.
- 5.4 Within eight days after a conversion of one or more Class B Shares into Class A Shares and Conversion Shares referred to in article 6, the Board shall (i) file a notification thereof with the Dutch trade register, which notification must at least include the authorized capital following the conversion, and (ii) register the conversion in the register of Shareholders as referred to in article 13.
- 5.5 No share certificates shall be issued.
- 5.6 The Company shall maintain a general share premium reserve for the benefit of the Shareholders.

6 Conversion of Shares

6.1 Each Class B Share can be converted into one Class A Share and one Conversion Share, subject to the provisions of this article 6. Class A Shares and Conversion Shares cannot be converted into other classes of Shares.

6A Voluntary conversion of Class B Shares

- 6A.1 Each holder of one or more Class B Shares may request the conversion of all or part of his Class B Shares into Class A Shares and Conversion Shares in the ratio set out in article 6.1 by means of a written request addressed to the Board (Conversion Request). The Conversion Request must:
 - (a) indicate the number of Class B Shares to which the Conversion Request pertains;
 - (b) include a representation by the requesting Shareholder that no depositary receipts have been issued with respect to the Class B Shares to which the Conversion Request pertains nor that these Class B Shares are encumbered with any usufruct, right of pledge, attachment or other encumbrance;
 - (c) provide for an irrevocable and unconditional power of attorney from the requesting Shareholder to the Company, with full power of substitution and governed by Dutch law, to perform the acts described in article 6A.2 on behalf of such Shareholder.

6A.2 Subject to article 6A.3:

- (a) upon receipt of a Conversion Request satisfactory to the Board, the Board shall, as soon as reasonably possible, resolve to convert the number of Class B Shares to which the Conversion Request pertains into Class A Shares and Conversion Shares in the ratio set out in article 6.1; and
- (b) promptly following such conversion, the requesting Shareholder shall be obliged to offer and transfer the Conversion Shares to the Company for no consideration and the Company shall accept such Conversion Shares.
- 6A.3 To the extent the Company would not be permitted under applicable law to acquire Conversion Shares in accordance with article 6A.2 paragraph (b), the Board shall, as soon as reasonably possible, convene a General Meeting in accordance with article 35, at which meeting it shall be proposed to the General Meeting to cancel such number of Shares (held in treasury) to allow again for the acquisition of Conversion Shares as referred to in article 6A.2 paragraph (b).

6B Mandatory conversion of Class B Shares

- 6B.1 A Class B Share shall automatically convert into Class A Shares and Conversion Shares in the ratio set out in article 6.1 upon the occurrence of a Conversion Event in respect of such Class B Share, subject to the provisions of this article 6B and with effect as of the Conversion Date. Upon the occurrence of a Conversion Event, the Shareholder concerned shall be obliged to notify the Board thereof by means of a written notice addressed to the Board.
- 6B.2 If at any time a Conversion Share is held by anyone other than the Company, such holder of Conversion Shares (**Transferor**) shall be obliged to offer and transfer such Conversion Shares to the Company unencumbered (without any usufruct, right of pledge, attachment or other encumbrance and without depositary receipts issued for such Conversion Shares) and for no consideration. If and for as long as the Transferor fails to offer and transfer the relevant Conversion Shares to the Company, the voting rights, Meeting Right and rights to receive distributions attached to the relevant Conversion Shares are suspended.
- 6B.3 If the Transferor fails to offer and transfer the relevant Conversion Shares to the Company within:
 - (a) sixty (60) days after the Conversion Date for Conversion Shares acquired as a result of a Conversion Event (c); or
 - (b) thirty (30) days after the Conversion Date for Conversion Shares acquired as a result of a Conversion Event other than (c), the Company is irrevocably empowered and authorized to offer and transfer the relevant Conversion Shares to the Company until such

necessary to verify the legal and beneficial ownership of Class B Shares, and to confirm that a Conversion Event has not occurred.

transaction occurs.

The Board may, from time to time, establish such policies and procedures relating to the general administration of the share capital structure as it may deem necessary or advisable, and may request that holders of Class B Shares furnish affidavits or other proof to the Board as it deems

6B.5 To the extent the Company would not be permitted under applicable law to acquire Conversion Shares in accordance with article 6B.2, the Board shall convene a General Meeting in accordance with article 35, at which meeting it shall be proposed to the General Meeting to cancel such number of Shares (held in treasury) to allow again for the acquisition of Conversion Shares as referred to in article 6B.2.

7 Issuance of Shares

- 7.1 Shares shall be issued pursuant to a resolution of the Board if the Board has been designated thereto by the General Meeting for a specific period and with due observance of applicable statutory provisions. Such designation by the General Meeting must state the number of Shares that may be issued.
 - The designation may be extended by specific consecutive periods with due observance of applicable statutory provisions. Unless otherwise stipulated at its grant, the designation may not be withdrawn.
- 7.2 If and insofar as the Board is not designated by the General Meeting, Shares shall be issued pursuant to a resolution of the General Meeting. The General Meeting shall, in addition to the Board, remain authorized to issue Shares if such is specifically stipulated in the resolution authorizing the Board to issue Shares as described in article 7.1.
- 7.3 The articles 7.1 and 7.2 shall apply by analogy to the granting of rights to subscribe for Shares, but shall not apply to an issue of Shares to a person exercising a previously granted right to subscribe for Shares.
- 7.4 A resolution of the General Meeting as referred to in this article 7 can only be adopted at the proposal of the Board.
- 7.5 If the resolution of the General Meeting to issue Shares or to designate the authority to issue Shares to the Board as referred to in article 7.1 is detrimental to the rights of holders of a specific class of Shares, the validity of such resolution of the General Meeting requires a prior or simultaneous approval by the group of holders of such class of Shares.

8 Pre-emptive rights

- 8.1 Each Shareholder shall have a pre-emptive right on any issuance of Class A Shares and Class B Shares in proportion to the aggregate amount of its Class A Shares and Class B Shares. No pre-emptive rights shall apply in respect of any issuance of Conversion Shares.
- 8.2 This pre-emptive right on any issuance of Class A Shares and Class B Shares does not apply to:
 - (a) Shares issued to employees of the Company or a Group Company;
 - (b) Shares that are issued against payment other than in cash; and
 - (c) Shares issued to a person exercising a previously granted right to subscribe for Shares.
- 8.3 Pre-emptive rights may be limited or excluded by a resolution of the General Meeting. Pre-emptive rights may also be limited or excluded by a resolution of the Board if the Board has been designated thereto by the General Meeting for a specific period and with due observance of applicable statutory provisions, and the Board has also been designated to issue Shares in accordance with article 7.1. The designation may be extended by specific consecutive periods with due observance of applicable statutory provisions. Unless otherwise stipulated at its grant, the designation may not be withdrawn.

- 8.4 A resolution of the General Meeting to limit or exclude pre-emptive rights and a resolution to designate the Board thereto, can only be adopted at the proposal of the Board and shall require a majority of at least two-thirds of the votes cast if less than half of the issued capital of the Company is represented at the General Meeting.
- 8.5 When adopting a resolution to issue Shares, the General Meeting or the Board shall determine how and during which period these pre-emptive rights may be exercised, subject to Section 2:96a DCC.
- 8.6 Article 8 shall apply by analogy to the granting of rights to subscribe for Shares.

9 Payment on Shares

- 9.1 Shares may only be issued against payment in full of the amount at which such Shares are issued and with due observance of the provisions of the Sections 2:80, 2:80a and 2:80b DCC.
- 9.2 Payment on Shares must be made in cash if no alternative contribution has been agreed. Payment other than in cash must be made in accordance with the provisions in Section 2:94b DCC. Payment in a currency other than euro may only be made with the consent of the Company and with due observance of the provisions of Section 2:93a DCC.
- 9.3 Shares issued to (i) current or former employees of the Company or a Group Company, (ii) current or former Directors under an equity compensation plan of the Company and (iii) holders of a right to subscribe for Shares granted in accordance with article 7.3 may be paid-up at the expense of the reserves of the Company, notwithstanding the provisions of article 34.
- 9.4 The Board may perform legal acts as referred to in Section 2:94 DCC without the prior approval of the General Meeting.

CHAPTER IV - OWN SHARES AND CAPITAL REDUCTION

10 Share repurchase and disposal of shares

- 10.1 The Company may repurchase fully paid-up Shares:
 - (a) for no consideration; or
 - (b) against consideration if and insofar as the Board has been authorized thereto by the General Meeting for a specific period and with due observance of applicable statutory provisions. The General Meeting shall determine in its authorization how many Shares the Company may repurchase, in what manner and at what price range.
- 10.2 The authorization by the General Meeting is not required if the Company repurchases fully paid-up Shares for the purpose of transferring these Shares to employees of the Company or a Group Company under any applicable equity compensation plan, provided that those Shares are quoted on an official list of a stock exchange.
- 10.3 Any disposal of Shares by the Company shall require a resolution of the Board. Such resolution shall also stipulate any conditions of the disposal.

11 Capital reduction

- 11.1 The General Meeting may only at the proposal of the Board resolve to reduce the Company's issued capital by (i) reducing the nominal value of Shares through amendment of these articles of association or (ii) cancelling Shares held by the Company itself.
- 11.2 A resolution of the General Meeting to reduce the Company's issued capital, shall require a majority of at least two-thirds of the votes cast if less than half of the issued capital of the Company is represented at the General Meeting.
- 11.3 If the resolution of the General Meeting to reduce the Company's issued capital by reducing the nominal value of Shares through amendment of these articles of association, as referred to above, is detrimental to the rights of holders of a specific class of Shares, the validity of such resolution of the General Meeting requires a prior or simultaneous approval by the group of holders of such class of Shares.

CHAPTER V - TRANSFER OF SHARES

12 Transfer of shares

- 12.1 The transfer of Shares shall require a deed executed for that purpose and, save in the event the Company itself is a party to such legal act, written acknowledgement by the Company of the transfer. Service of notice of such deed to the Company or of a true copy or extract of such deed will be the equivalent of such acknowledgement.
- 12.2 Article 12.1 shall apply *mutatis mutandis* to the creation of a right of pledge or usufruct on a Share, provided that a right of pledge may also be created without acknowledgement by or service of notice upon the Company, in which case Section 3:239 DCC applies and the acknowledgement by or service of notice upon the Company shall replace the announcement as referred to in Section 3:239(3) DCC.
- 12.3 If and as long as one or more Class A Shares are admitted to trading on the New York Stock Exchange, or if it may reasonably be expected that one or more Class A Shares shall shortly be admitted to trading on the New York Stock Exchange, the Board may resolve that the laws of the State of New York, United States of America, shall apply to the property law aspects of the Class A Shares, as a result of which the articles 12.1 and 12.2 shall not apply to the Class A Shares. Such resolution and the revocation thereof shall be made available for inspection on the Company's website and at the Dutch trade register.

The property law aspects of the Class A Shares (including the legal rules on ownership, legal title and transfer) in book-entry form, as included in the part of the register of shareholders kept by the relevant transfer agent, shall be governed by the laws of the State of New York, United States of America, in accordance with the applicable law on International Private laws as referred to in Title 10 of Book 10 DCC (*Boek 10 Internationaal privaatrecht*), especially Section 10:141 DCC.

CHAPTER VI - SHAREHOLDERS' REGISTER AND LIMITED RIGHTS TO SHARES

13 Shareholders' register

- 13.1 The Board must keep a shareholders' register; the Board may appoint a registrar to keep the register on its behalf. The register must be regularly updated. The shareholders' register may be kept in several copies and in several places. Part of the register may be kept outside the Netherlands to comply with applicable local law or pursuant to stock exchange rules.
- 13.2 Each Shareholder's name, address and further information as required by law or considered appropriate by the Board are recorded in the shareholders' register. Shareholders shall provide the Board with the necessary particulars in a timely fashion. Any consequences of not, or incorrectly, notifying such particulars will be the responsibility of the Shareholder concerned.
- 13.3 If a Shareholder so requests, the Board shall provide the Shareholder, free of charge, with written evidence of the information in the shareholders' register concerning the Shares registered in the Shareholder's name.
- 13.4 The articles 13.2 and 13.3 shall apply by analogy to holders of a usufruct or right of pledge on one or more Shares, with the exception of a holder of a right of pledge created without acknowledgement by or service of notice upon the Company.

14 Right of pledge on Shares

- 14.1 Class A Shares and Class B Shares can be pledged. Conversion Shares cannot be pledged.
- 14.2 Subject to article 12.3 (if applicable), if a Share is encumbered with a right of pledge, the voting rights attached to that Share shall vest in the Shareholder, unless at the creation of the right of pledge the voting rights were granted to the pledgee. A pledgee with voting rights shall have Meeting Right.
- 14.3 A Shareholder who as a result of a right of pledge does not have voting rights, shall have Meeting Right. A pledgee without voting rights shall not have Meeting Right.

15 Usufruct on Shares

- 15.1 Subject to article 12.3 (if applicable), if a usufruct is created on a Share, the voting rights attached to that Share shall vest in the Shareholder, unless at the creation of the usufruct the voting rights were granted to the usufructuary. A usufructuary with voting rights shall have Meeting Right.
- 15.2 A Shareholder who as a result of a usufruct does not have voting rights, shall have Meeting Right. A usufructuary without voting rights shall not have Meeting Right.

16 Depositary receipts

The Company shall not cooperate with the issuance of depositary receipts for Shares.

CHAPTER VII - MANAGEMENT AND SUPERVISION

17 Board: composition, appointment, suspension and dismissal

17.1 The Company shall be managed by the Board. The Board shall consist of one or more Executive Directors and one or more Non-Executive Directors. The number of Executive Directors and the number of Non-Executive Directors shall be determined by the Board. Only individuals may be Directors.

- 17.2 The Executive Directors and Non-Executive Directors shall be appointed as such by the General Meeting at the nomination of the Board. A nomination by the Board shall state whether a person is nominated for appointment as Executive Director or Non-Executive Director.
- 17.3 A Director shall be appointed for a term of approximately one year, which term of office shall lapse immediately after the close of the annual General Meeting held in the year after his appointment. A Director may be reappointed with due observance of the preceding sentence. At the proposal of the Board, the General Meeting may resolve to deviate from the term of office of approximately one year. A Non-Executive Director may be in office for a period not exceeding twelve (12) years, which period may or may not be interrupted, unless at the proposal of the Board the General Meeting resolves otherwise.
- 17.4 The General Meeting may at all times suspend or dismiss any Director. A resolution of the General Meeting to suspend or dismiss a Director other than at the proposal of the Board requires a two-thirds majority of the votes cast, representing more than one half of the issued capital of the Company. The Board may at all times suspend an Executive Director.
- 17.5 A suspension may be extended one or more times but may not last longer than three (3) months in aggregate. If at the end of that period, no decision has been taken on termination of the suspension or on dismissal, the suspension shall end. A suspension can be terminated by the General Meeting at any time.

18 Board: vacancy or inability

- 18.1 If the seat of an Executive Director is vacant or upon the inability of an Executive Director, the remaining Executive Directors shall temporarily be entrusted with the executive management of the Company, provided that the Board may provide for a temporary replacement. If the seats of all Executive Directors are vacant or upon the inability of all Executive Directors, the executive management of the Company shall temporarily be entrusted to the Non-Executive Directors, provided that the Board may provide for one or more temporary replacements.
- 18.2 If the seat of a Non-Executive Director is vacant or upon inability of a Non-Executive Director, the remaining Non-Executive Directors shall temporarily be entrusted with the performance of the duties and the exercise of the authorities of that Non-Executive Director, provided that the Board may provide for a temporary replacement. If the seats of all Non-Executive Directors are vacant or upon inability of all Non-Executive Directors, the General Meeting shall be authorized to temporarily entrust the performance of the duties and the exercise of the authorities of the Non-Executive Directors to one or more other individuals.
- 18.3 A Director shall in any event be unable to act within the meaning of the articles 18.1 and 18.2:
 - (a) during the period for which the Director has claimed inability in writing;
 - (b) during the Director's suspension; or
 - (c) during periods when the Company has not been able to contact the Director (including as a result of illness), provided that such period lasted longer than five consecutive days (or such other period as reasonably determined by the Board).

19 Board: Chairman of the Board, Chair & CEO and other titles

- 19.1 The Board may grant titles to Directors.
- 19.2 The Board shall designate a Non-Executive Director as Chairman of the Board. The Board may designate a Non-Executive Director as vice-chairman. If the Chairman of the Board is absent or unable to act, a vice-chairman, or another Non-Executive Director designated by the Board, is entrusted with the duties of the Chairman of the Board.
- 19.3 The Board shall designate an Executive Director as Chair & CEO.

20 Board: remuneration

- 20.1 The Company has a policy in respect of the remuneration of Executive Directors and Non-Executive Directors. The remuneration policy is adopted by the General Meeting at the proposal of the Board.
- 20.2 The remuneration of the Executive Directors shall be determined by the Board with observance of the remuneration policy adopted by the General Meeting. The Executive Directors shall not participate in the deliberations and decision-making regarding the determination of the remuneration of the Executive Directors.
- 20.3 The remuneration of the Non-Executive Directors shall be determined by the Board with observance of the remuneration policy adopted by the General Meeting.
- 20.4 A proposal with respect to remuneration schemes in the form of Shares or rights to subscribe for Shares shall be submitted by the Board to the General Meeting for its approval. Such proposal shall state at least the maximum number of Shares or rights to subscribe for Shares that may be granted to Directors and the criteria for making or amending such grants.

21 Board: tasks and duties

- 21.1 The Board shall be entrusted with the management of the Company and shall for such purpose have all the powers within the limits of the law that are not granted by these articles of association to others. In the performance of their tasks, the Directors shall be guided by the interests of the Company and the enterprise connected with it.
- 21.2 The Executive Directors are primarily responsible for all day-to-day operations of the Company. The Non-Executive Directors supervise (i) the Executive Directors' policy and performance of duties and (ii) the Company's general affairs and its business, and render advice and direction to the Executive Directors. The Executive Directors shall timely provide the Non-Executive Directors with the information they need to carry out their duties.
- 21.3 The Directors furthermore perform any duties allocated to them under or pursuant to the law or these articles of association.
- With due observance of these articles of association, the Board shall adopt Board Regulations dealing with its internal organization, the manner in which decisions are taken, any quorum requirements, the composition, duties and organization of committees and any other matters concerning the Board, the Executive Directors, the Non-Executive Directors and committees established by the Board.

- 21.5 The Board may allocate its duties and powers among the Directors pursuant to the Board Regulations or otherwise in writing, provided that the following duties and powers may not be allocated to the Executive Directors:
 - (a) supervising the performance of the Executive Directors;
 - (b) making a nomination for the appointment of Directors pursuant to article 17.2;
 - (c) determining an Executive Director's remuneration pursuant to article 20.2; and
 - (d) instructing an auditor pursuant to article 33.1.
- 21.6 Without prejudice to any other provisions of these articles of association, the Board shall require the approval of the General Meeting for resolutions regarding a significant change in the identity or nature of the Company or the enterprise connected with it, including in any event:
 - (a) the transfer of the business enterprise, or practically the entire business enterprise, to a third party;
 - (b) concluding or cancelling any long-lasting cooperation of the Company or a Subsidiary with any other legal person or company or as a fully-liable general partner in a partnership, provided that such cooperation or cancellation thereof is of material significance to the Company; and
 - (c) acquiring or disposing of a participating interest in the share capital of a company with a value of at least one-third of the Company's assets, as shown in the consolidated balance sheet with explanatory notes thereto according to the last adopted Annual Accounts, by the Company or a Subsidiary.

22 Board: decision-making

- 22.1 Resolutions of the Board shall be adopted by a simple majority of the votes cast, unless the Board Regulations provide for a qualified majority. Each Director shall have one vote. If there is a tie vote, the proposal shall be rejected.
- 22.2 At a meeting of the Board, a Director may only be represented by another Director holding a proxy in writing.
- 22.3 Meetings of the Board may be held by means of an assembly of Directors in person or by telephone, video conference or any other means of electronic communication, provided that all Directors participating in such meeting are able to communicate with each other simultaneously. Participation in a meeting held in any of the foregoing ways shall constitute presence at such meeting.
- 22.4 A document stating that one or more resolutions have been adopted by the Board and signed by the Chairman of the Board or by the chairperson and secretary of the particular meeting constitutes valid proof of those resolutions.
- 22.5 The Board may also adopt resolutions outside of a meeting, provided that such resolutions are recorded in writing or otherwise and that none of the Directors entitled to vote objects to this manner of decision-making.

- A Director shall not participate in deliberations and the decision-making process in the event of a direct or indirect personal conflict of interest between that Director and the Company and the enterprise connected with it. If the Board is unable to adopt a resolution as a result of all Directors being unable to participate in the deliberations and decision-making process due to such a conflict of interest, the decision shall nevertheless be taken by the Board, but the Board shall record in writing the reasons for the resolution.
- 22.7 The Board may determine pursuant to the Board Regulations or otherwise in writing that one or more Directors can lawfully adopt resolutions concerning matters belonging to their duties within the meaning of Section 2:129a(3) DCC.

23 Board: indemnification

- 23.1 The Company shall indemnify each current or former Director in any anticipated or pending action, suit, proceeding or investigation for any claim against that Director that such Director may derive from exercising his respective duties as a Director for any and all:
 - (a) costs and expenses, including but not limited to substantiated attorneys' fees, reasonably incurred in relation to that Director's
 defences in the relevant action, suit, proceeding or investigation or a settlement thereof;
 - (b) liabilities, losses, damages, fines, penalties and other claims and/or financial effects of judgements against that Director, excluding any reputational damages and (other) immaterial damages; and
 - (c) payments by that Director and/or any other financial effects resulting from a settlement of such action, suit, proceeding or investigation, excluding any reputational damages and (other) immaterial damages, subject to prior written approval of such settlement by the Company (such approval not to be unreasonably withheld),

provided he acted in good faith and in a manner he reasonably believed to be in, or not opposed to, the best interests of the Company or out of his mandate, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his conduct was unlawful.

- 23.2 Any indemnification by the Company referred to in article 23.1 shall be made only upon a determination by the Board that indemnification of the Director is proper under the circumstances because he had met the applicable standard of conduct set forth in article 23.1.
- 23.3 Indemnified amounts referred to in article 23.1 under (a) until (c) inclusive may be paid by the Company in advance of the final disposition of the relevant anticipated or pending action, suit or proceeding against that Director, upon a resolution of the Board with respect to the specific case.
- 23.4 A Director, current or former, shall not be entitled to any indemnification as mentioned in this article 23, if and to the extent:
 - (a) a competent court, a judicial tribunal or, in case of an arbitration, an arbitrator or arbitral panel has established by final judgement that is not open to challenge or appeal, that the acts or omissions of the current or former Director can be considered intentional, willfully reckless or seriously culpable, unless this would in the given circumstances be unacceptable according to the standards of reasonableness and fairness;

- (b) the costs or the decrease in assets of the current or former Director are/is covered by an insurance and the insurer started payment of the costs or the decrease in assets; or
- (c) the Company and/or a company in the group brought the procedure in question up before the relevant court, judicial tribunal or, in case of an arbitration, arbitrator or arbitral panel,

in which event he shall immediately repay any amount paid to him (in advance, as the case may be) by the Company under this article 23.

24 Representation

- 24.1 The Company shall be represented by the Board. Any Executive Director shall also be authorized to represent the Company.
- 24.2 The Board may appoint officers with general or limited power to represent the Company. Each officer shall be competent to represent the Company, subject to the restrictions imposed on him. The Board shall determine each officer's title. Such officers may be registered with the Dutch trade register, indicating the scope of their power to represent the Company.

CHAPTER VIII - GENERAL MEETING AND CLASS MEETINGS

25 General Meeting

- 25.1 General Meetings can be held in Amsterdam or Haarlemmermeer (including Schiphol Airport).
- 25.2 The annual General Meeting shall be held each year within six months after the end of the Company's financial year.
- 25.3 Other General Meetings shall be held as often as the Board or the Chair & CEO deems necessary.

26 General Meeting: convocation

- 26.1 General Meetings are convened by the Board or the Chair & CEO.
- 26.2 One or more Shareholders and/or other Persons with Meeting Right who individually or jointly represent at least the part of the Company's issued capital prescribed by law for this purpose, may request the Board in writing to convene a General Meeting setting out in detail the matters to be discussed. If the Board has not taken the steps necessary to ensure that the General Meeting could be held within the relevant statutory period after the request, the requesting Shareholders and/or other Persons with Meeting Right may at their request be authorized by the preliminary relief judge of the district court to convene a General Meeting.

27 General Meeting: notice and agenda

- 27.1 The notice of a General Meeting shall be given by the Board by means of an announcement with due observance of the statutory notice period and in accordance with the law.
- 27.2 The notice of a General Meeting shall state the items to be dealt with, the items to be discussed and which items to be voted on, the place and time of the meeting, the procedure for participating at the meeting whether or not by written proxy-holder, the address of the website of the Company and, if applicable, the procedure for participating at the meeting and exercising one's right to vote by electronic means of communication as referred to in article 28.4, with due observance of the relevant provisions of the law.

- 27.3 The notice of a General Meeting shall also state the Record Date and the manner in which the Persons with Meeting Right may procure their registration and exercise their rights.
- 27.4 A subject for discussion which has been requested in writing by one or more Shareholders and/or other Persons with Meeting Right who individually or jointly represent at least the part of the Company's issued capital prescribed by law for this purpose, shall be included in the notice of the General Meeting or shall be notified in the same manner as the other subjects for discussion, provided the Company has received the request (including the reasons for such request) not later than sixty days before the day of the meeting. Such written requests must comply with the conditions stipulated by the Board as posted on the Company's website.

28 General Meeting: admittance

- 28.1 Those Persons with Meeting Right and those Persons with Voting Rights who are listed on the Record Date for a General Meeting as such in a register designated for that purpose by the Board, are deemed Persons with Meeting Right or Persons with Voting Rights, respectively, for that General Meeting, regardless of who is entitled to the Shares at the date of the General Meeting.
- 28.2 In order for a person to be able to exercise Meeting Right and the right to vote in a General Meeting, that person must notify the Company in writing of his intention to do so no later than on the date and in the manner mentioned in the notice convening the General Meeting.
- 28.3 The Board may determine in the notice of a General Meeting that any vote cast prior to the meeting by means of electronic communication or by means of a letter, shall be deemed to be a vote cast in the meeting. Such a vote may not be cast prior to the Record Date for the General Meeting.
- 28.4 The Board may determine that each Person with Meeting Right has the right, in person or represented by a written proxy, to take part in, address and, to the extent applicable, to vote at the General Meeting by means of electronic communication, provided that such person can be identified via the same electronic means and is able to directly observe the proceedings and, to the extent applicable, to vote at the meeting. The Board may attach conditions to the use of the electronic communication, provided that these conditions are reasonable and necessary for the identification of the Person with Meeting Right and for the reliability and security of the communication. The conditions must be included in the notice of a General Meeting and be published on the Company's website.
- 28.5 The Directors are authorized to attend the General Meeting and shall, as such, have an advisory vote at the General Meeting.
- 28.6 The chairperson of the General Meeting decides on all matters relating to admission to the General Meeting. The chairperson of the General Meeting may admit third parties to the General Meeting.
- 28.7 The Company may direct that any person, before being admitted to a General Meeting, identifies himself by means of a valid passport or other means of identification and/or should be submitted to such security arrangements as the Company may consider to be appropriate under the given circumstances.

28.8 The General Meeting may be conducted in Dutch or English as determined by the chairperson of the General Meeting.

29 General Meeting: chairperson, secretary and minutes

- 29.1 The General Meeting shall be presided over by the Chairman of the Board or another Director designated for that purpose by the Board. If the Chairman of the Board is not present at the meeting and no other Director has been designated by the Board to preside over the General Meeting, the General Meeting itself shall appoint a chairperson.
- 29.2 The chairperson of the General Meeting shall appoint a secretary of the General Meeting.
- 29.3 Minutes of the proceedings at a General Meeting shall be kept by the secretary, unless a notarial record of the General Meeting is prepared at the request of the Board. The minutes shall be adopted by the chairperson and the secretary of the General Meeting and shall be signed by them as evidence thereof. A document stating that one or more resolutions have been adopted by the General Meeting and signed by the chairperson and secretary of the particular meeting constitutes valid proof of those resolutions.

30 General Meeting: decision-making

- 30.1 Each Class A Share confers the right to cast one vote at the General Meeting and each Class B Share confers the right to cast ten votes at the General Meeting. If and to the extent voting rights are not suspended, each Conversion Share confers the right to cast nine votes at the General Meeting.
- 30.2 To the extent the law or these articles of association do not require a qualified majority, all resolutions of the General Meeting shall be adopted by a simple majority of the votes cast.
- 30.3 The chairperson of the General Meeting shall decide on the method of voting.
- 30.4 Abstentions, blank votes and invalid votes shall not be counted as votes.
- 30.5 The ruling by the chairperson of the General Meeting on the outcome of a vote shall be decisive.
- 30.6 All disputes concerning voting for which neither the law nor these articles of association provide a solution are decided by the chairperson of the General Meeting.
- 30.7 No votes may be cast at the General Meeting for a Share held by the Company or a Subsidiary, nor for any Share for which the Company or a Subsidiary holds the depositary receipts. The Company or a Subsidiary may not cast a vote in respect of a Share on which it holds a right of pledge or a usufruct. However, holders of a right of pledge or a usufruct on Shares held by the Company or a Subsidiary are not excluded from voting, if the right of pledge or the usufruct was created before the Share belonged to the Company or a Subsidiary.
- 30.8 When determining how many votes are cast by Shareholders, how many Shareholders are present or represented, or which part of the Company's issued capital is represented at the General Meeting, no account shall be taken of Shares for which, pursuant to the law or these articles of association, no vote can be cast.

31 Class meetings

- 31.1 The provisions of these articles of association with respect to General Meetings, save for article 25.2, shall apply *mutatis mutandis* to a meeting of holders of Class A Shares and other persons entitled to attend such meeting.
- 31.2 The provisions of these articles of association with respect to General Meetings, save for article 25.2 and article 28.5, shall apply *mutatis mutandis* to a meeting of holders of Class B Shares or a meeting of holders of Conversion Shares (as applicable) and other persons entitled to attend such meeting, provided that the applicable meeting shall appoint its own chairman, and furthermore provided that for as long as Class B Shares or Conversion Shares (as applicable) are not admitted to listing and trading on a stock exchange with the cooperation of the Company:
 - (a) notice of a meeting as referred to in this article shall be given no later than on the fifteenth day before the date of the meeting and the convocation notice shall be sent to the addresses as included in the shareholders' register;
 - (b) resolutions may be adopted in writing without holding a meeting as referred to in this article, provided such resolutions are adopted by the unanimous vote of all holders of Class B Shares entitled to vote or Conversion Shares entitled to vote (as applicable); and
 - (c) valid resolutions may be adopted if the formalities for convening and holding of meetings as referred to in this article have not been complied with, if adopted by unanimous vote in a meeting at which all issued and outstanding Class B Shares or Conversion Shares (as applicable) are represented.

CHAPTER IX - FINANCIAL YEAR, ANNUAL ACCOUNTS AND AUDITOR

32 Financial year and annual accounts

- 32.1 The Company's financial year shall be the calendar year.
- 32.2 Annually, within the term set by law, the Board shall prepare the Annual Accounts. The Annual Accounts must be accompanied by an auditor's statement as referred to in article 33.3, the Management Report, and the additional information to the extent that this information is required.
- 32.3 The Annual Accounts shall be signed by the Directors; if one or more of their signatures is lacking, this shall be stated, giving the reasons therefor.
- 32.4 The Annual Accounts shall be adopted by the General Meeting.

33 Auditor

- 33.1 The General Meeting shall instruct an Auditor to audit the Annual Accounts. If the General Meeting fails to issue the instructions to an Auditor, the Board shall be authorized to do so. The Executive Directors shall not participate in the deliberations and decision-making regarding instructing an Auditor to audit the Annual Accounts.
- 33.2 The instructions issued to the Auditor may only be revoked by the General Meeting and, if the Board issued the instructions, by the Board, for valid reasons and in accordance with Section 2:393(2) DCC.
- 33.3 The Auditor shall report the findings of the audit to the Board and present the results of the audit in a statement on the true and fair view provided by the Annual Accounts.

CHAPTER X - RESULT AND DISTRIBUTIONS

34 Profits and distributions

- 34.1 Distribution of profits shall be made after adoption of the Annual Accounts from which it appears that the same is permitted.
- 34.2 Distributions may be made only to the extent the Company's equity exceeds the sum of its paid up and called up part of its issued capital and the reserves which must be maintained pursuant to the law.
- 34.3 The Board may determine which part of the profits shall be reserved, with due observance of the Company's policy on reserves and dividends. The Company may have a policy on reserves and dividends to be determined and amended by the Board.
- 34.4 The General Meeting may resolve to distribute any part of the profits remaining after reservation in accordance with article 34.3. If the General Meeting does not resolve to distribute these profits in whole or in part, such profits (or any profits remaining after distribution) shall also be reserved.
- 34.5 The General Meeting may only resolve to distribute to the Shareholders a dividend in kind or in the form of Shares at the proposal of the Board.
- 34.6 The Board, or the General Meeting at the proposal of the Board, may resolve to make distributions from the share premium reserve or other distributable reserves maintained by the Company.
- 34.7 The Board may resolve to make interim distributions on Shares, provided that an interim statement of assets and liabilities drawn up in accordance with the statutory requirements shows that the requirement of article 34.2 has been fulfilled, and with observance of (other) applicable statutory provisions.
- 34.8 The Board, or the General Meeting at the proposal of the Board, may resolve that a distribution on Shares shall not be paid in whole or in part in cash but in kind or in the form of Shares, or decide that Shareholders shall be given the option to receive the distribution in cash or in kind and/or in the form of Shares (and with due observance of articles 7 and 8), and may determine the conditions under which such option can be given to the Shareholders.
- 34.9 In calculating the amount of any distribution on Shares, Shares held by the Company shall be disregarded, unless such Shares are encumbered with a usufruct or right of pledge.
- 34.10 Any and all distributions on the Class A Shares and Class B Shares shall be made in such a way that on each Class A Share and Class B Share an equal amount or value will be distributed, provided that and with observance of the following order of priority:
 - (a) in the event of a distribution of profits in respect of a financial year, the Conversion Share Distribution shall first be distributed on each issued and outstanding Conversion Share; and
 - (b) following the Conversion Share Distribution, no further distribution shall be made on Conversion Shares in respect of such financial year.

35 Notices and payments

- 35.1 The date on which dividends and other distributions shall be made payable shall be announced in accordance with the law and published on the Company's website.
- 35.2 Distributions shall be payable on the date determined by the Board.
- 35.3 The persons entitled to a distribution shall be the relevant Shareholders, holders of a usufruct on Shares and holders of a right of pledge on Shares, at a date to be determined by the Board for that purpose. This date shall not be earlier than the date on which the distribution was announced.
- 35.4 Distributions which have not been claimed upon the expiry of five years and one day after the date when they became payable will be forfeited to the Company and will be carried to the reserves.
- 35.5 The Board may determine that distributions on Shares will be made payable either in euro or in another currency.

CHAPTER XI - AMENDMENT OF THE ARTICLES OF ASSOCIATION, DISSOLUTION AND LIQUIDATION

36 Amendment of the articles of association

- 36.1 The General Meeting may resolve to amend these articles of association at the proposal of the Board.
- 36.2 If a proposal to amend these articles of association is to be submitted to the General Meeting, the notice of such meeting must state so and a copy of the proposal, including the verbatim text thereof, shall be deposited and kept available at the Company's office for inspection by, and must be made available free of charge to, Shareholders and other Persons with Meeting Right, until the conclusion of the meeting. An amendment of these articles of association shall be laid down in a notarial deed.

37 Dissolution and liquidation

- 37.1 The General Meeting may resolve to dissolve the Company at the proposal of the Board.
- 37.2 If the Company is dissolved pursuant to a resolution of the General Meeting, the Directors shall become liquidators of the dissolved Company's property. The General Meeting may decide to appoint other persons as liquidators.
- 37.3 During liquidation, to the extent possible these articles of association shall continue to apply.
- 37.4 The balance remaining after payment of the debts of the dissolved Company shall be transferred to the Shareholders as follows and in the following order of priority:
 - (a) an amount equal to the nominal value of Conversion Shares shall first be transferred on each Conversion Share to the holders of the Conversion Shares; and
 - (b) the balance remaining thereafter shall be transferred *pro rata* in proportion to the number of Class A Shares and Class B Shares held by each Shareholder.

38 Federal forum provision

The sole and exclusive forum for any complaint asserting a cause of action arising under the United States Securities Act of 1933, as amended, to the fullest extent permitted by applicable law, shall be the federal district courts of the United States of America. This shall not apply to causes of action arising under the United States Securities Exchange Act of 1934, as amended.

39 First financial year

The first financial year of the Company shall end on the thirty-first day of December two thousand twenty-one. This article and its heading shall cease to exist after the end of the first financial year.

40 Transitory provisions authorized capital

- 40.1 The provisions of articles 5.1 and 5.2 shall only come into effect if and when Shares are issued to parties holding shares in Wall Box Chargers, S.L. prior to the Closing Date (**Authorized Capital Condition Precedent**), which shall be evidenced by the filing by the Company with the Dutch trade register that states that the Authorized Capital Condition Precedent has been fulfilled.
- 40.2 Upon these articles of association taking effect and prior to the fulfilment of the Authorized Capital Condition Precedent, the articles 5.1 and 5.2 shall read as follows:
 - "5.1 The authorized capital of the Company equals sixty thousand one hundred twenty-two euro and sixteen eurocent (EUR 60,122.16).
 - 5.2 The authorized capital of the Company is divided into five hundred thousand (500,000) Class A Shares, with a nominal value of twelve eurocent (EUR 0.12) each, one hundred (100) Class B Shares, with a nominal value of one euro and twenty eurocent (EUR 1.20) each, and two (2) Conversion Shares, with a nominal value of one euro and eight eurocent (EUR 1.08) each."
- 40.3 This article and its heading shall cease to exist upon the fulfilment of the Authorized Capital Condition Precedent.

Final provisions

Finally, the person appearing declared:

prior to the subject amendment to the articles of association comes into effect, the issued and fully paid-up capital of the Company amounts to forty-five thousand euro (EUR 45,000), consisting of three hundred seventy-five thousand (375,000) shares, with a nominal value of twelve eurocent (EUR 0.12) each, numbered 1 up to and including 375,000 (Existing Shares);

through the subject amendment to the articles of association taking effect, the Existing Shares are converted into three hundred seventy-five thousand (375,000) class A shares, with a nominal value of twelve eurocent (EUR 0.12), numbered A-1 up to and including A-375,000; and

therefore as of the moment the subject amendment to the articles of association comes into effect, the issued and fully paid up capital of the Company amounts to forty-five thousand euro (EUR 45,000), consisting of three hundred seventy-five thousand (375,000) class A shares, with a nominal value of twelve eurocent (EUR 0.12) each, numbered A-1 up to and including A-375,000.

Accountant's Certificate

With respect to the conversion of the Company into a public limited liability company Joinson & Spice B.V., accountants in Amsterdam, the Netherlands, have prepared an accountant's certificate in accordance with Section 2:72 of the Dutch Civil Code, showing that on the twenty-eighth day of September two thousand twenty-one the value of the assets of the Company were at least equal to forty-five thousand euro (EUR 45,000). A copy of the accountant's certificate shall be attached to this deed (*Annex II*).

End

The person appearing is known to me, civil law notary.

This deed was executed in Amsterdam, the Netherlands, on the date stated in the first paragraph of this deed. The contents of the deed have been stated and clarified to the person appearing. The person appearing has declared not to wish the deed to be fully read out, to have noted the contents of the deed timely before its execution and to agree with the contents. After limited reading, this deed was signed first by the person appearing and thereafter by me, civil law notary.

WARRANT ASSIGNMENT, ASSUMPTION AND AMENDED & RESTATED AGREEMENT between

KENSINGTON CAPITAL ACQUISITION CORP. II

and

CONTINENTAL STOCK TRANSFER & TRUST COMPANY

THIS ASSIGNMENT, ASSUMPTION AND AMENDED & RESTATED WARRANT AGREEMENT (this "Agreement"), dated as of October 1, 2021 (the "Effective Date"), is by and between Kensington Capital Acquisition Corp. II ("KCAC"), Wallbox, N.V., a public company with limited liability (naamloze vennootschap) incorporated under the laws of the Netherlands, having its official seat in Amsterdam, the Netherlands and registered with the Dutch trade register under number 83012559 (the "Company"), and Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "Warrant Agent"), also referred to herein as the "Transfer Agent").

WHEREAS, KCAC and the Warrant Agent are parties to that certain Warrant Agreement, dated as of February 25, 2021 (the "Existing Warrant Agreement");

WHEREAS, in accordance with Section 9.8 of the Existing Warrant Agreement, KCAC and the Warrant Agent agree to amend and restate the Existing Warrant Agreement in its entirety as contemplated hereunder;

WHEREAS, on February 25, 2021, KCAC entered into that certain Private Placement Warrants Purchase Agreement with Kensington Capital Sponsor II, LLC, a Delaware limited liability company (the "Sponsor"), pursuant to which the Sponsor will purchase 8,000,000 redeemable warrants (or up to 8,800,000 redeemable warrants if the Over-allotment Option (as defined below) in connection with the Offering (as defined below) is exercised in full) simultaneously with the closing of the Offering (the "Private Placement Warrants") at a purchase price of \$0.75 per Private Placement Warrant; and

WHEREAS, in order to finance KCAC's transaction costs in connection with an intended initial Business Combination (as defined below), the Sponsor or an affiliate of the Sponsor or certain KCAC's officers and directors may, but are not obligated to, loan KCAC funds as KCAC required, of which up to \$2,000,000 of such loans were convertible into up to an additional 2,666,666 Private Placement Warrants at a price of \$0.75 per warrant (the "Working Capital Warrants"); and

WHEREAS, KCAC engaged in an initial public offering (the "Offering") of units of the KCAC's equity securities, each such unit comprised of one share of Common Stock (as defined below) and one-fourth of one Public Warrant (as defined below) (the "Units") and, in connection therewith, has determined to issue and deliver up to 5,000,000 warrants (or up to 5,750,000 warrants if the Over-allotment Option is exercised in full) to public investors in the Offering (the "Public Warrants" and, together with the Private Placement Warrants and the Working Capital Warrants, the "Warrants"). Each whole Warrant entitles the holder thereof to purchase one share of Class A common stock of KCAC, par value \$0.0001 per share ("Common Stock"), for \$11.50 per share, subject to adjustment as described herein;

WHEREAS, on June 9, 2021, the Company, Orion Merger Sub Corp., a Delaware corporation and a wholly-owned subsidiary of the Company ("Merger Sub"), Wall Box Chargers, S.L., a Spanish limited liability company (sociedad limitada) ("Wallbox") and KCAC entered into that certain Business Combination Agreement (the "BCA");

WHEREAS, upon the terms and subject to the conditions of the BCA, on the Effective Date (a) each holder of Wallbox securities will take steps to exchange by means of a contribution in kind its Wallbox securities in exchange for the issuance of shares of the Company, as a result of which Wallbox will become a wholly-owned subsidiary of the Company, (b) each holder of KCAC common stock will take steps to exchange by means of a contribution in kind its KCAC common stock in exchange for the issuance of shares of the Company and (c) Merger Sub will merge with and into KCAC (the "Merger"), with KCAC continuing as the surviving company after the Merger, as a result of which, KCAC will become a direct, wholly-owned subsidiary of the Company;

WHEREAS, upon consummation of the Merger, as provided in Section 4.4 of the Existing Warrant Agreement, (i) the Public Warrants and Private Placement Warrants issued thereunder will no longer be exercisable for Class A common stock, \$0.0001 par value per share, of KCAC ("KCAC Class A Shares") but instead will be exercisable (subject to the terms and conditions of the Existing Warrant Agreement as amended hereby) for a number of Class A ordinary shares of of the Company with a nominal value of 0.12 (the "Ordinary Shares") equal to the number of KCAC Class A Shares for which such warrants were exercisable immediately prior to the Merger subject to adjustment as described herein (such warrants as so adjusted and amended, the "Warrants") and (ii) the Warrants shall be assumed by the Company;

WHEREAS, in connection with the transactions contemplated by the BCA, KCAC desires to assign to the Company, and the Company's desires to assume, all of KCAC's rights, interests and obligations under the Existing Warrant Agreement;

WHEREAS, the consummation of the transactions contemplated by the BCA will constitute a Business Combination (as defined in Section 3.2 of the Existing Warrant Agreement);

WHEREAS, the Company desires the Warrant Agent to act on behalf of the Company, and the Warrant Agent is willing to so act, in connection with the issuance, registration, transfer, exchange, redemption and exercise of the Warrants;

WHEREAS, the Company desires to provide for the form and provisions of the Warrants, the terms upon which they shall be issued and exercised, and the respective rights, limitation of rights, and immunities of the Company, the Warrant Agent, and the holders of the Warrants; and

WHEREAS, all acts and things have been done and performed which are necessary to make the Warrants, when executed on behalf of the Company and countersigned by or on behalf of the Warrant Agent (if a physical certificate is issued), as provided herein, the valid, binding and legal obligations of the Company, and to authorize the execution and delivery of this Agreement.

NOW, THEREFORE, in consideration of the mutual agreements herein contained, the parties hereto agree as follows:

- 1. Assignment and Assumption; Amendment; Appointment of Warrant Agent.
- 1.1 KCAC hereby assigns to the Company all of KCAC's right, title and interest in and to the Existing Warrant Agreement and the Warrants (each as amended hereby) as of the Effective Time. The Company hereby assumes, and agrees to pay, perform, satisfy and discharge in full, as the same become due, all of KCAC's liabilities and obligations under the Existing Warrant Agreement and the Warrants (each as amended hereby) arising from and after the Effective Time.
- 1.2 KCAC and the Warrant Agent hereby amend the Existing Warrant Agreement, and the Warrants issued thereunder, in its entirety in the form of this Agreement as of the Effective Time.
- 1.3 The Company hereby confirms the appointment of the Warrant Agent to act as agent for the Company for the Warrants, and the Warrant Agent hereby accepts such appointment and agrees to perform the same in accordance with the terms and conditions set forth in this Agreement.
 - 2. Warrants
- 2.1 <u>Form of Warrant</u>. Each Warrant was issued in registered form only. All of the Public Warrants are represented by one or more book-entry certificates (each, a "*Book-Entry Warrant Certificate*").
- 2.2 Effect of Countersignature. If a physical certificate is issued, unless and until countersigned by the Warrant Agent pursuant to this Agreement, a Warrant represented by such physical certificate shall be invalid and of no effect and may not be exercised by the holder thereof.
 - 2.3 Registration.
- 2.3.1 Warrant Register. The Warrant Agent shall maintain books (the "Warrant Register"), for the registration of original issuance and the registration of transfer of the Warrants. Upon the initial issuance of the Warrants, the Warrant Agent shall issue and register the Warrants in the names of the respective holders thereof in such denominations and otherwise in accordance with instructions delivered to the Warrant Agent by the Company. Ownership of beneficial interests in the Public Warrants shall be shown on, and the transfer of such ownership shall be effected through, records maintained by institutions that have accounts with the Depository Trust Company (the "Depositary") (such institution, with respect to a Warrant in its account, a "Participant").

If the Depositary subsequently ceases to make its book-entry settlement system available for the Public Warrants, the Company may instruct the Warrant Agent regarding making other arrangements for book-entry settlement. In its sole discretion, the Company may instruct the Warrant Agent to deliver to the Depositary (i) written instructions to deliver to the Warrant Agent for cancellation each Book-Entry Public Warrant and (ii) definitive certificates in physical form evidencing such Warrants (each, a "Definitive Warrant Certificate") which shall be in the form annexed hereto as Exhibit A.

Physical certificates, if issued, shall be signed by, or bear the facsimile signature of, the Chairman of the Board, Chief Executive Officer, Chief Financial Officer, Secretary or other principal officer of the Company. In the event the person whose facsimile signature has been placed upon any Warrant shall have ceased to serve in the capacity in which such person signed the Warrant before such Warrant is issued, it may be issued with the same effect as if he or she had not ceased to be such at the date of issuance.

- 2.3.2 Registered Holder. Prior to due presentment for registration of transfer of any Warrant, the Company and the Warrant Agent may deem and treat the person in whose name such Warrant is registered in the Warrant Register (the "Registered Holder") as the absolute owner of such Warrant and of each Warrant represented thereby (notwithstanding any notation of ownership or other writing on any physical Certificate made by anyone other than the Company or the Warrant Agent), for the purpose of any exercise thereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.
 - 2.4 [Reserved]
- 2.5 No Fractional Warrants. The Company shall not issue fractional Warrants. If a holder of Warrants would be entitled to receive a fractional Warrant, the Company shall round down to the nearest whole number the number of Warrants to be issued to such holder.
- 2.6 Private Placement and Working Capital Warrants. The Private Placement Warrants and Working Capital Warrants shall be identical to the Public Warrants, except that so long as they are held by the Sponsor or initial lender, as applicable or any of their respective Permitted Transferees (as defined below), the Private Placement Warrants and Working Capital Warrants: (i) may be exercised for cash or on a cashless basis, pursuant to subsection 3.3.1(c) hereof, (ii) may not be transferred, assigned or sold until thirty (30) days after the completion by the Company of an initial Business Combination (as defined below), and (iii) shall not be redeemable by the Company pursuant to Section 6.1 hereof; provided, however, that in the case of (ii), the Private Placement Warrants, Working Capital Warrants and any Ordinary Shares held by the Sponsor or any of its Permitted Transferees and issued upon exercise of the Private Placement Warrants or Working Capital Warrants may be transferred by the holders thereof:
 - (a) to KCAC's officers or directors, any affiliates or family members of any of KCAC's officers or directors, any members of the Sponsor or any affiliates of the members of the Sponsor, any affiliates of the Sponsor or any employees of such affiliates;

- (b) in the case of an individual, transfers by gift to a member of the individual's immediate family, to a trust, the beneficiary of which is a member of the individual's immediate family or an affiliate of such person, or to a charitable organization;
 - (c) in the case of an individual, transfers by virtue of laws of descent and distribution upon death of the individual;
 - (d) in the case of an individual, transfers pursuant to a qualified domestic relations order;
- (e) by virtue of the laws of the State of Delaware or the Sponsor's limited liability company agreement upon dissolution of the Sponsor; or
- (f) in the event of the Company's liquidation, merger, capital stock exchange, reorganization or other similar transaction which results in all of the Company's stockholders having the right to exchange their Ordinary Shares for cash, securities or other property; provided, however, that in the case of clauses (a) through (e), these permitted transferees (the "Permitted Transferees") must enter into a written agreement agreeing to be bound by the transfer restrictions in this Agreement and the other restrictions contained in the Registration Rights and Lock-up Agreement, dated as of the date hereof, by and among the Company and the other parties thereto (the "Registration Rights Agreement") (including, for the avoidance of doubt, the provisions with respect to the Lock-Up Period, as defined therein with respect to such transferees) and provided, further, that any transfers under clauses (a) through (e) shall be subject to the Registration Rights Agreement and made only to the extent permitted under the Original Holder Lock-Up Period, as defined therein with respect to such transferees.
- 2.7 Working Capital Warrants. Each of the Working Capital Warrants shall be identical to the Private Placement Warrants.
- 3. Terms and Exercise of Warrants.
- 3.1 Warrant Price. Each Warrant shall, when countersigned by the Warrant Agent (if a physical certificate is issued), entitle the Registered Holder thereof, subject to the provisions of such Warrant and of this Agreement, to purchase from the Company the number of Ordinary Shares stated therein, at the price of \$11.50 per share, subject to the adjustments provided in Section 4 hereof and in the last sentence of this Section 3.1. The term "Warrant Price" as used in this Agreement shall mean the price per share at which the Ordinary Shares may be purchased at the time a Warrant is exercised. The Company in its sole discretion may lower the Warrant Price at any time prior to the Expiration Date (as defined below) for a period of not less than twenty (20) Business Days, provided, that the Company shall provide at least twenty (20) days prior written notice of such reduction to Registered Holders of the Warrants and, provided further that any such reduction shall be identical among all of the Warrants.

3.2 <u>Duration of Warrants</u>. A Warrant may be exercised only during the period (the "*Exercise Period*") commencing on the date that is thirty (30) days after the Effective Date, and terminating at 5:00 p.m., New York City time on the earliest to occur of: (x) the date that is five (5) years after the Effective Date and (y) other than with respect to the Private Placement Warrants and Working Capital Warrants then held by the Sponsor or initial lender, as applicable, or any of their respective Permitted Transferees in connection with a redemption pursuant to <u>Section 6.1</u> hereof, the Redemption Date (as defined below) as provided in <u>Section 6.3</u> hereof (the "*Expiration Date*"); provided, however, that the exercise of any Warrant shall be subject to the satisfaction of any applicable conditions, as set forth in <u>subsection 3.3.2</u> below with respect to an effective registration statement. Except with respect to the right to receive the Redemption Price (as defined below) (other than with respect to a Private Placement Warrant or a Working Capital Warrant held by the Sponsor or initial lender, as applicable, or their respective Permitted Transferees, in connection with a redemption pursuant to <u>Section 6.1 hereof</u>) in the event of a redemption (as set forth in <u>Section 6</u> hereof), each Warrant (other than a Private Placement Warrant or a Working Capital Warrant held by the Sponsor or initial lender, or their respective Permitted Transferees, in the event of a redemption pursuant to <u>Section 6.1</u> hereof) not exercised on or before the Expiration Date shall become void, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at 5:00 p.m. New York City time on the Expiration Date. The Company in its sole discretion may extend the duration of the Warrants by delaying the Expiration Date; <u>provided</u>, that the Company shall provide at least twenty (20) days prior written notice of any such extension to Registered Holders of the Warrants and, provided further that any such extension

3.3 Exercise of Warrants.

3.3.1 Payment. Subject to the provisions of the Warrant and this Agreement, a Warrant may be exercised by the Registered Holder thereof by delivering to the Warrant Agent, or at the office of its successor as Warrant Agent, in the Borough of Manhattan, City and State of New York (i) the Definitive Warrant Certificate evidencing the Warrants to be exercised, or, in the case of a Book-Entry Warrant Certificate, the Warrants to be exercised on the records of the Depositary to an account of the Warrant Agent at the Depositary designated for such purposes in writing by the Warrant Agent to the Depositary from time to time, (ii) an election to purchase Ordinary Shares pursuant to the exercise of a Warrant, properly completed and executed by the Registered Holder on the reverse of the Definitive Warrant Certificate or, in the case of a Book-Entry Warrant Certificate, properly delivered by the Participant in accordance with the Depositary's procedures, and (iii) payment in full of the Warrant Price for each Ordinary Share as to which the Warrant is exercised and any and all applicable taxes due in connection with the exercise of the Warrant, the exchange of the Warrant for the Ordinary Shares and the issuance of such Ordinary Shares, as follows:

(a) in lawful money of the United States, in good certified check or good bank draft payable to the Warrant Agent or by wire transfer of immediately available funds;

- (b) in the event of a redemption pursuant to <u>Section 6</u> hereof in which the Company's Board of directors (the "**Board**") has elected to require all holders of the Warrants to exercise such Warrants on a "cashless basis," by surrendering the Warrants for that number of Ordinary Shares equal to the quotient obtained by dividing (x) the product of the number of Ordinary Shares underlying the Warrants, multiplied by the excess of the "Fair Market Value", as defined in this <u>subsection 3.3.1(b)</u>, over the Warrant Price by (y) the Fair Market Value. Solely for purposes of this <u>subsection 3.3.1(b)</u>, <u>Section 6.2</u> and <u>Section 6.4</u>, the "Fair Market Value" shall mean the average last reported sale price of the Ordinary Shares for the ten (10) trading days ending on the third trading day prior to the date on which the notice of redemption is sent to the holders of the Warrants, pursuant to <u>Section 6</u> hereof;
- (c) with respect to any Private Placement Warrant or Working Capital Warrant, so long as such Private Placement Warrant or Working Capital Warrant is held by the Sponsor or initial lender, as applicable, or their respective Permitted Transferees, by surrendering the Warrants for that number of Ordinary Shares equal to the quotient obtained by dividing (x) the product of the number of Ordinary Shares underlying the Warrants, multiplied by the excess of the "Sponsor Fair Market Value", as defined in this <u>subsection 3.3.1(c)</u>, over the Warrant Price by (y) the Sponsor Fair Market Value. Solely for purposes of this <u>subsection 3.3.1(c)</u>, the "*Sponsor Fair Market Value*" shall mean the average last reported sale price of the Ordinary Shares for the ten (10) trading days ending on the third trading day prior to the date on which notice of exercise of the Private Placement Warrant or Working Capital Warrant is sent to the Warrant Agent;
 - (d) as provided in Section 6.2 with respect to a Make-Whole Exercise; or
 - (e) as provided in Section 7.4 hereof.
- 3.3.2 <u>Issuance of Ordinary Shares on Exercise</u>. As soon as practicable after the exercise of any Warrant and the clearance of the funds in payment of the Warrant Price (if payment is pursuant to <u>subsection 3.3.1(a)</u>), the Company shall issue to the Registered Holder of such Warrant a bookentry position for the number of Ordinary Shares to which he, she or it is entitled, registered in such name or names as may be directed by him, her or it, and if such Warrant shall not have been exercised in full, a new book-entry position or countersigned Warrant, as applicable, for the number of Ordinary Shares as to which such Warrant shall not have been exercised. Notwithstanding the foregoing, the Company shall not be obligated to deliver any Ordinary Shares pursuant to the exercise of a Warrant and shall have no obligation to settle such Warrant exercise unless a registration statement under the Securities Act of 1933, as amended (the "Securities Act") with respect to the Ordinary Shares underlying the Public Warrants is then effective and a prospectus relating thereto is current, subject to the Company's satisfying its obligations under Section 7.4, or a valid exemption from registration is available. No Warrant shall be exercisable and the Company shall not be obligated to issue Ordinary Shares upon exercise of a Warrant unless the Ordinary Shares issuable upon such Warrant exercise have been registered, qualified or deemed to be exempt under the securities laws of the state of residence of the Registered Holder of the Warrants. The Company may require holders of Public Warrants to settle the Warrant on a "cashless basis" pursuant to Section 7.4. If, by reason of any exercise of Warrants on a "cashless basis", the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in an Ordinary Share, the Company shall round down to the nearest whole number, the number of Ordinary Shares to be issued to such holder.

- 3.3.3 <u>Valid Issuance</u>. All Ordinary Shares issued upon the proper exercise of a Warrant in conformity with this Agreement shall be validly issued, fully paid and non-assessable.
- 3.3.4 <u>Date of Issuance</u>. Each person in whose name any book-entry position for Ordinary Shares is issued shall for all purposes be deemed to have become the holder such Ordinary Shares on the date on which the Warrant, or book-entry position representing such Warrant, was surrendered and payment of the Warrant Price was made, irrespective of the date of delivery of such certificate in the case of a certificated Warrant, except that, if the date of such surrender and payment is a date when the share transfer books of the Company or book-entry system of the Warrant Agent are closed, such person shall be deemed to have become the holder of such Ordinary Shares at the close of business on the next succeeding date on which the share transfer books or book-entry system are open.
- 3.3.5 Maximum Percentage. A holder of a Warrant may notify the Company in writing in the event it elects to be subject to the provisions contained in this subsection 3.3.5; however, no holder of a Warrant shall be subject to this subsection 3.3.5 unless he, she or it makes such election. If the election is made by a holder, the Warrant Agent shall assist with the exercise of the holder's Warrant, and such holder shall not have the right to exercise such Warrant, to the extent that after giving effect to such exercise, such person (together with such person's affiliates), to the Warrant Agent's actual knowledge, would beneficially own in excess of 4.8% or 9.8% (or such other amount as a holder may specify)(the "Maximum Percentage") of the Ordinary Shares outstanding immediately after giving effect to such exercise. For purposes of the foregoing sentence, the aggregate number of Ordinary Shares beneficially owned by such person and its affiliates shall include the number of Ordinary Shares issuable upon exercise of the Warrant with respect to which the determination of such sentence is being made, but shall exclude Ordinary Shares that would be issuable upon (x) exercise of the remaining, unexercised portion of the Warrant beneficially owned by such person and its affiliates and (y) exercise or conversion of the unexercised or unconverted portion of any other securities of the Company beneficially owned by such person and its affiliates (including, without limitation, any convertible notes or convertible preferred shares or warrants) subject to a limitation on conversion or exercise analogous to the limitation contained herein. Except as set forth in the preceding sentence, for purposes of this paragraph, beneficial ownership shall be calculated in accordance with Section 13(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). For purposes of the Warrant, in determining the number of outstanding Ordinary Shares, the holder may rely on the number of outstanding Ordinary Shares as reflected in (1) the Company's most recent Annual Report on Form 10-K, Quarterly Report on Form 10-Q, Current Report on Form 8-K or other public filing with the Securities and Exchange Commission (the "Commission") as the case may be, (2) a more recent public announcement by the Company or (3) any other notice by the Company or the Transfer Agent setting forth the number of Ordinary Shares outstanding. For any reason at any time, upon the written request of the holder of the Warrant, the Company shall, within two (2) Business Days, confirm orally and in writing to such holder the number of Ordinary Shares then outstanding. In any case, the number of outstanding Ordinary Shares shall be determined after giving effect to the conversion

or exercise of equity securities of the Company by the holder and its affiliates since the date as of which such number of outstanding Ordinary Shares was reported. By written notice to the Company, the holder of a Warrant may from time to time increase or decrease the Maximum Percentage applicable to such holder to any other percentage specified in such notice; <u>provided, however</u>, that any such increase shall not be effective until the sixty-first (61st) day after such notice is delivered to the Company.

- 4. Adjustments.
- 4.1 Stock Dividends.
- 4.1.1 Split-Ups. If after the date hereof, and subject to the provisions of Section 4.6 below, the number of outstanding Ordinary Shares is increased by a stock dividend payable in Ordinary Shares, stock split or reclassification of the Ordinary Shares or other similar event, then, on the effective date of such stock dividend, split-up or similar event, the number of Ordinary Shares issuable on exercise of each Warrant shall be increased in proportion to such increase in the outstanding Ordinary Shares. A rights offering to holders of the Ordinary Shares entitling holders to purchase Ordinary Shares at a price less than the "Fair Market Value" (as defined below) shall be deemed a stock dividend of a number of Ordinary Shares equal to the product of (i) the number of Ordinary Shares actually sold in such rights offering (or issuable under any other equity securities sold in such rights offering that are convertible into or exercisable for Ordinary Shares) multiplied by (ii) one (1) minus the quotient of (x) the price per Ordinary Share paid in such rights offering divided by (y) the Fair Market Value. For purposes of this subsection 4.1.1, (i) if the rights offering is for securities convertible into or exercisable for Ordinary Shares, in determining the price payable for Ordinary Shares, there shall be taken into account any consideration received for such rights, as well as any additional amount payable upon exercise or conversion and (ii) "Fair Market Value" means the volume weighted average price of the Ordinary Shares as reported during the ten (10) trading day period ending on the trading day prior to the first date on which the Ordinary Shares trade on the applicable exchange or in the applicable market, regular way, without the right to receive such rights.
- 4.1.2 Extraordinary Dividends. If the Company, at any time while the Warrants are outstanding and unexpired, shall pay a dividend or make a distribution in cash, securities or other assets to the holders of Ordinary Shares on account of such Ordinary Shares (or other of the Company's share capital into which the Warrants are convertible), other than (a) as described in subsection 4.1.1 above or (b) Ordinary Cash Dividends (as defined below) (any such non-excluded event being referred to herein as an "Extraordinary Dividend"), then the Warrant Price shall be decreased, effective immediately after the effective date of such Extraordinary Dividend, by the amount of cash and/or the fair market value (as determined by the Board, in good faith) of any securities or other assets paid on each Ordinary Share in respect of such Extraordinary Dividend. For purposes of this subsection 4.1.2, "Ordinary Cash Dividends" means any cash dividend or cash distribution which, when combined on a per share basis, with the per share amounts of all other cash dividends and cash distributions paid on the Ordinary Shares during the 365-day period ending on the date of declaration of such dividend or distribution to the extent it does not exceed \$0.50 (being 5% of the offering price of the Units in the Offering) (as adjusted to appropriately reflect any of the events referred to in other subsections of this Section 4 and

excluding cash dividends or cash distributions that resulted in an adjustment to the Warrant Price or to the number of Ordinary Shares issuable on exercise of each Warrant) but only with respect to the amount of the aggregate cash dividends or cash distributions equal to or less than \$0.50. Solely for purposes of illustration, if the Company, at a time while the Warrants are outstanding and unexpired, pays a cash dividend of \$0.35 per share and previously paid an aggregate of \$0.40 of cash dividends and cash distributions on the Ordinary Shares during the 365-day period ending on the date of declaration of such \$0.35 per share dividend, then the Warrant Price will be decreased, effectively immediately after the effective date of such \$0.35 per share dividend, by \$0.25 (the absolute value of the difference between \$0.75 per share (the aggregate amount of all cash dividends and cash distributions paid or made in such 365-day period, including such \$0.35 dividend) and \$0.50 per share (the greater of (x) \$0.50 per share and (y) the aggregate amount of all cash dividends and cash distributions paid or made in such 365-day period prior to such \$0.35 dividend)).

- 4.2 <u>Aggregation of Shares</u>. If after the date hereof, and subject to the provisions of <u>Section 4.6</u> hereof, the number of outstanding Ordinary Shares is decreased by a consolidation, combination or reclassification of Ordinary Shares or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the number of Ordinary Shares issuable on exercise of each Warrant shall be decreased in proportion to such decrease in outstanding Ordinary Shares.
- 4.3 Adjustments in Exercise Price. Whenever the number of Ordinary Shares purchasable upon the exercise of the Warrants is adjusted, as provided in subsection 4.1.1 or Section 4.2 above, the Warrant Price shall be adjusted (to the nearest cent) by multiplying such Warrant Price immediately prior to such adjustment by a fraction (x) the numerator of which shall be the number of Ordinary Shares purchasable upon the exercise of the Warrants immediately prior to such adjustment, and (y) the denominator of which shall be the number of Ordinary Shares so purchasable immediately thereafter.
- 4.4 Replacement of Securities upon Reorganization, etc. In case of any reclassification or reorganization of the outstanding Ordinary Shares (other than a change under subsections 4.1.1 or 4.1.2 or Section 4.2 hereof or that solely affects the par value of such Ordinary Shares), or in the case of any merger or consolidation of the Company with or into another corporation (other than a consolidation or merger in which the Company is the continuing corporation and that does not result in any reclassification or reorganization of the outstanding Ordinary Shares), or in the case of any sale or conveyance to another corporation or entity of the assets or other property of the Company as an entirety or substantially as an entirety in connection with which the Company is dissolved, the holders of the Warrants shall thereafter have the right to purchase and receive, upon the basis and upon the terms and conditions specified in the Warrants and in lieu of the Ordinary Shares of the Company immediately theretofore purchasable and receivable upon the exercise of the rights represented thereby, the kind and amount of shares or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, that the holder of the Warrants would have received if such holder had exercised his, her or its Warrant(s) immediately prior to such event (the "Alternative Issuance");

provided, however, that (i) if the holders of the Ordinary Shares were entitled to exercise a right of election as to the kind or amount of securities, cash or other assets receivable upon such consolidation or merger, then the kind and amount of securities, cash or other assets constituting the Alternative Issuance for which each Warrant shall become exercisable shall be deemed to be the weighted average of the kind and amount received per share by the holders of the Ordinary Shares in such consolidation or merger that affirmatively make such election, and (ii) if a tender, exchange or redemption offer shall have been made to and accepted by the holders of the Ordinary Shares under circumstances in which, upon completion of such tender or exchange offer, the maker thereof, together with members of any group (within the meaning of Rule 13d-5(b)(1) under the Exchange Act (or any successor rule)) of which such maker is a part, and together with any affiliate or associate of such maker (within the meaning of Rule 12b-2 under the Exchange Act (or any successor rule)) and any members of any such group of which any such affiliate or associate is a part, own beneficially (within the meaning of Rule 13d-3 under the Exchange Act (or any successor rule)) more than 50% of the outstanding Ordinary Shares, the holder of a Warrant shall be entitled to receive as the Alternative Issuance, the highest amount of cash, securities or other property to which such holder would actually have been entitled as a shareholder if such Warrant holder had exercised the Warrant prior to the expiration of such tender or exchange offer, accepted such offer and all of the Ordinary Shares held by such holder had been purchased pursuant to such tender or exchange offer, subject to adjustments (from and after the consummation of such tender or exchange offer) as nearly equivalent as possible to the adjustments provided for in this Section 4; provided, further, however, that if less than 70% of the consideration receivable by the holders of the Ordinary Shares in the applicable event is payable in the form of capital stock or shares in the successor entity that is listed for trading on a national securities exchange or is quoted in an established over-the-counter market, or is to be so listed for trading or quoted immediately following such event, and if the Registered Holder properly exercises the Warrant within thirty (30) days following the public disclosure of the consummation of such applicable event by the Company pursuant to a Current Report on Form 8-K filed with the Commission, the Warrant Price shall be reduced by an amount (in dollars) (but in no event less than zero) equal to the difference of (i) the Warrant Price in effect prior to such reduction minus (ii) (A) the Per Share Consideration (as defined below) minus (B) the Black-Scholes Warrant Value (as defined below). The "Black-Scholes Warrant Value" means the value of a Warrant immediately prior to the consummation of the applicable event based on the Black-Scholes Warrant Model for a Capped American Call on Bloomberg Financial Markets ("Bloomberg"). For purposes of calculating such amount, (1) Section 6 of this Agreement shall be taken into account, (2) the price of each Ordinary Share shall be the volume weighted average price of the Ordinary Shares as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event, (3) the assumed volatility shall be the 90 day volatility obtained from the HVT function on Bloomberg determined as of the trading day immediately prior to the day of the announcement of the applicable event, and (4) the assumed risk-free interest rate shall correspond to the U.S. Treasury rate for a period equal to the remaining term of the Warrant. "Per Share Consideration" means (i) if the consideration paid to holders of the Ordinary Shares consists exclusively of cash, the amount of such cash per Ordinary Share, and (ii) in all other cases, the volume weighted average price of the Ordinary Shares as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event. If any reclassification or reorganization also results in a change in Ordinary Shares covered by subsection 4.1.1, then such adjustment shall be made pursuant to subsection 4.1.1 or Sections 4.2, 4.3 and this Section 4.4. The provisions of this Section 4.4 shall similarly apply to successive reclassifications, reorganizations, mergers or consolidations, sales or other transfers. In no event will the Warrant Price be reduced to less than the par value per share issuable upon exercise of the Warrant.

- 4.5 Notices of Changes in Warrant. Upon every adjustment of the Warrant Price or the number of Ordinary Shares issuable upon exercise of a Warrant, the Company shall give written notice thereof to the Warrant Agent, which notice shall state the Warrant Price resulting from such adjustment and the increase or decrease, if any, in the number of Ordinary Shares purchasable at such price upon the exercise of a Warrant, setting forth in reasonable detail the method of calculation and the facts upon which such calculation is based. Upon the occurrence of any event specified in Sections 4.1, 4.2, 4.3 or 4.4, the Company shall give written notice of the occurrence of such event to each holder of a Warrant, at the last address set forth for such holder in the Warrant Register, of the record date or the effective date of the event. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such event.
- 4.6 No Fractional Shares. Notwithstanding any provision contained in this Agreement to the contrary, the Company shall not issue fractional Ordinary Shares upon the exercise of Warrants. If, by reason of any adjustment made pursuant to this Section 4, the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share, the Company shall, upon such exercise, round down to the nearest whole number of Ordinary Shares to be issued to such holder.]
- 4.7 Form of Warrant. The form of Warrant need not be changed because of any adjustment pursuant to this Section 4, and Warrants issued after such adjustment may state the same Warrant Price and the same number of Ordinary Shares as is stated in the Warrants initially issued pursuant to this Agreement; provided, however, that the Company may at any time in its sole discretion make any change in the form of Warrant that the Company may deem appropriate and that does not affect the substance thereof, and any Warrant thereafter issued or countersigned, whether in exchange or substitution for an outstanding Warrant or otherwise, may be in the form as so changed.
- 4.8 Other Events. In case any event shall occur affecting the Company as to which none of the provisions of preceding subsections of this Section 4 are strictly applicable, but which would require an adjustment to the terms of the Warrants in order to (i) avoid an adverse impact on the Warrants and (ii) effectuate the intent and purpose of this Section 4, then, in each such case, the Company shall appoint a firm of independent public accountants, investment banking or other appraisal firm of recognized national standing, which shall give its opinion as to whether or not any adjustment to the rights represented by the Warrants is necessary to effectuate the intent and purpose of this Section 4 and, if they determine that an adjustment is necessary, the terms of such adjustment. The Company shall adjust the terms of the Warrants in a manner that is consistent with any adjustment recommended in such opinion.

- 5. Transfer and Exchange of Warrants.
- 5.1 <u>Registration of Transfer</u>. The Warrant Agent shall register the transfer, from time to time, of any outstanding Warrant upon the Warrant Register, upon surrender of such Warrant for transfer, in the case of certificated warrants, properly endorsed with signatures properly guaranteed and accompanied by appropriate instructions for transfer. Upon any such transfer, a new Warrant representing an equal aggregate number of Warrants shall be issued and the old Warrant shall be cancelled by the Warrant Agent. In the case of certificated Warrants, the Warrants so cancelled shall be delivered by the Warrant Agent to the Company from time to time upon request.
- 5.2 <u>Procedure for Surrender of Warrants</u>. Warrants may be surrendered to the Warrant Agent, together with a written request for exchange or transfer, and thereupon the Warrant Agent shall issue in exchange therefor one or more new Warrants as requested by the Registered Holder of the Warrants so surrendered, representing an equal aggregate number of Warrants; <u>provided</u>, <u>however</u>, that in the event that a Warrant surrendered for transfer bears a restrictive legend (as in the case of the Private Placement Warrants and Working Capital Warrants), the Warrant Agent shall not cancel such Warrant and issue new Warrants in exchange thereof until the Warrant Agent has received an opinion of counsel for the Company stating that such transfer may be made and indicating whether the new Warrants must also bear a restrictive legend.
- 5.3 <u>Fractional Warrants</u>. The Warrant Agent shall not be required to effect any registration of transfer or exchange which shall result in the issuance of a warrant certificate or book entry position for a fraction of a warrant.
 - 5.4 Service Charges. No service charge shall be made for any exchange or registration of transfer of Warrants.
- 5.5 Warrant Execution and Countersignature. If a physical certificate is issued, The Warrant Agent is hereby authorized to countersign and to deliver, in accordance with the terms of this Agreement, the Warrants required to be issued pursuant to the provisions of this Section 5, and the Company, whenever required by the Warrant Agent, shall supply the Warrant Agent with Warrants duly executed on behalf of the Company for such purpose.
 - 6. Redemption.
- 6.1 Redemption of Warrants for when the price per share equals or exceeds \$18.00. Subject to Section 6.5 hereof, not less than all of the outstanding Warrants may be redeemed, at the option of the Company, at any time while they are exercisable and prior to their expiration, at the office of the Warrant Agent, upon notice to the Registered Holders of the Warrants, as described in Section 6.3 below, at the price (the "Redemption Price") of \$0.01 per Warrant, provided that the last reported sales price of the Reference Value equals or exceeds \$18.00 per share (subject to adjustment in compliance with Section 4 hereof), on each of twenty (20) trading days within the thirty (30) trading-day period ending on the third Business Day prior to the date on which notice of the redemption is given and provided that there is an effective registration statement covering the Ordinary Shares issuable upon exercise of the Warrants, and a current prospectus relating thereto, available throughout the 30-day Redemption Period (as defined in Section 6.3 below) or the Company has elected to require the exercise of the Warrants on a "cashless basis" pursuant to subsection 3.3.1.

6.2 Redemption of Warrants when the price per share equals or exceeds \$10.00. Not less than all of the outstanding Warrants may be redeemed, at the option of the Company, commencing once they are first exercisable and prior to their expiration, at the office of the Warrant Agent, upon notice to the Registered Holders of the Warrants, as described in Section 6.3 below, at a Redemption Price of \$0.10 per Warrant, provided that the last reported sales price of the Ordinary Shares reported has been at least \$10.00 per share (subject to adjustment in compliance with Section 4 hereof), on the trading day prior to the date on which notice of the redemption is given, provided that the Private Placement warrants are also concurrently called for redemption at the same price and terms as the outstanding Public Warrants, and provided that there is an effective registration statement covering the issuance of the Ordinary Shares issuable upon exercise of the Warrants, and a current prospectus relating thereto, available throughout the 30-day Redemption Period (as defined in Section 6.3 below). During the Redemption Period in connection with a redemption pursuant to this Section 6.2, Registered Holders of the Warrants may elect to exercise their Warrants on a "cashless basis" pursuant to subsection 3.3.1 and receive a number of Ordinary Shares determined by reference to the table below, based on the Redemption Date (calculated for purposes of the table as the period to expiration of the Warrants) and the "Fair Market Value" (as such term is defined in subsection 3.3.1(b)) (a "Make-Whole Exercise").

Redemption Date	Fair Market Value of Class A Common Stock								
(period to expiration of warrants)	≤\$10.00	\$11.00	\$12.00	\$13.00	\$14.00	\$15.00	\$16.00	\$17.00	≥\$18.00
60 months	0.261	0.281	0.297	0.311	0.324	0.337	0.348	0.358	0.361
57 months	0.257	0.277	0.294	0.310	0.324	0.337	0.348	0.358	0.361
54 months	0.252	0.272	0.291	0.307	0.322	0.335	0.347	0.357	0.361
51 months	0.246	0.268	0.287	0.304	0.320	0.333	0.346	0.357	0.361
48 months	0.241	0.263	0.283	0.301	0.317	0.332	0.344	0.356	0.361
45 months	0.235	0.258	0.279	0.298	0.315	0.330	0.343	0.356	0.361
42 months	0.228	0.252	0.274	0.294	0.312	0.328	0.342	0.355	0.361
39 months	0.221	0.246	0.269	0.290	0.309	0.325	0.340	0.354	0.361
36 months	0.213	0.239	0.263	0.285	0.305	0.323	0.339	0.353	0.361
33 months	0.205	0.232	0.257	0.280	0.301	0.320	0.337	0.352	0.361
30 months	0.196	0.224	0.250	0.274	0.297	0.316	0.335	0.351	0.361
27 months	0.185	0.214	0.242	0.268	0.291	0.313	0.332	0.350	0.361
24 months	0.173	0.204	0.233	0.260	0.285	0.308	0.329	0.348	0.361
21 months	0.161	0.193	0.223	0.252	0.279	0.304	0.326	0.347	0.361
18 months	0.146	0.179	0.211	0.242	0.271	0.298	0.322	0.345	0.361
15 months	0.130	0.164	0.197	0.230	0.262	0.291	0.317	0.342	0.361
12 months	0.111	0.146	0.181	0.216	0.250	0.282	0.312	0.339	0.361
9 months	0.090	0.125	0.162	0.199	0.237	0.272	0.305	0.336	0.361
6 months	0.065	0.099	0.137	0.178	0.219	0.259	0.296	0.331	0.361
3 months	0.034	0.065	0.104	0.150	0.197	0.243	0.286	0.326	0.361
0 months	-	-	0.042	0.115	0.179	0.233	0.281	0.323	0.361

The exact Fair Market Value and Redemption Date (as defined below) may not be set forth in the table above, in which case, if the Fair Market Value is between two values in the table or the Redemption Date is between two redemption dates in the table, the number of Ordinary Shares to be issued for each Warrant exercised in a Make-Whole Exercise will be determined by a straight-line interpolation between the number of shares set forth for the higher and lower Fair Market Values and the earlier and later redemption dates, as applicable, based on a 365- or 366-day year, as applicable.

The share prices set forth in the column headings of the table above shall be adjusted as of any date on which the number of shares issuable upon exercise of a Warrant is adjusted pursuant to Section 4. The adjusted share prices in the column headings shall equal the share prices immediately prior to such adjustment, multiplied by a fraction, the numerator of which is the number of shares deliverable upon exercise of a Warrant immediately prior to such adjustment and the denominator of which is the number of shares deliverable upon exercise of a Warrant as so adjusted. The number of shares in the table above shall be adjusted in the same manner and at the same time as the number of shares issuable upon exercise of a Warrant. In no event will the number of shares issued in connection with a Make-Whole Exercise exceed 0.361 Ordinary Shares per Warrant (subject to adjustment).

- 6.3 <u>Date Fixed for, and Notice of, Redemption</u>. In the event that the Company elects to redeem all of the Warrants pursuant to <u>Section 6.1</u> or <u>6.2</u>, the Company shall fix a date for the redemption (the "*Redemption Date*"). Notice of redemption shall be mailed by first class mail, postage prepaid, by the Company not less than thirty (30) days prior to the Redemption Date (the "*30-day Redemption Period*") to the Registered Holders of the Warrants to be redeemed at their last addresses as they shall appear on the registration books. Any notice mailed in the manner herein provided shall be conclusively presumed to have been duly given whether or not the Registered Holder received such notice.
- 6.4 Exercise After Notice of Redemption. The Warrants may be exercised, for cash (or on a "cashless basis" in accordance with <u>subsection 3.3.1(b)</u> or <u>Section 6.2</u> of this Agreement) at any time after notice of redemption shall have been given by the Company pursuant to <u>Section 6.3</u> hereof and prior to the Redemption Date. In the event that the Company determines to require all holders of Warrants to exercise their Warrants on a "cashless basis" pursuant to <u>subsection 3.3.1</u>, the notice of redemption shall contain the information necessary to calculate the number of shares of Common Stock to be received upon exercise of the Warrants, including the "Fair Market Value" (as such term is defined in <u>subsection 3.3.1(b)</u> hereof) in such case. On and after the Redemption Date, the record holder of the Warrants shall have no further rights except to receive, upon surrender of the Warrants, the Redemption Price.
- 6.5 Exclusion of Private Placement Warrants and Working Capital Warrants. The Company agrees that the redemption rights provided in Section 6.1 hereof shall not apply to the Private Placement Warrants or the Working Capital Warrants if at the time of the redemption such Private Placement Warrants or the Working Capital Warrants or Working Capital Warrants are transferred (other than to Permitted Transferees in accordance with Section 2.5), the Company may redeem the Private Placement Warrants and the Working Capital Warrants pursuant to Section 6.1 hereof, provided that the criteria for redemption are met, including the opportunity of the holder of any such Private Placement Warrants and Working Capital Warrants to exercise the Private Placement Warrants or Working Capital Warrants prior to redemption pursuant to Section 6.4. Private Placement Warrants or Working Capital Warrants that are transferred to persons other than Permitted Transferees shall upon such transfer cease to be Private Placement Warrants or Working Capital Warrants and shall become Public Warrants under this Agreement.

- 7. Other Provisions Relating to Rights of Holders of Warrants.
- 7.1 No Rights as Shareholder. A Warrant does not entitle the Registered Holder thereof to any of the rights of a shareholder of the Company, including, without limitation, the right to receive dividends, or other distributions, exercise any preemptive rights to vote or to consent or to receive notice as shareholders in respect of the meetings of shareholders or the election of directors of the Company or any other matter.
- 7.2 Lost, Stolen, Mutilated, or Destroyed Warrants. If any Warrant is lost, stolen, mutilated, or destroyed, the Company and the Warrant Agent may on such terms as to indemnity or otherwise as they may in their discretion impose (which shall, in the case of a mutilated Warrant, include the surrender thereof), issue a new Warrant of like denomination, tenor, and date as the Warrant so lost, stolen, mutilated, or destroyed. Any such new Warrant shall constitute a substitute contractual obligation of the Company, whether or not the allegedly lost, stolen, mutilated, or destroyed Warrant shall be at any time enforceable by anyone.
- 7.3 <u>Reservation of Ordinary Shares</u>. The Company shall at all times reserve and keep available a number of unissued Ordinary Shares in its authorized share capital as provided for in the Company's articles of association that shall be sufficient to permit the exercise in full of all outstanding Warrants issued pursuant to this Agreement.
 - 7.4 Registration of Ordinary Shares; Cashless Exercise at Company's Option.
- 7.4.1 Registration of the Ordinary Shares. The Company agrees that as soon as practicable, but in no event later than twenty (20) Business days after the Effective Date, it shall use its commercially reasonable efforts to file with the Commission a post-effective amendment to the registration statement or a new registration statement for the registration, under the Securities Act, of the issuance of the Ordinary Shares issuable upon exercise of the Warrants. The Company shall use its commercially reasonable efforts to cause the same to become effective within sixty (60) days after the closing of the Effective Date and to maintain the effectiveness of such registration statement, and a current prospectus relating thereto, until the expiration of the Warrants in accordance with the provisions of this Agreement. If any such registration statement has not been declared effective by the 60th Business Day following the Effective Date, holders of the Warrants shall have the right, during the period beginning on the 61st Business Day after the Effective Date and ending upon such registration statement being declared effective by the Commission, and during any other period when the Company shall fail to have maintained an effective registration statement covering the Ordinary Shares issuable upon exercise of the Warrants, to exercise such Warrants on a "cashless basis," by exchanging the Warrants (in accordance with Section 3(a)(9) of the Securities Act (or any successor rule) or another exemption) for that number of Ordinary Shares equal to the quotient obtained by dividing (x) the product of the number of Ordinary Shares underlying the Warrants, multiplied by the excess of the "Fair Market Value" (as defined below) over the Warrant Price by (y) the Fair Market Value. The nominal value of Ordinary Shares issued upon exercise of Warrants on a

"cashless basis" may be paid-up at the expense of the reserves of the Company. Solely for purposes of this <u>subsection 7.4.1</u>, "Fair Market Value" shall mean the volume weighted average price of the Ordinary Shares as reported during the ten (10) trading day period ending on the trading day prior to the date that notice of exercise is received by the Warrant Agent from the holder of such Warrants or its securities broker or intermediary. The date that notice of cashless exercise is received by the Warrant Agent shall be conclusively determined by the Warrant Agent. In connection with the "cashless exercise" of a Public Warrant, the Company shall, upon request, provide the Warrant Agent with an opinion of counsel for the Company (which shall be an outside law firm with securities law experience) stating that (i) the exercise of the Warrants on a cashless basis in accordance with this <u>subsection 7.4.1</u> is not required to be registered under the Securities Act and (ii) the Ordinary Shares issued upon such exercise shall be freely tradable under United States federal securities laws by anyone who is not an affiliate (as such term is defined in Rule 144 under the Securities Act (or any successor rule)) of the Company and, accordingly, shall not be required to bear a restrictive legend. Except as provided in <u>subsection 7.4.2</u>, for the avoidance of any doubt, unless and until all of the Warrants have been exercised, the Company shall continue to be obligated to comply with its registration obligations under the first three sentences of this <u>subsection 7.4.1</u>.

7.4.2 <u>Cashless Exercise at Company's Option</u>. If the Ordinary Shares are at the time of any exercise of a Warrant not listed on a national securities exchange such that it satisfies the definition of a "covered security" under Section 18(b)(1) of the Securities Act (or any successor rule), the Company may, at its option, (i) require holders of Public Warrants who exercise Public Warrants to exercise such Public Warrants on a "cashless basis" in accordance with Section 3(a)(9) of the Securities Act (or any successor rule) as described in <u>subsection 7.4.1</u> and (ii) in the event the Company so elects, the Company shall (x) not be required to file or maintain in effect a registration statement for the registration, under the Securities Act, of the Ordinary Shares issuable upon exercise of the Warrants, notwithstanding anything in this Agreement to the contrary, and (y) use its best efforts to register or qualify for sale the Ordinary Shares issuable upon exercise of the Public Warrants under the blue sky laws of the state of residence of the exercising Public Warrant holder to the extent an exemption is not available.

- 8. Concerning the Warrant Agent and Other Matters.
- 8.1 <u>Payment of Taxes</u>. The Company shall from time to time promptly pay all taxes and charges that may be imposed upon the Company or the Warrant Agent in respect of the issuance or delivery of Ordinary Shares upon the exercise of the Warrants, but the Company shall not be obligated to pay any transfer taxes in respect of the Warrants or such Ordinary Shares.
 - 8.2 Resignation, Consolidation, or Merger of Warrant Agent.
- 8.2.1 Appointment of Successor Warrant Agent. The Warrant Agent, or any successor to it hereafter appointed, may resign its duties and be discharged from all further duties and liabilities hereunder after giving sixty (60) days' notice in writing to the Company. If the office of the Warrant Agent becomes vacant by resignation or incapacity to act or otherwise, the Company shall appoint in writing a successor Warrant Agent in place of the Warrant Agent. If the

Company shall fail to make such appointment within a period of thirty (30) days after it has been notified in writing of such resignation or incapacity by the Warrant Agent or by the holder of a Warrant (who shall, with such notice, submit his Warrant for inspection by the Company), then the holder of any Warrant may apply to the Supreme Court of the State of New York for the County of New York for the appointment of a successor Warrant Agent at the Company's cost. Any successor Warrant Agent, whether appointed by the Company or by such court, shall be a corporation organized and existing under the laws of the State of New York, in good standing and having its principal office in the Borough of Manhattan, City and State of New York, and authorized under such laws to exercise corporate trust powers and subject to supervision or examination by federal or state authority. After appointment, any successor Warrant Agent shall be vested with all the authority, powers, rights, immunities, duties, and obligations of its predecessor Warrant Agent with like effect as if originally named as Warrant Agent hereunder, without any further act or deed; but if for any reason it becomes necessary or appropriate, the predecessor Warrant Agent shall execute and deliver, at the expense of the Company, an instrument transferring to such successor Warrant Agent all the authority, powers, and rights of such predecessor Warrant Agent hereunder; and upon request of any successor Warrant Agent the Company shall make, execute, acknowledge, and deliver any and all instruments in writing for more fully and effectually vesting in and confirming to such successor Warrant Agent all such authority, powers, rights, immunities, duties, and obligations.

- 8.2.2 Notice of Successor Warrant Agent. In the event a successor Warrant Agent shall be appointed, the Company shall give notice thereof to the predecessor Warrant Agent and the Transfer Agent for the Ordinary Shares not later than the effective date of any such appointment.
- 8.2.3 Merger or Consolidation of Warrant Agent. Any corporation into which the Warrant Agent may be merged or with which it may be consolidated or any corporation resulting from any merger or consolidation to which the Warrant Agent shall be a party shall be the successor Warrant Agent under this Agreement without any further act.
 - 8.3 Fees and Expenses of Warrant Agent.
- 8.3.1 Remuneration. The Company agrees to pay the Warrant Agent reasonable remuneration for its services as such Warrant Agent hereunder and shall, pursuant to its obligations under this Agreement, reimburse the Warrant Agent upon demand for all expenditures that the Warrant Agent may reasonably incur in the execution of its duties hereunder.
- 8.3.2 <u>Further Assurances</u>. The Company agrees to perform, execute, acknowledge, and deliver or cause to be performed, executed, acknowledged, and delivered all such further and other acts, instruments, and assurances as may reasonably be required by the Warrant Agent for the carrying out or performing of the provisions of this Agreement.

8.4 Liability of Warrant Agent.

- 8.4.1 <u>Reliance on Company Statement</u>. Whenever in the performance of its duties under this Agreement, the Warrant Agent shall deem it necessary or desirable that any fact or matter be proved or established by the Company prior to taking or suffering any action hereunder, such fact or matter (unless other evidence in respect thereof be herein specifically prescribed) may be deemed to be conclusively proved and established by a statement signed by the Chief Executive Officer, Chief Financial Officer, Secretary or Chairman of the Board of the Company and delivered to the Warrant Agent. The Warrant Agent may rely upon such statement for any action taken or suffered in good faith by it pursuant to the provisions of this Agreement.
- 8.4.2 <u>Indemnity</u>. The Warrant Agent shall be liable hereunder only for its own gross negligence, willful misconduct or bad faith. The Company agrees to indemnify the Warrant Agent and save it harmless against any and all liabilities, including judgments, costs and reasonable counsel fees, for anything done or omitted by the Warrant Agent in the execution of this Agreement, except as a result of the Warrant Agent's gross negligence, willful misconduct or bad faith.
- 8.4.3 Exclusions. The Warrant Agent shall have no responsibility with respect to the validity of this Agreement or with respect to the validity or execution of any Warrant (except its countersignature thereof). The Warrant Agent shall not be responsible for any breach by the Company of any covenant or condition contained in this Agreement or in any Warrant. The Warrant Agent shall not be responsible to make any adjustments required under the provisions of Section 4 hereof or responsible for the manner, method, or amount of any such adjustment or the ascertaining of the existence of facts that would require any such adjustment; nor shall it by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of any Ordinary Shares to be issued pursuant to this Agreement or any Warrant or as to whether any Ordinary Shares shall, when issued, be valid and fully paid and non-assessable.
- 8.5 <u>Acceptance of Agency</u>. The Warrant Agent hereby accepts the agency established by this Agreement and agrees to perform the same upon the terms and conditions herein set forth and among other things, shall account promptly to the Company with respect to Warrants exercised and concurrently account for, and pay to the Company, all monies received by the Warrant Agent for the purchase of Ordinary Shares through the exercise of the Warrants.
- 8.6 Waiver. The Warrant Agent has no right of set-off or any other right, title, interest or claim of any kind ("Claim") in, or to any distribution of, the Trust Account (as defined in that certain Investment Management Trust Agreement, dated as of the date hereof, by and between the Company and the Warrant Agent as trustee thereunder) and hereby agrees not to seek recourse, reimbursement, payment or satisfaction for any Claim against the Trust Account for any reason whatsoever. The Warrant Agent hereby waives any and all Claims against the Trust Account and any and all rights to seek access to the Trust Account.
 - 9. Miscellaneous Provisions.
- 9.1 <u>Successors</u>. All the covenants and provisions of this Agreement by or for the benefit of the Company or the Warrant Agent shall bind and inure to the benefit of their respective successors and assigns.

9.2 Notices. Any notice, statement or demand authorized by this Agreement to be given or made by the Warrant Agent or by the holder of any Warrant to or on the Company shall be sufficiently given when so delivered if by hand or overnight delivery or if sent by certified mail or private courier service within five (5) days after deposit of such notice, postage prepaid, addressed (until another address is filed in writing by the Company with the Warrant Agent), as follows:

Wallbox N.V. Carrer del Foc, 68 Barcelona, Spain 08038 Attention: Enric Asuncion With a copy in each case to:

Latham & Watkins LLP 811 Main Street Suite 3700 Houston, TX 77002 Attn: Ryan Maierson

Any notice, statement or demand authorized by this Agreement to be given or made by the holder of any Warrant or by the Company to or on the Warrant Agent shall be sufficiently given when so delivered if by hand or overnight delivery or if sent by certified mail or private courier service within five (5) days after deposit of such notice, postage prepaid, addressed (until another address is filed in writing by the Warrant Agent with the Company), as follows:

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, NY 10004 Attention: Compliance Department

With a copy in each case to:

Latham & Watkins LLP 811 Main Street Suite 3700 Houston, TX 77002 Attn: Ryan Maierson

9.3 Applicable Law. The validity, interpretation, and performance of this Agreement and of the Warrants shall be governed in all respects by the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. The Company hereby agrees that any action, proceeding or claim against it arising out of or relating in any way to this Agreement shall be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. The Company hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum. Notwithstanding the foregoing, the provisions of this paragraph will not apply to suits brought to enforce any liability or duty created by the Exchange Act or any other claim for which the federal district courts of the United States of America are the sole and exclusive forum.

Any person or entity purchasing or otherwise acquiring any interest in the Warrants shall be deemed to have notice of and to have consented to the forum provisions in this Section 9.3. If any action, the subject matter of which is within the scope the forum provisions above, is filed in a court other than a court located within the State of New York or the United States District Court for the Southern District of New York (a "NY foreign action") in the name of any warrant holder, such warrant holder shall be deemed to have consented to: (x) the personal jurisdiction of the state and federal courts located within the State of New York or the United States District Court for the Southern District of New York in connection with any action brought in any such court to enforce the forum provisions (an "NY enforcement action"), and (y) having service of process made upon such warrant holder in any such NY enforcement action by service upon such warrant holder's counsel in the NY foreign action as agent for such warrant holder.

- 9.4 <u>Persons Having Rights under this Agreement</u>. Nothing in this Agreement shall be construed to confer upon, or give to, any person or corporation other than the parties hereto and the Registered Holders of the Warrants any right, remedy, or claim under or by reason of this Agreement or of any covenant, condition, stipulation, promise, or agreement hereof. All covenants, conditions, stipulations, promises, and agreements contained in this Agreement shall be for the sole and exclusive benefit of the parties hereto and their successors and assigns and of the Registered Holders of the Warrants.
- 9.5 Examination of the Warrant Agreement. A copy of this Agreement shall be available at all reasonable times at the office of the Warrant Agent in the Borough of Manhattan, City and State of New York, for inspection by the Registered Holder of any Warrant. The Warrant Agent may require any such holder to submit his Warrant for inspection by it.
- 9.6 <u>Counterparts</u>. This Agreement may be executed in any number of original or facsimile counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.
- 9.7 Effect of Headings. The section headings herein are for convenience only and are not part of this Agreement and shall not affect the interpretation thereof.
- 9.8 Amendments. This Agreement may be amended by the parties hereto without the consent of any Registered Holder for the purpose of curing any ambiguity, or curing, correcting or supplementing any defective provision contained herein or adding or changing any other provisions with respect to matters or questions arising under this Agreement as the parties may deem necessary or desirable and that the parties deem shall not adversely affect the interest of the Registered Holders. All other modifications or amendments, including any amendment to increase the Warrant Price or shorten the Exercise Period, shall require the vote or written consent of the Registered Holders of 50% of the then outstanding Public Warrants, and, solely with respect to any amendment to the terms of, or any provision of this Agreement with respect to the Private Placement Warrants or Working Capital Warrants, 50% of the number of then outstanding Private Placement Warrants or Working Capital Warrants, as applicable. Notwithstanding the foregoing, the Company may lower the Warrant Price or extend the duration of the Exercise Period pursuant to Sections 3.1 and 3.2, respectively, without the consent of the Registered Holders.

9.9 Severability. This Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect
the validity or enforceability of this Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term
or provision, the parties hereto intend that there shall be added as a part of this Agreement a provision as similar in terms to such invalid or
unenforceable provision as may be possible and be valid and enforceable.

Exhibit A Form of Warrant Certificate

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

KENSINGTON CAPITAL ACQUISITION CORP. II

By: /s/ Justin Mirro

Name: Justin Mirro

Title: Chief Executive Officer

CONTINENTAL STOCK TRANSFER & TRUST

COMPANY, as Warrant Agent

By: /s/ Ana Gois

Name: Ana Gois Title: Vice President

WALLBOX N.V.

By: /s/ Enric Asunción Escorsa

Name: Enric Asunción Escorsa Title: Chief Executive Officer

EXHIBIT A

[Form of Warrant Certificate]

[FACE]

Number

Warrants

THIS WARRANT SHALL BE VOID IF NOT EXERCISED PRIOR TO THE EXPIRATION OF THE EXERCISE PERIOD PROVIDED FOR IN THE WARRANT AGREEMENT DESCRIBED BELOW

WALLBOX N.V.

Incorporated Under the Laws of the Netherlands

CUSIP	[
-------	---

Warrant Certificate

This Warrant Certificate certifies that [], or registered assigns, is the registered holder of [] warrants evidenced hereby (the "Warrants" and each, a "Warrant") to purchase ordinary shares with a nominal value of 0.12 the "Ordinary Shares") of Wallbox N.V., a public company with limited liability (naamloze vennootschap) incorporated under the laws of the Netherlands, having its official seat in Amsterdam, the Netherlands and registered with the Dutch trade register under number 83012559 (the "Company"). Each Warrant entitles the holder, upon exercise during the Period set forth in the Assignment, Assumption and Amended and Restated Warrant Agreement, dated as of October 1, 2021 (as amended from time to time, the "Warrant Agreement"), to receive from the Company that number of fully paid and non-assessable Ordinary Shares as set forth below, at the exercise price (the "Exercise Price") as determined pursuant to the Warrant Agreement, payable in lawful money (or through "cashless exercise" as provided for in the Warrant Agreement) of the United States of America upon surrender of this Warrant Certificate and payment of the Exercise Price at the office or agency of the Warrant Agent referred to below, subject to the conditions set forth herein and in the Warrant Agreement.

Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Each whole Warrant is initially exercisable for one fully paid and non-assessable Ordinary Share. No fractional shares will be issued upon exercise of any Warrant. If, upon the exercise of a Warrant, a holder would be entitled to receive a fractional interest in an Ordinary Share, the Company will, upon exercise, round down to the nearest whole number the number of Ordinary Shares to be issued to the Warrant holder. The number of Ordinary Shares issuable upon exercise

of the Warrants is subject to adjustment upon the occurrence of certain events set forth in the Warrant Agreement. The initial Exercise Price per Ordinary Share for any Warrant is equal to \$11.50 per share. The Exercise Price is subject to adjustment upon the occurrence of certain events set forth in the Warrant Agreement. The Warrants may be redeemed subject to certain conditions as set forth in the Warrant Agreement. Subject to the conditions set forth in the Warrant Agreement, the Warrants may be exercised only during the Exercise Period and to the extent not exercised by the end of such Exercise Period, such Warrants shall become void.

Reference is hereby made to the further provisions of this Warrant Certificate set forth on the reverse hereof and such further provisions shall for all purposes have the same effect as though fully set forth at this place.

This Warrant Certificate shall not be valid unless countersigned by the Warrant Agent, as such term is used in the Warrant Agreement.

This Warrant Certificate shall be governed by and construed in accordance with the internal laws of the State of New York, without regard to conflicts of laws principles thereof.

WALLBOX N.V.
By:
Name:
Title:
CONTINENTAL STOCK TRANSFER & TRUST COMPANY as Warrant Agent
By:
Name:
Title:

[Form of Warrant Certificate] [Reverse]

The Warrants evidenced by this Warrant Certificate are part of a duly authorized issue of Warrants entitling the holder on exercise to receive Ordinary Shares and are issued or to be issued pursuant to an Assignment, Assumption and Amended and Restated Warrant Agreement, dated as of October 1, 2021 (as amended from time to time, the "Warrant Agreement"), duly executed and delivered by the Company to Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "Warrant Agent"), which Warrant Agreement is hereby incorporated by reference in and made a part of this instrument and is hereby referred to for a description of the rights, limitation of rights, obligations, duties and immunities thereunder of the Warrant Agent, the Company and the holders (the words "holders" or "holder" meaning the Registered Holders or Registered Holder) of the Warrants. A copy of the Warrant Agreement may be obtained by the holder hereof upon written request to the Company. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Warrants may be exercised at any time during the Exercise Period set forth in the Warrant Agreement. The holder of Warrants evidenced by this Warrant Certificate may exercise them by surrendering this Warrant Certificate, with the form of election to purchase set forth hereon properly completed and executed, together with payment of the Exercise Price as specified in the Warrant Agreement (or through "cashless exercise" as provided for in the Warrant Agreement) at the principal corporate trust office of the Warrant Agent. In the event that upon any exercise of Warrants evidenced hereby the number of Warrants exercised shall be less than the total number of Warrants evidenced hereby, there shall be issued to the holder hereof or his, her or its assignee, a new Warrant Certificate evidencing the number of Warrants not exercised.

Notwithstanding anything else in this Warrant Certificate or the Warrant Agreement, no Warrant may be exercised unless at the time of exercise (i) a registration statement covering the Ordinary Shares to be issued upon exercise is effective under the Securities Act, or a valid exemption from registration is available, and (ii) a prospectus thereunder relating to the Ordinary Shares is current, except through "cashless exercise" as provided for in the Warrant Agreement.

The Warrant Agreement provides that upon the occurrence of certain events the number of Ordinary Shares issuable upon exercise of the Warrants set forth on the face hereof may, subject to certain conditions, be adjusted. If, upon exercise of a Warrant, the holder thereof would be entitled to receive a fractional interest in an Ordinary Share, the Company shall, upon exercise, round down to the nearest whole number of Ordinary Shares to be issued to the holder of the Warrant.

Warrant Certificates, when surrendered at the principal corporate trust office of the Warrant Agent by the Registered Holder thereof in person or by legal representative or attorney duly authorized in writing, may be exchanged, in the manner and subject to the limitations provided in the Warrant Agreement, but without payment of any service charge, for another Warrant Certificate or Warrant Certificates of like tenor evidencing in the aggregate a like number of Warrants.

Upon due presentation for registration of transfer of this Warrant Certificate at the office of the Warrant Agent, a new Warrant Certificate or Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee(s) in exchange for this Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any tax or other governmental charge imposed in connection therewith.

The Company and the Warrant Agent may deem and treat the Registered Holder(s) hereof as the absolute owner(s) of this Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone), for the purpose of any exercise hereof, of any distribution to the holder(s) hereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.

Neither the Warrants nor this Warrant Certificate entitles any holder hereof to any rights of a shareholder of the Company.

Election to Purchase

(To Be Executed Upon Exercise of Warrant)

The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, to receive [] Ordinary Shares and herewith tenders payment for such Ordinary Shares to the order of Wallbox N.V., a public company with limited liability (naamloze vennootschap) incorporated under the laws of the Netherlands, having its official seat in Amsterdam, the Netherlands and registered with the Dutch trade register under number 83012559(the "Company") in the amount of \$[] in accordance with the terms hereof. The undersigned requests that a certificate for such Ordinary Shares be registered in the name of [] whose address is [] and that such Ordinary Shares be delivered to [] whose address is []. If said number of shares is less than all of the Ordinary Shares purchasable hereunder, the undersigned requests that a new Warrant Certificate representing the remaining balance of such Ordinary Shares be registered in the name of [], whose address is [] and that such Warrant Certificate be delivered to [], whose address is [].

In the event that the Warrant has been called for redemption by the Company pursuant to Section 6 of the Warrant Agreement and the Company has required cashless exercise pursuant to Section 6.4 of the Warrant Agreement, the number of Ordinary Shares that this Warrant is exercisable for shall be determined in accordance with subsection 3.3.1(b) and Section 6.4 of the Warrant Agreement.

In the event that the Warrant has been called for redemption by the Company pursuant to <u>Section 6.2</u> of the Warrant Agreement and a holder thereof elects to exercise its Warrant pursuant to a Make-Whole Exercise, the number of Ordinary Shares that this Warrant is exercisable for shall be determined in accordance with <u>Section 6.2</u> of the Warrant Agreement.

In the event that the Warrant is a Private Placement Warrant that is to be exercised on a "cashless" basis pursuant to <u>subsection 3.3.1(c)</u> of the Warrant Agreement, the number of Ordinary Shares that this Warrant is exercisable for shall be determined in accordance with <u>subsection 3.3.1(c)</u> of the Warrant Agreement.

In the event that the Warrant is to be exercised on a "cashless" basis pursuant to <u>Section 7.4</u> of the Warrant Agreement, the number of Ordinary Shares that this Warrant is exercisable for shall be determined in accordance with <u>Section 7.4</u> of the Warrant Agreement.

In the event that the Warrant may be exercised, to the extent allowed by the Warrant Agreement, through cashless exercise (i) the number of Ordinary Shares that this Warrant is exercisable for would be determined in accordance with the relevant section of the Warrant Agreement which allows for such cashless exercise and (ii) the holder hereof shall complete the following: The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, through the cashless exercise provisions of the Warrant Agreement, to receive Ordinary Shares. If said number of shares is less than all of the Ordinary Shares purchasable hereunder (after giving effect to the cashless exercise), the undersigned requests that a new Warrant Certificate representing the remaining balance of such Ordinary Shares be registered in the name of [], whose address is [] and that such Warrant Certificate be delivered to [], whose address is [].

[Signature Page Follows]

Date:	, 20	
		(Signature)
		(Address)
		(Tax Identification Number)
Signature (Guaranteed:	

THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15 (OR ANY SUCCESSOR RULE)).

REGISTRATION RIGHTS AND LOCK-UP AGREEMENT

This Registration Rights and Lock-Up Agreement (this "Agreement"), dated as of October 1, 2021, is among Wallbox N.V., a public company with limited liability incorporated under the laws of the Netherlands (naamloze vennootschap) (the "Company") and the parties listed on Schedule A (each, a "Holder" and collectively, the "Holders"). Capitalized terms used but not defined herein have the meanings assigned to them in the Business Combination Agreement, dated as of the date hereof (the "Business Combination Agreement"), among the Company, Kensington Capital Acquisition Corp. II, a Delaware corporation ("Kensington"), Orion Merger Sub Corp., a Delaware corporation ("Merger Sub"), and Wall Box Chargers, S.L., a company organized under the laws of Spain ("Wallbox").

WHEREAS, the Company, Kensington, Merger Sub and Wallbox are parties to the Business Combination Agreement, pursuant to which, among other things, (a) pursuant to the terms of that certain Contribution and Exchange Agreement (the "Exchange Agreement") and in accordance with the Dutch Civil Code (Burgerlijk Wetboek), Kensington, the Company, Merger Sub and Wallbox will enter into a business combination transaction pursuant to which, among other things, (a) pursuant to the Exchange Agreement each holder of Wallbox convertible notes (the "Convertible Noteholders"), after the execution of the Business Combination Agreement and prior to the Exchange Effective Time, will convert its respective convertible notes in exchange for the issuance of Ordinary Shares to be subscribed for by such Convertible Noteholder (such conversions and exchanges of Company Convertible Notes, collectively, the "Convert Exchange"), (b) pursuant to the Exchange Agreement each holder of Wallbox ordinary shares (including ordinary shares issued as a result of the conversion of the convertible notes) (the "Wallbox Ordinary Shareholders"), effective immediately prior to the Merger Effective Time (the "Exchange Effective Time"), will contribute its respective Wallbox ordinary shares to the Company in exchange for the issuance of Ordinary Shares to be subscribed for by such Wallbox Ordinary Shareholder (such contributions and exchanges of Company Ordinary Shares, collectively, the "Ordinary Exchange" and, together with the Convert Exchange, the "Exchanges"), (c) as a result of the Ordinary Exchange Wallbox will become a wholly-owned subsidiary of the Company and (d) following the consummation of the Exchanges, Merger Sub will merge with and into Kensington, with Kensington surviving such merger and, as a result of such Merger, all shares of Kensington Common Stock (other than Excluded Shares, as defined in the Business Combination Agreement) outstanding immediately prior to the Merger Effective Time shall be converted into shares of New Kensington Common Stock, which shares shall immediately thereafter be exchanged for the right to receive the Merger Consideration in the form of Ordinary A Shares, as set forth in the Business Combination Agreement, and thereafter the New Kensington Common Stock shall be exchanged by means of a contribution in kind for Ordinary A Shares with the result that Kensington will become a direct wholly-owned subsidiary of Holdco (the merger, together with the automatic exchange, the "Merger"). Concurrently with the signing of the Business Combination Agreement and on September 29, 2021, certain investors have entered into subscription agreements relating to the purchase of Ordinary Shares immediately following the consummation of the Merger (the "PIPE");

WHEREAS, the Company and the Holder designated as an "Original Holder" on <u>Schedule A</u> (the "**Original Holder**") are parties to the Registration Rights Agreement dated as of February 25, 2021 (the "**Prior Agreement**");

WHEREAS, the Original Holder currently holds an aggregate of 5,750,000 shares of the Kensington's Class B common stock, par value \$0.0001 per share (the "Founder Shares");

WHEREAS, the Original Holder currently holds an aggregate of 8,800,000 redeemable warrants (the "**Kensington Warrants**") to purchase shares of Kensington's Class A common stock, par value \$0.0001 per share, at an exercise price of \$11.50 per share;

WHEREAS, the Holders designated as "New Holders" on <u>Schedule A</u>, being all of the Wallbox Ordinary Shareholders as of the date of the Business Combination Agreement, (the "**New Holders**") will receive Ordinary Shares upon consummation of the Closing pursuant to the Business Combination Agreement;

WHEREAS, at the Effective Time, the Founder Shares will be converted into the same number of Ordinary A Shares and the Kensington Warrants will be converted into the same number of warrants of the Company, each warrant with the right to acquire one (1) Ordinary A Share (the "Private Placement Warrants"); and

WHEREAS, contingent upon the Closing and effective as of the Effective Time, the parties to the Prior Agreement desire to terminate the Prior Agreement and to provide for certain rights and obligations included herein and to include the New Holders.

NOW, THEREFORE, in consideration of the foregoing, the parties hereby agree as follows:

ARTICLE I DEFINITIONS

Section 1.1. Definitions. For purposes of this Agreement, the following terms and variations thereof have the meanings set forth below:

"Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or Chief Financial Officer of the Company, after consultation with outside counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) the Company has a bona fide business purpose for not making such information public.

"Agreement" shall have the meaning given in the Preamble.

"Board" shall mean the Board of Directors of the Company.

"Business Combination" shall mean any merger, share exchange, asset acquisition, stock purchase, recapitalization, reorganization or other similar business combination with one or more businesses, involving the Company.

"Business Combination Agreement" shall have the meaning given in the Preamble.

"Business Day" means a day other than Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by law to close.

"Change in Control" means the transfer (whether by tender offer, merger, stock purchase, consolidation or other similar transaction), in one transaction or a series of related transactions, to a person or group of affiliated persons of the Company's voting securities if, after such transfer, such person or group of affiliated persons would hold more than 50% of outstanding voting securities of the Company (or surviving entity) or would otherwise have the power to control the board of directors of the Company or to direct the operations of the Company.

- "Commission" means the Securities and Exchange Commission.
- "Company" shall have the meaning given in the Preamble.
- "Convert Exchange" shall have the meaning given in the Recitals.
- "Demand Registration" shall have the meaning given in subsection 2.1.1.
- "Demand Requesting Holder" shall have the meaning given in subsection 2.1.1.
- "Demanding Holders" shall have the meaning given in subsection 2.1.1.
- "Effectiveness Deadline" shall have the meaning given in subsection 2.3.1.
- "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
- "Form F-1" means a Registration Statement on Form F-1 or any comparable successor form or forms thereto.
- "Form F-3" means a Registration Statement on Form F-3 or any comparable successor form or forms thereto.
- "Founder Shares" shall have the meaning given in the Recitals.
- "Holders" shall have the meaning given in the Preamble.
- "Exchange Agreement" shall have the meaning given in the Recitals.
- "Exchange Effective Time" shall have the meaning given in the Recitals.
- "Exchanges" shall have the meaning given in the Recitals.
- "Kensington" shall have the meaning given in the Preamble.
- "Maximum Number of Securities" shall have the meaning given in subsection 2.1.4.
- "Merger Sub" shall have the meaning given in the Preamble.
- "Misstatement" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus in the light of the circumstances under which they were made not misleading.
 - "New Holder Lock-up Period" shall have the meaning given in subsection 5.1.1.
 - "New Holders" shall have the meaning given in the Recitals.
 - "New Registration Statement" shall have the meaning given in subsection 2.3.4.
 - "Ordinary A Shares" means Class A ordinary shares of the Company with a nominal value of 0.12 per share.

- "Ordinary B Shares" means Class B ordinary shares of the Company with a nominal value of 1.20 per share.
- "Ordinary Exchange" shall have the meaning given in the Recitals.
- "Ordinary Shares" means Ordinary A Shares and Ordinary B Shares.
- "Original Holder" shall have the meaning given in the Recitals.
- "Original Holder Lock-Up Period" shall have the meaning given in subsection 5.1.2.
- "Piggyback Registration" shall have the meaning given in subsection 2.3.1.
- "PIPE" shall have the meaning given in the Recitals.
- "Prior Agreement" shall have the meaning given in the Recitals.
- "Private Placement Warrants" shall have the meaning given in the Recitals.
- "Pro Rata" shall have the meaning given in subsection 2.1.4.
- "Prospectus" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

"Registrable Security", "Registrable Securities" shall mean (a) the Private Placement Warrants (including any Ordinary Shares issued or issuable upon the exercise of any such Private Placement Warrants), (b) any outstanding Ordinary Shares or any other equity security (including the shares of Ordinary Shares issued or issuable upon the exercise of any other equity security) of the Company held by a Holder as of the Closing Date (including the Ordinary Shares issued pursuant to the Business Combination Agreement), and (c) any other equity security of the Company issued or issuable with respect to any such share of Ordinary Shares by way of a share dividend or share split or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred, new certificates or evidence of book-entry entitlements for such securities not bearing a legend restricting further transfer shall have been delivered by the Company to the transferee, and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; (D) such securities may be sold without registration pursuant to Rule 144 or any successor rule promulgated under the Securities Act (but with no volume or other restrictions or limitations including as to manner or timing of sale); or (F) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"Registration" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

- "Registration Expenses" shall mean the out-of-pocket expenses of a Registration or Underwritten Offering, including, without limitation, the following:
- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Ordinary Shares is then listed;
- (B) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
 - (C) printing, messenger, telephone and delivery expenses;
 - (D) reasonable fees and disbursements of counsel for the Company;
- (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration or Underwritten Offering; and
- (F) reasonable fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders or the majority-in interest of the Takedown Requesting Holders, as applicable.
- "Registration Statement" shall mean any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.
 - "Requesting Holder" shall have the meaning given in subsection 2.3.5.
 - "Resale Shelf Registration Statement" shall have the meaning given in subsection 2.3.1.
 - "Securities Act" shall mean the Securities Act of 1933, as amended from time to time.
 - "Selling Holders" means any Holder electing to sell any of its Registrable Securities in a Registration.
 - "SEC Guidance" shall have the meaning given in subsection 2.3.4.
 - "Takedown Requesting Holder" shall have the meaning given in subsection 2.3.5.
- "Transfer" means to, directly or indirectly, sell, transfer, assign, pledge, encumber, hypothecate or similarly dispose of, either voluntarily or involuntarily, or to enter into any contract, option or other arrangement or understanding with respect to the sale, transfer, assignment, pledge, encumbrance, hypothecation or similar disposition of, any interest owned by a person or any interest (including a beneficial interest) in, or the ownership, control or possession of, any interest owned by a person.
- "Underwriter" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.
- "Underwritten Registration" or "Underwritten Offering" shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public, including for the avoidance of doubt an Underwritten Shelf Takedown.

- "Underwritten Shelf Takedown" shall have the meaning given in subsection 2.3.5.
- "Wallbox" shall have the meaning given in the Preamble.
- "Wallbox Convertible Notes" shall have the meaning given in the Recitals.
- "Wallbox Ordinary Shares" shall have the meaning given in the Recitals.

ARTICLE II REGISTRATION

Section 2.1. Demand Registration.

- 2.1.1 Request for Registration. Subject to the provisions of subsection 2.1.4 and Section 2.4 hereof, at any time and from time to time following the Effective Time (but subject to Article V), Holders holding at least 15% in interest of the then-outstanding number of Registrable Securities held by all Holders (such Holders, the "Demanding Holders") may make a written demand for Registration of all or part of their Registrable Securities on Form F-3 (or, if Form F-3 is not available to be used by the Company at such time, on Form F-1 or another appropriate form permitting Registration of such Registrable Securities for resale by such Demanding Holders), which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand a "Demand Registration"). The Company shall, within ten (10) days of the Company's receipt of the Demand Registration, notify, in writing, all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Demand Requesting Holder") shall so notify the Company, in writing, within five (5) days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Demand Requesting Holder(s) to the Company, such Demand Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall effect, as soon thereafter as practicable, but not more than sixty (60) days immediately after the Company's receipt of the Demand Registration, the Registration of all Registrable Securities requested by the Demanding Holders and Demand Requesting Holders pursuant to such Demand Registration. Under no circumstances shall the Company be obligated to effect more than an aggregate of (i) three (3) Registrations pursuant to a Demand Registration under this subsection 2.1.1 initiated by New Holders, or (ii) one (1) Registration pursuant to a Demand Registration under this subsection 2.1.1 initiated by the Original Holder.
- 2.1.2 Effective Registration. Notwithstanding the provisions of subsection 2.1.1 above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; provided, further, however, that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency, the Registration Statement with respect to such Registration shall be deemed not to have been declared effective for purposes of counting Registrations under subsection 2.1.1 above unless and until (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with

such Registration and accordingly notify the Company in writing, but in no event later than five (5) days, of such election; <u>provided, further, however</u>, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration becomes effective or has been terminated.

- 2.1.3 <u>Underwritten Offering</u>. Subject to the provisions of <u>subsection 2.1.4</u> and <u>Section 2.4</u> hereof, if a majority-in-interest of the Demanding Holders advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such Demanding Holder or Demand Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.1.3</u> shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company, which Underwriter(s) shall be reasonably acceptable to a majority-in-interest of the Demanding Holders initiating the Demand Registration.
- 2.1.4 Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Registration pursuant to a Demand Registration, in good faith, advises the Company, the Demanding Holders and the Demand Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Demand Requesting Holders (if any) desire to sell, taken together with all other Ordinary Shares or other equity securities that the Company desires to sell and the Ordinary Shares, if any, as to which a Registration has been requested pursuant to separate written contractual piggy-back registration rights held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and the Demand Requesting Holders (if any) (pro rata based on the respective number of Registrable Securities that each Demanding Holder and Demand Requesting Holder (if any) has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Demand Requesting Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), Ordinary Shares or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), Ordinary Shares or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.
- 2.1.5 <u>Demand Registration Withdrawal</u>. A majority-in-interest of the New Holders or the Original Holder, as the case may be, in the case of a Registration under <u>subsection 2.1.1</u> initiated by the New Holders or the Original Holder, as the case may be, or a majority-in-interest of the Demand Requesting Holders (if any) shall have the right to withdraw from a Registration pursuant to such Demand Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter(s) (if any) of their intention to withdraw from such Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such Demand Registration. If a majority-in-interest of the Demanding Holders initiating a Demand Registration

or a majority-in-interest of the Demand Requesting Holders (if any) withdraws from a proposed offering pursuant to this <u>Section 2.1.5</u>, then such registration shall not count as a Demand Registration provided for in <u>Section 2.1</u>. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration prior to its withdrawal under this <u>subsection 2.1.5</u>.

Section 2.2. Piggyback Registration.

- 2.2.1 Piggyback Rights. If the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including, without limitation, pursuant to Section 2.1), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter(s), if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its reasonable best efforts to cause the managing Underwriter(s) of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.
- 2.2.2 Reduction of Piggyback Registration. If the managing Underwriter(s) in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of Ordinary Shares that the Company desires to sell, taken together with (i) the Ordinary Shares, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to Section 2.2 hereof, and (iii) the Ordinary Shares, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:
 - (i) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, Ordinary Shares or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, pro rata, based on the

- respective number of Registrable Securities that each Holder has so requested, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), Ordinary Shares, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities; and
- (ii) If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, Ordinary Shares or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1, pro rata based on the respective number of Registrable Securities that each Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Holders have requested to be included in such Underwritten Registration, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), Ordinary Shares or other equity securities has not been reached under the foregoing clauses (A), (B) and (C), Ordinary Shares or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.
- 2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this subsection 2.2.3.
- 2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a Registration pursuant to a Demand Registration effected under <u>Section 2.1</u> hereof, and there shall be no limit on the number of Piggyback Registrations.
- 2.2.5 <u>Right to Terminate Registration</u>. The Company shall have the right to terminate or withdraw any registration initiated by it under this <u>Section 2.2</u> prior to the effectiveness of such registration whether or not any Holder of Registrable Securities has elected to include securities in such registration.

Section 2.3. Resale Shelf Registration Rights

2.3.1 Registration Statement Covering Resale of Registrable Securities. The Company shall prepare and file or cause to be prepared and filed with the Commission, no later than forty-five (45) days following the Closing Date (the "Filing Deadline"), a Registration Statement for an offering to be made on a continuous basis pursuant to Rule 415 of the Securities Act or any successor thereto registering the resale from time to time by Holders of all of the Registrable Securities held by the Holders (the "Resale Shelf Registration Statement"). The Resale Shelf Registration Statement shall be on Form F-3 (or, if Form F-3 is not available to be used by the Company at such time, on Form F-1 or another appropriate form permitting Registration of such Registrable Securities for resale). If the Resale Shelf Registration Statement is initially filed on Form F-1 and thereafter the Company becomes eligible to use Form F-3 for secondary sales, the Company shall, as promptly as practicable, cause such Resale Shelf Registration Statement to be amended, or shall file a new replacement Resale Shelf Registration Statement, such that the Resale Shelf Registration Statement is on Form F-3. The Company shall use reasonable best efforts to cause the Resale Shelf Registration Statement to be declared effective as soon as possible after filing, but in no event later than sixty (60) days following the Filing Deadline (the "Effectiveness Deadline"); provided, however, that the Effectiveness Deadline shall be extended to one hundred twenty (120) days after the Filing Deadline if the Registration Statement is reviewed by, and receives comments from, the Commission; provided, however, that the Company's obligations to include the Registrable Securities held by a Holder in the Resale Shelf Registration Statement are contingent upon such Holder furnishing in writing to the Company such information regarding the Holder, the securities of the Company held by the Holder and the intended method of disposition of the Registrable Securities as shall be reasonably requested by the Company to effect the registration of the Registrable Securities, and the Holder shall execute such documents in connection with such registration as the Company may reasonably request that are customary of a selling stockholder in similar situations. Once effective, the Company shall use reasonable best efforts to keep the Resale Shelf Registration Statement and Prospectus included therein continuously effective and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, to ensure that another Registration Statement is available, under the Securities Act at all times until the earliest of (i) the date on which all Registrable Securities and other securities covered by such Registration Statement have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement and (ii) the date on which all Registrable Securities and other securities covered by such Registration Statement have ceased to be Registrable Securities. The Registration Statement filed with the Commission pursuant to this subsection 2.3.1 shall contain a prospectus in such form as to permit any Holder to sell such Registrable Securities pursuant to Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) at any time beginning on the effective date for such Registration Statement (subject to lock-up restrictions provided in Section 5.1 of this Agreement), and shall provide that such Registrable Securities may be sold pursuant to any method or combination of methods legally available to, and requested by, Holders.

2.3.2 Notification and Distribution of Materials. The Company shall notify the Holders in writing of the effectiveness of the Resale Shelf Registration Statement as soon as practicable, and in any event within one (1) Business Day after the Resale Shelf Registration Statement becomes effective, and shall furnish to them, without charge, such number of copies of the Resale Shelf Registration Statement (including any amendments, supplements and exhibits), the Prospectus contained therein (including each preliminary prospectus and all related amendments and supplements) and any documents incorporated by reference in the Resale Shelf Registration Statement or such other documents as the Holders may reasonably request in order to facilitate the sale of the Registrable Securities in the manner described in the Resale Shelf Registration Statement.

- 2.3.3 Amendments and Supplements. Subject to the provisions of Section 2.3.1 above, the Company shall promptly prepare and file with the Commission from time to time such amendments and supplements to the Resale Shelf Registration Statement and Prospectus used in connection therewith as may be necessary to keep the Resale Shelf Registration Statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all the Registrable Securities. If any Resale Shelf Registration Statement filed pursuant to Section 2.3.1 is filed on Form F-3 and thereafter the Company becomes ineligible to use Form F-3 for secondary sales, the Company shall promptly notify the Holders of such ineligibility and use its best efforts to file a shelf registration on an appropriate form as promptly as practicable to replace the shelf registration statement on Form F-3 and have such replacement Resale Shelf Registration Statement declared effective as promptly as practicable and to cause such replacement Resale Shelf Registration Statement is available or, if not available, that another Resale Shelf Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities; provided, however, that at any time the Company once again becomes eligible to use Form F-3, the Company shall cause such replacement Resale Shelf Registration Statement to be amended, or shall file a new replacement Resale Shelf Registration Statement is once again on Form F-3.
- 2.3.4 SEC Cutback. Notwithstanding the registration obligations set forth in this Section 2.3, in the event the Commission informs the Company that all of the Registrable Securities cannot, as a result of the application of Rule 415, be registered for resale as a secondary offering on a single registration statement, the Company agrees to promptly (i) inform each of the Holders thereof and use its reasonable best efforts to file amendments to the Resale Shelf Registration Statement as required by the Commission and/or (ii) withdraw the Resale Shelf Registration Statement and file a new registration statement (a "New Registration Statement") on Form F-3, or if Form F-3 is not then available to the Company for such registration statement, on such other form available to register for resale the Registrable Securities as a secondary offering; provided, however, that prior to filing such amendment or New Registration Statement, the Company shall use its reasonable best efforts to advocate with the Commission for the registration of all of the Registrable Securities in accordance with any publicly-available written or oral guidance, comments, requirements or requests of the Commission staff (the "SEC Guidance"). Notwithstanding any other provision of this Agreement, if any SEC Guidance sets forth a limitation on the number of Registrable Securities permitted to be registered on a particular Registration Statement as a secondary offering (and notwithstanding that the Company used diligent efforts to advocate with the Commission for the registration of all or a greater number of Registrable Securities), unless otherwise directed in writing by a Holder as to further limit its Registrable Securities to be included on the Registration Statement, the number of Registrable Securities to be registered on such Registration Statement will be reduced on a pro rata basis based on the total number of Registrable Securities held by the Holders, subject to a determination by the Commission that certain Holders must be reduced first based on the number of Registrable Securities held by such Holders. In the event the Company amends the Resale Shelf Registration Statement or files a New Registration Statement, as the case may be, under clauses (i) or (ii) above, the Company will use its reasonable best efforts to file with the Commission, as promptly as allowed by the Commission or SEC Guidance provided to the Company or to registrants of securities in general, one or more registration statements on Form F-3 or such other form available to register for resale those Registrable Securities that were not registered for resale on the Resale Shelf Registration Statement, as amended, or the New Registration Statement.
- 2.3.5 <u>Underwritten Shelf Takedown</u>. At any time and from time to time after a Resale Shelf Registration Statement has been declared effective by the Commission, the Holders may request to sell all or any portion of the Registrable Securities in an underwritten offering that is registered pursuant to the Resale Shelf Registration Statement (each, an "**Underwritten Shelf Takedown**"); provided, however, that the Company shall only be obligated to effect an Underwritten Shelf Takedown if such offering shall include securities with a total offering price (including piggyback securities and before deduction of

underwriting discounts) reasonably expected to exceed, in the aggregate, \$50,000,000. All requests for Underwritten Shelf Takedowns shall be made by giving written notice to the Company at least ten (10) days prior to the public announcement of such Underwritten Shelf Takedown, which shall specify the approximate number of Registrable Securities proposed to be sold in the Underwritten Shelf Takedown and the expected price range (net of underwriting discounts and commissions) of such Underwritten Shelf Takedown. The Company shall include in any Underwritten Shelf Takedown the securities requested to be included by any Holder (each a "Takedown Requesting Holder") at least 48 hours prior to the public announcement of such Underwritten Shelf Takedown pursuant to written contractual piggyback registration rights of such Holder (including those set forth herein). All such Holders proposing to distribute their Registrable Securities through an Underwritten Shelf Takedown under this subsection 2.3.5 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Takedown Requesting Holders initiating the Underwritten Shelf Takedown.

- 2.3.6 Reduction of Underwritten Shelf Takedown. If the managing Underwriter(s) in an Underwritten Shelf Takedown, in good faith, advise the Company and the Takedown Requesting Holders in writing that the dollar amount or number of Registrable Securities that the Takedown Requesting Holders desire to sell, taken together with all other Ordinary Shares or other equity securities that the Company desires to sell, exceeds the Maximum Number of Securities, then the Company shall include in such Underwritten Shelf Takedown, as follows: (i) first, the Registrable Securities of the Takedown Requesting Holders, on a Pro Rata basis, that can be sold without exceeding the Maximum Number of Securities; and (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Ordinary Shares or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities.
 - 2.3.7 Registrations effected pursuant to this Section 2.3 shall be counted as Demand Registrations effected pursuant to Section 2.1.

Section 2.4. Restrictions on Registration Rights. Notwithstanding anything to the contrary contained herein, the Company shall not be obligated to (but may, at its sole option) file a Registration Statement pursuant to a Demand Registration request made under Section 2.1 if (A) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to subsection 2.1.1 and that the Company continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective (B) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (C) in the good faith judgment of the Board such Registration would be seriously detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be seriously detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement. In such event, the Company shall have the right to defer such filing for a period of not more than thirty (30) days; provided, however, that the Company shall not defer its obligation in this manner more than once in any 12-month period. Notwithstanding anything to the contrary contained in this Agreement, no Registration shall be effected or permitted and no Registration Statement shall become effective, with respect to any Registrable Securit

ARTICLE III COMPANY PROCEDURES

- **Section 3.1. General Procedures**. If at any time on or after the Effective Time the Company is required to effect the Registration of Registrable Securities, the Company shall use its reasonable best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:
- 3.1.1 prepare and file with the Commission as soon as practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;
- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by the Holders or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;
- 3.1.3 prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriter(s), if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4 prior to any public offering of Registrable Securities, use its reasonable best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;

- 3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;
- 3.1.8 advise each Holder of Registrable Securities covered by such Registration Statement, promptly after the Company receives notice thereof, of the time when such registration statement has been declared effective or a supplement to any Prospectus forming a part of such registration statement has been filed;
- 3.1.9 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus, furnish a copy thereof to each seller of such Registrable Securities or its counsel;
- 3.1.10 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;
- 3.1.11 permit a representative of the Holders, the Underwriter(s), if any, and any attorney or accountant retained by such Holders or Underwriter(s) to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter(s), attorney or accountant in connection with the Registration; <u>provided</u>, <u>however</u>, that such representatives or Underwriter(s) enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information;
- 3.1.12 obtain a "cold comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration, in customary form and covering such matters of the type customarily covered by "cold comfort" letters as the managing Underwriter(s) may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders and such managing Underwriter;
- 3.1.13 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriter(s), if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter(s) may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;
- 3.1.14 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter(s) of such offering;
- 3.1.15 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);

- 3.1.16 if a Registration, including an Underwritten Offering, involves the Registration of Registrable Securities involving gross proceeds in excess of \$50,000,000, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter(s) in any Underwritten Offering; and
- 3.1.17 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.
- Section 3.2. Registration Expenses. Including as set forth in Section 2.1.5, all Registration Expenses shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing the Holders.
- Section 3.3. Requirements for Participation in Underwritten Offerings. No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.
- Section 3.4. Suspension of Sales; Adverse Disclosure. The Company shall promptly notify each of the Holders in writing if a Registration Statement or Prospectus contains a Misstatement and, upon receipt of such written notice from the Company, each of the Holders shall forthwith discontinue disposition of Registrable Securities until he, she or it has received copies of a supplemented or amended Prospectus correcting the Misstatement, provided that the Company hereby covenants promptly to prepare and file any required supplement or amendment correcting any Misstatement promptly after the time of such notice and, if necessary, to request the immediate effectiveness thereof. If the filing, initial effectiveness or continued use of a Registration Statement or Prospectus included in any Registration Statement at any time (a) would require the Company to make an Adverse Disclosure, (b) would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, or (c) in the good faith judgment of the Board, which judgment shall be documented in writing and provided to the Holders in the form of a written certificate signed by the Chairman of the Board, such filing, initial effectiveness or continued use of a Registration Statement would be materially detrimental to the Company. The Company shall have the right to defer the filing, initial effectiveness or continued use of any Registration Statement pursuant to (a), (b) or (c) for a period of not more than thirty (30) days and the Company shall not defer any such filing, initial effectiveness or use of a Registration Statement pursuant to this Section 3.4 no more than twice or for more than a total of sixty (60) days (in each case counting deferrals initiated pursuant to (a), (b) and (c) in the aggregate) in any 12-month period.
- Section 3.5. Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell Ordinary Shares held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

Section 3.6. Limitations on Registration Rights. The Company shall not hereafter enter into any agreement with respect to its securities which is inconsistent with or violates the rights granted to the Holders of Registrable Securities in this Agreement and in the event of any conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.

ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

Section 4.1. Indemnification

- 4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and agents and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriter(s), their officers and directors and each person who controls (within the meaning of the Securities Act) such Underwriter(s) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.
- 4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls (within the meaning of the Securities Act) the Company against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriter(s), their officers, directors and each person who controls (within the meaning of the Securities Act) such Underwriter(s) to the same extent as provided in the foregoing with respect to indemnification of the Company.
- 4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided, however, that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with

counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld, conditioned or delayed). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution (pursuant to subsection 4.1.5) to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.
- 4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this subsection 4.1.5 were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this subsection 4.1.5. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this subsection 4.1.5 from any person who was not guilty of such fraudulent misrepresentation.

ARTICLE V LOCK-UP

Section 5.1. Lock-Up.

- 5.1.1 Except as permitted by Section 5.2, each New Holder shall not Transfer any Ordinary Shares beneficially owned or owned of record by such Holder until the earliest of: (i) the date that is one (1) year from the Closing Date, (ii) the last consecutive trading day where the sale price of the Ordinary Shares equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after the Closing Date, or (iii) such date on which the Company completes a liquidation, merger, stock exchange or other similar transaction that results in all of the Company's stockholders having the right to exchange their Ordinary Shares for cash, securities or other property (the "New Holder Lock-up Period").
- 5.1.2 Except as permitted by Section 5.2, the Original Holder shall not Transfer (A) any Private Placement Warrants for a period ending thirty (30) days after the Closing Date (the "Original Holder Warrant Lock-up Period") or (B) any Ordinary Shares beneficially owned or owned of record by the Original Holder until the earliest of: (i) the date that is one (1) year from the Closing Date, (ii) the last consecutive trading day where the sale price of the Ordinary Shares equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after the Closing Date, or (iii) such date on which the Company completes a liquidation, merger, stock exchange or other similar transaction that results in all of the Company's stockholders having the right to exchange their Ordinary Shares for cash, securities or other property (the "Original Holder Share Lock-up Period", and together with the Original Holder Warrants Lock-up Period, the "Original Holder Lock-up Period").

Section 5.2. Exceptions. The provisions of <u>Section 5.1</u> shall not apply to:

- 5.2.1 transactions relating to Ordinary Shares acquired in open market transactions;
- 5.2.2 Transfers of Ordinary Shares or any security convertible into or exercisable or exchangeable for Ordinary Shares as a bona fide gift;
- 5.2.3 Transfers of Ordinary Shares to a trust, or other entity formed for estate planning purposes for the primary benefit of the spouse, domestic partner, parent, sibling, child or grandchild of the undersigned or any other person with whom the undersigned has a relationship by blood, marriage or adoption not more remote than first cousin;
 - 5.2.4 Transfers by will or intestate succession upon the death of the undersigned;
 - 5.2.5 the Transfer of Ordinary Shares pursuant to a qualified domestic order or in connection with a divorce settlement;
- 5.2.6 if the undersigned is a corporation, partnership (whether general, limited or otherwise), limited liability company, trust or other business entity, (i) Transfers to another corporation, partnership, limited liability company, trust or other business entity that controls, is controlled by or is under common control or management with the undersigned, (ii) distributions of Ordinary Shares to partners, limited liability company members or stockholders of the undersigned;
 - 5.2.7 Transfers to the Company's officers, directors or their affiliates;
- 5.2.8 pledges of Ordinary Shares or other Registrable Securities as security or collateral in connection with any borrowing or the incurrence of any indebtedness by any Holder; <u>provided</u>, <u>however</u>, that such borrowing or incurrence of indebtedness is secured by a portfolio of assets or equity interests issued by multiple issuers;

- 5.2.9 Transfers pursuant to a bona fide third-party tender offer, merger, stock sale, recapitalization, consolidation or other transaction involving a Change in Control of the Company; <u>provided</u>, <u>however</u>, that in the event that such tender offer, merger, recapitalization, consolidation or other such transaction is not completed, the Ordinary Shares subject to this Agreement shall remain subject to this Agreement;
- 5.2.10 the establishment of a trading plan pursuant to Rule 10b5-1 promulgated under the Exchange Act; <u>provided</u>, <u>however</u>, that such plan does not provide for the Transfer of Ordinary Shares or any securities convertible into or exercisable or exchangeable for Ordinary Shares during the New Holder Lock-Up Period or Original Holder Lock-Up Period, as applicable;
- 5.2.11 Transfers of Ordinary Shares to satisfy tax withholding obligations in connection with the exercise of options to purchase Ordinary Shares or the vesting of stock-based awards;
- 5.2.12 Transfers of Ordinary Shares in payment on a "net exercise" or "cashless" basis of the exercise or purchase price with respect to the exercise of options to purchase Ordinary Shares;
 - 5.2.13 Transfers of Ordinary Shares purchased by any Holder in the PIPE; and
- 5.2.14 With respect to 5.1.1., after 180 days from the Closing Date, Transfers solely to the extent required to cover tax obligations of such New Holder or its direct and indirect shareholders.

provided, however, that in the case of any Transfer pursuant to Sections 5.2.2 through 5.2.7, each donee, distributee or other transferee shall agree in writing, in form and substance reasonably satisfactory to the Company, to be bound by the provisions of this Agreement.

Section 5.3. Early Release of Lock-Up Restrictions. In the event that any Holder is granted a discretionary waiver or termination of the restrictions set forth in Section 5.1 above, such discretionary release or waiver shall apply pro rata to all Holders based on the number of shares held.

ARTICLE VI TERMINATION

- **Section 6.1. Termination**. This Agreement shall terminate upon the earliest to occur of: (i) the termination of the Business Combination Agreement, and (ii) the date on which neither the Holders nor any of their permitted assignees hold any Registrable Securities.
- Section 6.2. Effect of Business Combination Termination. In the event of a termination of this Agreement as a result of the termination of the Business Combination Agreement, this Agreement shall become void and the Prior Agreement shall continue in full force and effect.

ARTICLE VIII GENERAL PROVISIONS

Section 7.1. Notices. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in person, by e-mail or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses or e-mail addresses (or at such other address or email address for a party as shall be specified in a notice given in accordance with this <u>Section 7.1.</u>):

If to the Company, to it at:

Wallbox N.V. Carrer del Foc, 68 Barcelona, Spain 08038 Attention: Enric Asunción Escorsa

Email: enric@wallbox.com

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 811 Main Street Suite 3700 Houston, TX 77002 United States Attention: Ryan Maierson Email: Ryan.Maierson@lw.com

If to a Holder, to the address or email address set forth for Holder on the signature page hereof.

Section 7.2. Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

Section 7.3. Entire Agreement; Assignment. This Agreement constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes all prior agreements and undertakings, both written and oral, among the parties, or any of them, with respect to the subject matter hereof. This Agreement shall not be assigned (whether pursuant to a merger, by operation of law or otherwise), by any party without the prior express written consent of the other parties hereto, except that a Holder may, without consent, assign such Holder's rights under this Agreement to any transferee of Ordinary Shares permitted under Sections 5.2.2-5.2.7 (such transferees, "Permitted Transferees").

Section 7.4. Parties in Interest. This Agreement shall be binding upon and inure solely to the benefit of each party hereto (and its respective permitted assigns), and nothing in this Agreement, express or implied, is intended to or shall confer upon any other person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

Section 7.5. Governing Law. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (I) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THIS AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN NEW YORK COUNTY IN THE STATE OF NEW YORK.

Section 7.6. Waiver of Jury Trial. EACH OF THE PARTIES HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT. EACH OF THE PARTIES HERETO (I) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THAT FOREGOING WAIVER AND (II) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY, AS APPLICABLE, BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 7.6.

Section 7.7. Headings; Interpretation. The descriptive headings contained in this Agreement are included for convenience of reference only and shall not affect in any way the meaning or interpretation of this Agreement. The parties have participated jointly in the negotiation and drafting of this Agreement. If any ambiguity or question of intent arises, this Agreement will be construed as if drafted jointly by the parties and no presumption or burden of proof will arise favoring or disfavoring any party because of the authorship of any provision of this Agreement. Unless the context of this Agreement clearly requires otherwise, use of the masculine gender shall include the feminine and neutral genders and vice versa, and the definitions of terms contained in this Agreement are applicable to the singular as well as the plural forms of such terms. The words "includes" or "including" shall mean "including without limitation." The words "hereof," "hereby," "herein," "hereunder" and similar terms in this Agreement shall refer to this Agreement as a whole and not any particular section or article in which such words appear, the word "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends and such phrase shall not mean simply "if." Any reference to a law shall include any rules and regulations promulgated thereunder, and shall mean such law as from time to time amended, modified or supplemented. References herein to any contract (including this Agreement) mean such contract as amended, supplemented or modified from time to time in accordance with the terms thereof.

Section 7.8. Counterparts. This Agreement may be executed and delivered (including by facsimile or portable document format (pdf) transmission) in counterparts, and by the different parties hereto in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.

Section 7.9. Specific Performance. The parties hereto agree that irreparable damage would occur in the event any provision of this Agreement was not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof, in addition to any other remedy at law or in equity. Each of the parties hereby further waives (a) any defense in any action for specific performance that a remedy at law would be adequate and (b) any requirement under any Law to post security or a bond as a prerequisite to obtaining equitable relief.

Section 7.10. Expenses. Except as otherwise provided herein, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such costs and expenses, whether or not the transactions contemplated hereby are consummated.

Section 7.11. Amendment. This Agreement may be amended in writing by the parties hereto at any time prior to the Effective Time. Following the Effective Time, this Agreement may not be amended except by an instrument in writing signed by (i) the Company, (ii) the Original Holder (provided the Original Holder or its Permitted Transferee(s) holds Registrable Securities at the time of such amendment), and (iii) New Holders holding at least a majority in interest of the then-outstanding number of Registrable Securities held by all New Holders (provided the New Holders or their Permitted Transferees hold Registrable Securities at the time of such amendment).

Section 7.12. Waiver. At any time, (i) the Company may (a) extend the time for the performance of any obligation or other act of any Holder, (b) waive any inaccuracy in the representations and warranties of any Holder contained herein or in any document delivered by such Holder pursuant hereto and (c) waive compliance with any agreement of such Holder or any condition to its own obligations contained herein. At any time, (i) the Holders may (a) extend the time for the performance of any obligation or other act of the Company, (b) waive any inaccuracy in the representations and warranties of the Company contained herein or in any document delivered by the Company pursuant hereto and (c) waive compliance with any agreement of the Company or any condition to their own obligations contained herein. Any such extension or waiver shall be valid if set forth in an instrument in writing signed by the party or parties to be bound thereby.

Section 7.13. Further Assurances. At the request of the Company, in the case of any Holder, or at the request of any Holder, in the case of the Company, and without further consideration, each party shall execute and deliver or cause to be executed and delivered such additional documents and instruments and take such further action as may be reasonably necessary to consummate the transactions contemplated by this Agreement.

Section 7.14. No Strict Construction. The language used in this Agreement shall be deemed to be the language chosen by the parties to express their mutual intent and no rule of strict construction shall be applied against any party.

(Next Page is Signature Page)

COMPANY:

Wallbox N.V.

By: /s/ Enric Asunción Escorsa

Name: Enric Asunción Escorsa Title: Chief Executive Officer

[Signature Page to Registration Rights Agreement]

ORIGINAL HOLDER:

KENSINGTON CAPITAL SPONSOR II LLC

By: Kensington Capital Partners, LLC

Its: Managing Member

By: /s/ Justin Mirro

Name: Justin Mirro Title: Managing Member

Address:

Email:

HOLDER:

INVERSIONES FINANCIERAS PERSEO S.L.

By: /s/ Diego Diaz /s/ Javier Salazar

Print Name:

Diego Diaz Javier Salazar

Title: Co-managing Director Co-managing Director

Address: Plaza Euskadi 5, Bilbao, Spain

Email: mluengo@iberdrola.es

HOLDER:

Leandro Martĺn Sigman Gold

By: /s/ Leandro Sigman

Print

Name: Leandro Sigman

Title:

Address: C/ Manuel Pombo Angulo 28, 3ª Planta

Madrid (Spain)

Email: Ana.Mondedeu@InsudPharma.com

HOLDER:

Mingkiri S.L.

By: /s/ Marc Sabe

Print

Name: Marc Sabe

Title: Authorized Representative

Address: Marques de Sentmenat 97Barcelona

(Spain)

Email: msabe@eurofred.com

HOLDER:

FPCI SINO French Innovation Fund II

By: /s/ Jacobo Abitbol

Print Cathay Innovation SAS, duly represented

Name: by Mr. Jacobo Abitbol Title: Authorized Signatory

Address: 52 rue d' Anjou, Paris (France)

Email: jacky.abitbol@cathay.fr

HOLDER:

ORILLA ASSET MANAGEMENT, S.L.

By: /s/ Francisco Jose Riberas Mera

Print

Name: Francisco Jose Riberas Mera

Title: Sole Director

Address: Alfonso XII, 16 Madrid (Spain)

Email: regues@gestamp.com

HOLDER:

Aleix Rull Sanahuja

By: /s/ Aleix Rull

Print

Name: Aleix Rull

Title:

Address: E1Catllar, at Avenida Catalunya 14

Tarragona (Spain)

Email: aleix@wallbox.com

HOLDER:

AM GESTIÓ S.L.

By: /s/ Javier Alonso Martín

Print

Name: Javier Alonso Martín Title: Authorized Signatory

Address: Calle Rosselló nº 224, 3ºA

Barcelona (Spain)

Email: xavi@amgestio.es

HOLDER:

Black Label Equity I SCR S.A.

By: /s/ José Luis Diaz

Print

Name: José Luis Diaz Title: Authorized Signatory

Address: Plaza de la Independencia 6

Madrid (Spain)

Email: lsanchezalciturri@labelinvestments.com

HOLDER:

Carlos Torres Vila

By: /s/ Carlos Torres

Print

Name: Carlos Torres

Title:

Address: C/ Azul núm. 4 Madrid (Spain)

Email: ctvila@gmail.com

HOLDER:

Consilium S.L.

By: /s/ Marc Puig Guasch

Print Name:

Marc Puig Guasch

Title: Jointly Authorized Signatory

Address: Entenza 325 9º planta Barcelona Spain)

Email: marc.puig@puig.es

HOLDER:

Consilium S.L.

By: /s/ Marian Puig Guasch

Print

Name: Marian Puig Guasch

Title: Jointly Authorized Signatory

Address: Entenza 325 9º planta Barcelona Spain)

Email: marian.puig.jr@isdin.com

HOLDER:

Copec Overseas SPA

By: /s/ Juan Carlos Balmaceda

Print

Name: Juan Carlos Balmaceda Title: Legal Representative

Calle Isidora Goyenechea 2,915, Las

Address: Condes, Santiago (Chile)

Email: bwalsh@windventures.vc

HOLDER:

Copec Overseas SPA

By: /s/ Leonardo Ljubetic

Print

Name: Leonardo Ljubetic Title: Legal Representative

Calle Isidora Goyenechea 2,915, Las

Address: Condes, Santiago (Chile)

Email: bwalsh@windventures.vc

HOLDER:

David Riba Magrazo

By: /s/ David Riba

Print

Name: David Riba

Title:

Address: C/ Paris 70^a Barcelona (Spain)

Email: david_riba_m@hotmail.com

HOLDER:

Eduard Castañeda Mañé

By: /s/ Eduard Castañeda Mañé

Print

Name: Eduard Castañeda Mañé

Title:

Address: Calle Florenci Vives, número 3, puerta 2ª

Tarragona (Spain)

Email: eduard@wallbox.com

HOLDER:

Endeavor Catalyst III, L.P.

By: /s/ Allen Taylor

Print

Name: Allen Taylor

Title: Authorized Signatory

Address: 900 Broadway, Suite 301, New York, NY 10003,

USA,

Email: maria.enriquez@endeavor.org

HOLDER:

Infisol 3000, S.L.

By: /s/ Pol Soler

Print

Name: Pol Soler Masferrer Title: Authorized Representative

Address: Calle Josep Irla i Bosch, números 1-3 Barcelona

(Spain)

Email: psoler@quadis.es

HOLDER:

Jaime Carvajal Urquijo

By: /s/ Jaime Carvajal Urquijo

Print

Name: Jaime Carvajal Urquijo

Title:

Address: Calle Hermanos Becquer 10Madrid (Spain)

Email: jcarvajalur7@gmail.com

HOLDER:

Jaume Santacana Senpau

By: /s/ Jaume Santacana

Print

Name: Jaume Santacana

Title:

Address: C/ Ferran Puig 74 3 ° 3ª Barcelona (Spain)

Email: jsantacana@eurofred.com

HOLDER:

José Maria Tarragó

By: /s/ José Maria Tarrago

Print

Name: José Maria Tarrago

Title:

Address: Cerdanyola del Valles, at Calle Can Miró, número

10 Barcelona (Spain)

Email: jose.maria.tarrago@gmail.com

HOLDER:

Jordi Cano Zamora

By: /s/ Jordi Cano Zamora

Print

Name: Jordi Cano Zamora

Title:

Address: Dr Ferran 7-9 08034 Barcelona (Spain)

Email: jordi@wallbox.com

HOLDER:

Jordi Lainz Gavalda

By: /s/ Jordi Lainz Gavalda

Print

Name: Jordi Lainz Gavalda

Title:

Address: C Foc 68 Barcelona (Spain)

Email: jordi.lainz@wallbox.com

HOLDER:

Juan Campmany Ibañez

By: /s/ Juan Campmany

Print

Name: Juan Campmany Ibañez

Title:

Address: Barcelona, at Calle Pau Alcover, 50

(Spain)

Email: jcampmany10@yahoo.es

HOLDER:

Kariega Ventures S.L.

By: /s/ Enric Asunción Escorsa

Print

Name: Enric Asunción Escorsa

Title: Sole Director

Address: Av. Diagonal 419, 4 Planta 08008

Barcelona (Spain)

Email: enric@wallbox.com

HOLDER:

Klaus Kersting

By: /s/ Klaus Kersting

Print

Name: Klaus Kersting

Title:

Address: Medinya C / Migdia 40 Girona (Spain)

Email: Klaus.Kersting@idiada.com

HOLDER:

Marina Planas López

By: /s/ Marina Planas López

Print

Name: Marina Planas López

Title:

Address: Castelldefels, Calle Manuel Girona, número 54

Barcelona (Spain)

Email: marina@the-ntwk.com

HOLDER:

Night's Watch Partners S.L.

By: /s/ Manuel Marín Berja

Print

Name: Manuel Marín Berja

Title: Sole Director

Address: Calle Sector Oficios, 23 Tres Cantos (Madrid)

Email: manu@aequusglobalcapital.com

HOLDER:

Oriol Riba Magrazo

By: /s/ Oriol Riba Magrazo

Print

Name: Oriol Riba Magrazo

Title:

Address: Aribau 218 08006 Barcelona (Spain)

Email: oriol.riba@wallbox.com

HOLDER:

Seaya Ventures II, Fondo De Capital Riesgo

By: /s/ Beatriz Gonzalez Ordoñez

Print

Name: Beatriz Gonzalez Ordoñez Title: Authorized Signatory

Address: Calle Alcalá, número 54 Madrid (Spain)

Email: bg@seayaventures.com

HOLDER:

Tarci Tech S.L.

By: /s/ José Maria Tarragó Pujol

Print

Name: José Maria Tarragó Pujol Title: Joint and several director

Address: Calle Fernando Puig 83 Barcelona (Spain)

Email: jose.maria.tarrago@gmail.com

HOLDER:

Wabisabi Inversión y Servicios S.L.

By: /s/ Javier Monzón De Caceres

Print

Name: Javier Monzón De Caceres Title: Joint and Several Director

Address: C/ Segre 16 Madrid (Spain)

Email: jm@wabisabix.com

Original Holder

Kensington Capital Sponsor II LLC

New Holders

	Number of Shares	
Name of Holder	Class A	Class B
Kariega Ventures, S.L.	-	18,618,950
Mingkiri, S.L.	14,992,038	
Infisol 3000, S.L.	12,952,774	
Inversiones Financieras Perseo, S.L.	16,697,530	
Seaya Ventures II, Fondo de Capital Riesgo	11,505,865	
Black Label Equity I SCR, S.A.	9,110,175	
AM Gestió, S.L.	7,689,293	
FPCI SINO French Innovation Fund II	7,512,888	
Copec Overseas SpA	4,434,713	
Consilium, S.L.	2,907,554	
Mr. Eduard Castañeda Mañe	-	4,631,843
Mr. Juan Campmany Ibañez	2,602,700	
Tarci Tech, S.L.	2,281,219	
Mr. Jordi Cano Zamora	2,173,737	
Night's Watch Partners, S.L.	1,890,572	
Orilla Asset Management, S.L.	2,778,142	
Catalyst	1,320,629	
Mr. José Tarrago Pujol	915,765	
Mr. Jaume Santacana Senpau	771,170	
Mr. Oriol Riba Magrazo	747,071	
Mr. David Riba Magrazo	578,377	
Ms. María Planas Lopez	554,278	
Mr. Aleix Rull Sanahuja	433,783	
Mr. Carlos Torres Vila	381,006	
Mr. Klaus Kersting	362,450	
Mr. Leandro Sigman Gold	339,074	
Mr. Jordi Lainz Gavalda	291,116	
Wabisabi Inversión y Servicios, S.L.	289,429	
Mr. Jaime Carvajal Urquijo	265,089	
Total	106,778,437	23,250,793

Final Form

SUBSCRIPTION AGREEMENT

Kensington Capital Acquisition Corp. II 1400 Old Country Road, Suite 301 Westbury, NY 11590 Wallbox B.V. Carrer del Foc, 68 Barcelona, Spain 08038

Ladies and Gentlemen:

In connection with the proposed business combination (the "Transaction") pursuant to the Business Combination Agreement (as it may be amended, restated or otherwise modified from time to time, the "Transaction Agreement"), dated as of June 9, 2021, among Kensington Capital Acquisition Corp. II, a Delaware corporation (the "SPAC"), Wallbox B.V., a private limited liability company (besloten vennootschap met beperkte aansprakelijkheid) organized under the laws of the Netherlands (to be converted into a limited liability company (naamloze vennootschap) prior to the Subscription Closing), having its official seat in Amsterdam, the Netherlands, and registered with the Dutch trade register under number 83012559 (the "Company"), Orion Merger Sub Corp., a Delaware corporation and a wholly-owned subsidiary of the Company ("Merger Sub"), Wall Box Chargers, S.L., a company organized under the laws of Spain ("Wallbox"), and the other parties thereto, pursuant to which, among other things, (i) the shareholders of Wallbox will contribute their shares of Wallbox to the Company in exchange for the issue of the Company's ordinary shares ("Ordinary Shares") to such shareholders, with Wallbox becoming a wholly-owned subsidiary of the Company, (ii) the Merger Sub will merge with and into the SPAC, with the SPAC as the surviving company of the merger and becoming a wholly-owned subsidiary of the Company, (iii) securities of the SPAC will be automatically cancelled and reissued (the "New SPAC Securities"), which securities will be exchanged with an exchange agent (the "Exchange Agent") for the right to receive securities of the Company, and (iv) the Exchange Agent will contribute the New SPAC Securities to the Company, which in exchange will issue securities of the Company to the Exchange Agent for the benefit of the securityholders of the SPAC, each of the undersigned desires to subscribe for and accept from the Company, and the Company desires to issue to the undersigned, that number of Ordinary Shares set forth on the signature page hereto for a subscription price of \$10.00 per share (the "Per Share Price" and the aggregate of such Per Share Price for all Shares subscribed for by the undersigned being referred to herein as the "Subscription Price"), on the terms and subject to the conditions contained herein and in the Transaction Agreement. In connection with the Transaction, certain other institutional "accredited investors" (each, an "Other Subscriber") (as defined in rule 501 under the Securities Act of 1933, as amended (the "Securities Act")) have entered into separate subscription agreements with the SPAC and the Company (the "Other Subscription Agreements"), pursuant to which, among other things, such investors have, severally and not jointly, together with the undersigned pursuant to this Subscription Agreement, agreed to subscribe for and accept an aggregate of [] Ordinary Shares at the Per Share Price (each such investor, including each of the undersigned, a "Subscriber" and together, the "Subscribers"). Additionally, on June 9, 2021, certain subscribers entered into the subscription agreements dated as of such date (the "PIPE Subscription Agreements") with respect to the purchase of an aggregate of 10,000,000 Ordinary Shares for a subscription price of \$10.00 per share. In connection with the Transaction, the undersigned, the SPAC and the Company agree as follows:

- 1. <u>Subscription</u>. Subject to the terms and conditions of this Subscription Agreement, the undersigned hereby, severally and not jointly, irrevocably subscribes for and agrees to acquire from the Company such number of Ordinary Shares as is set forth on its respective signature page of this Subscription Agreement on the terms and subject to the conditions provided for herein (the "<u>Shares</u>"). The undersigned understands and agrees that the Company reserves the right to accept or reject the undersigned's subscription for the Shares for any reason or for no reason, in whole or in part, at any time prior to its acceptance by the Company, and the same shall be deemed to be accepted by the Company only when this Subscription Agreement is signed by a duly authorized person by or on behalf of the Company; the Company may do so in counterpart form. In the event of rejection of the subscription by the Company, the undersigned's payment hereunder will be promptly returned to the undersigned in proportion to the rejected portion of the subscription. In the event of rejection of the entire subscription by the Company or the termination of this subscription in accordance with the terms hereof, the undersigned's payment hereunder will be returned promptly to the undersigned along with this Subscription Agreement, and this Subscription Agreement shall be null and void and have no force or effect.
- 2. Closing. The closing of the issuance of the Shares contemplated hereby (the "Subscription Closing") is contingent upon the substantially concurrent consummation of the Transaction (the "Transaction Closing"). The Subscription Closing shall occur on the date of, and substantially concurrent with, the consummation of the Transaction Closing (the "Transaction Closing Date"). The Company represents and warrants that (i) the scheduled Transaction Closing Date is October 1, 2021 (which date may be delayed in the Company's discretion, in which case any such delay shall be communicated to the Subscriber), (ii) it reasonably expects all conditions to the Transaction Closing to be satisfied or waived on the scheduled Transaction Closing Date, and (iii) wire instructions for delivery of the Subscription Price by the undersigned have been provided supplementally. The undersigned shall deliver to Continental Stock Transfer & Trust Company, as escrow agent (the "Escrow Agent"), at least one (1) business day prior to the Transaction Closing Date (which shall be October 1, 2021, unless otherwise communicated to the Subscriber), the Subscription Price, which shall be held in a segregated escrow account for the benefit of the Subscribers (the "Escrow Account") until the Subscription Closing pursuant to the terms of a customary escrow agreement, which shall be on terms and conditions reasonably satisfactory to the undersigned (the "Escrow Agreement") to be entered into by the undersigned, the Company and the Escrow Agent, by wire transfer of United States dollars in immediately available funds to the account specified by the Company in the wire instructions; provided, however, that in the event the undersigned is not legally permitted to deliver the Subscription Price in accordance with this sentence or is otherwise expected by its primary regulator to deliver payment against delivery of the Shares, the undersigned shall instruct its custodian bank to deliver to the Company, by 10:00 a.m. (New York time) on the Transaction Closing Date, the Subscription Price by wire transfer of United States dollars in immediately available funds to the account specified by the Company in the wire instructions. On the Transaction Closing Date, upon

satisfaction (or, if applicable, waiver) of the conditions set forth in Section 3 hereof and prior to the release of the Subscription Price by the undersigned, the Company shall deliver to the undersigned (i) the Shares in book-entry form, free and clear of any liens or other restrictions whatsoever (other than those arising under state or federal securities laws as set forth herein), in the name of the undersigned (or its nominee in accordance with its delivery instructions) or to a custodian designated by the undersigned, as applicable, and (ii) a copy of the records of the Company's transfer agent (the "Transfer Agent") showing the undersigned (or such nominee or custodian) as the owner of the Shares on and as of the Transaction Closing Date; provided that, (x) if such book entry is made prior to the Company's receipt of the Subscription Price from the undersigned and (y) such Subscription Price is not received by the Company on the Transaction Closing Date, then without limiting any rights of any party under this Subscription Agreement, the Company may, without any action of the undersigned, cause such book entries to be automatically cancelled, void and of no further force and effect. If the Transaction Closing does not occur within two (2) business days of October 1, 2021 (or such later date communicated to the Subscriber), the Escrow Agent (or the Company, as applicable) shall promptly (but not later than one (1) business day thereafter) return the Subscription Price has been released by the Escrow Agent or if the Subscription Price was paid directly to the Company) shall promptly (but not later than one (1) business day thereafter) return the Subscription Price to the undersigned by wire transfer of U.S. dollars in immediately available funds to the account specified by the undersigned, and any book-entries shall be deemed cancelled.

Notwithstanding anything to the contrary in Section 9 hereof, if this Subscription Agreement terminates following the delivery by the undersigned of the Subscription Price for the Shares, the Escrow Agent (or the Company, if the Subscription Price was paid directly to the Company) shall promptly (but not later than one (1) business day thereafter) return the Subscription Price to the undersigned by wire transfer of U.S. dollars in immediately available funds to the account specified by the undersigned, without any deduction for or on account of any tax, withholding, charges, or set-off, whether or not the Transaction Closing shall have occurred. Notwithstanding anything to the contrary in Section 9 hereof, if this Subscription Agreement terminates following the Transaction Closing, the undersigned shall promptly upon the return to the undersigned by wire transfer of U.S. dollars in immediately available funds to the account specified by the undersigned, without any deduction for or on account of any tax, withholding, charges, or set-off of the Subscription Price by the Escrow Agent or the Company, as applicable, transfer and deliver (and execute and deliver or cause to be executed and delivered such additional documents and instruments and take such further action as may be reasonably necessary to effectuate such transfer and delivery of) the Shares to the Company.

For the purposes of this Subscription Agreement, "business day" means any day other than a Saturday, Sunday or a day on which the Federal Reserve Bank of New York or any banks in the Netherlands are closed.

3. Closing Conditions.

- a. The obligations of the Company and the SPAC to consummate the transactions contemplated hereunder are subject to the conditions that, at the Subscription Closing:
 - i. all representations and warranties of the undersigned contained in this Subscription Agreement shall be true and correct in all material respects (other than representations and warranties that are qualified as to "materiality" or "Material Adverse Effect", which representations and warranties shall be true and correct in all respects) at and as of the Subscription Closing as though made on the Subscription Closing (except for those representations and warranties that speak as of a specific date, which shall be so true and correct in all material respects as of such specified date), and consummation of the Subscription Closing shall constitute a reaffirmation by the undersigned of each of the representations, warranties and agreements of such party contained in this Subscription Agreement as of the Subscription Closing; and
 - ii. the undersigned shall have performed or complied in all material respects with all agreements and covenants required by this Subscription Agreement to be performed or complied by the undersigned at or prior to the Subscription Closing.
- b. The obligations of the undersigned to consummate the transactions contemplated hereunder are subject to the conditions that, at the Subscription Closing:
 - i. all representations and warranties of the Company and the SPAC contained in this Subscription Agreement shall be true and correct in all material respects (other than representations and warranties that are qualified as to "materiality" or "Material Adverse Effect", which representations and warranties shall be true and correct in all respects) at and as of the Subscription Closing as though made on the Subscription Closing (except for those representations and warranties that speak as of a specific date, which shall be so true and correct in all material respects as of such specified date), other than representations and warranties that are qualified as to materiality or Material Adverse Effect, which representations and warranties shall be true and correct as of such specified date in all respects), and consummation of the Subscription Closing shall constitute a reaffirmation by the Company and the SPAC to the undersigned of each of the representations, warranties and agreements of such party contained in this Subscription Agreement as of the Subscription Closing;
 - ii. the Company and the SPAC shall have performed or complied in all material respects with all agreements and covenants required by this Subscription Agreement to be performed or complied by the Company and the SPAC, respectively, at or prior to the Subscription Closing;

- iii. the terms of the Transaction Agreement (as the same exists as of the date hereof in the form provided to the Subscriber) shall not have been amended in a manner that would reasonably be expected to materially and adversely affect the benefits the undersigned would reasonably expect to receive under this Subscription Agreement;
- iv. there shall have been no amendment, waiver or modification to one or more of the Other Subscription Agreements that reasonably would be expected to materially benefit one or more of the Other Subscribers thereunder unless the undersigned has been offered the same benefits; and
- v. the Shares shall have been approved for listing on the New York Stock Exchange (the "NYSE") or Nasdaq, subject to notice of issuance thereof.
- c. The obligations of each of the Company, the SPAC and the undersigned to consummate the transactions contemplated hereunder are subject to the conditions that, at the Subscription Closing:
 - i. no governmental authority shall have enacted, issued, promulgated, enforced or entered any judgment, order, law, rule or regulation (whether temporary, preliminary or permanent) which is then in effect and has the effect of making consummation of the transactions contemplated hereby illegal or otherwise restraining or prohibiting consummation of the transactions contemplated hereby, and no governmental authority shall have instituted or threatened in writing a proceeding seeking to impose any such restraint or prohibition;
 - ii. all conditions precedent to the Transaction Closing set forth in the Transaction Agreement, including all necessary approvals of the Company's shareholders and the SPAC's stockholders and regulatory approvals, if any, shall have been satisfied or waived as determined by the parties to the Transaction Agreement (other than those conditions which, by their nature, are to be satisfied at the closing of the Transaction, but subject to the satisfaction thereof at the Transaction Closing) and the Transaction Closing shall have been or will be consummated substantially concurrently with the Subscription Closing; and
 - iii. no suspension of the qualification of the Shares for offering or sale or trading in any jurisdiction, or initiation or threatening of any proceedings for any of such purposes, shall have occurred and be continuing.

- 4. <u>Further Assurances</u>. At the Subscription Closing, the parties hereto shall execute and deliver or cause to be executed and delivered such additional documents and instruments and take such further action as may be reasonably necessary to consummate the transactions contemplated by this Subscription Agreement.
 - 5. (i) SPAC Representations and Warranties. The SPAC represents and warrants to the undersigned that:
- a. The SPAC has been duly incorporated, is validly existing and is in good standing under the laws of the State of Delaware, with corporate power and authority to own, lease and operate its properties and conduct its business as presently conducted and to enter into, deliver and perform its obligations under this Subscription Agreement.
- b. This Subscription Agreement and the Transaction Agreement have been duly authorized, executed and delivered by the SPAC and are enforceable in accordance with their respective terms, except as may be limited or otherwise affected by (i) bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other laws relating to or affecting the rights of creditors generally, and (ii) principles of equity, whether considered at law or equity.
- c. The compliance by the SPAC with all of the provisions of this Subscription Agreement and the consummation of the transactions herein will not (i) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge or encumbrance upon any of the property or assets of the SPAC pursuant to the terms of any indenture, mortgage, deed of trust, loan agreement, lease, license or other agreement or instrument to which the SPAC is a party or by which the SPAC is bound or to which any of the property or assets of the SPAC are subject, which would have a material adverse effect on the business, properties, financial condition, stockholders' equity or results of operations of the SPAC (a "SPAC Material Adverse Effect") or materially affect the legal authority of the SPAC to comply in all material respects with the terms of this Subscription Agreement; (ii) result in any violation of the provisions of the organizational documents of the SPAC; or (iii) result in any violation of any statute or any judgment, order, rule or regulation of any court or governmental agency or body, domestic or foreign, having jurisdiction over the SPAC or any of its properties that would have a SPAC Material Adverse Effect or materially affect the validity of the Shares or the legal authority of the SPAC to comply with this Subscription Agreement.
- d. The SPAC is not required to obtain any consent, waiver, authorization or order of, give any notice to, or make any filing or registration with, any court or other federal, state, local or other governmental authority, self-regulatory organization (including NYSE) or other person in connection with the execution, delivery and performance of this Subscription Agreement (including, without limitation, the issuance of the Shares by the Company), other than (i) filings with the Securities and Exchange Commission (the "Commission"), (ii) filings required

by applicable state securities laws, (iii) filings required by NYSE, including with respect to obtaining stockholder approval, (iv) filings required to consummate the Transaction as provided under the definitive documents relating to the Transaction, and (v) where the failure of which to obtain would not be reasonably likely to have a SPAC Material Adverse Effect or have a material adverse effect on the SPAC's ability to consummate the transactions contemplated hereby, including the issuance and sale of the Shares by the Company.

- e. The SPAC has not received any written communication from a governmental entity that alleges that the SPAC is not in compliance with or is in default or violation of any applicable law, except where such non-compliance, default or violation would not be reasonably likely to have, individually or in the aggregate, a SPAC Material Adverse Effect.
- f. The issued and outstanding shares of Class A common stock, par value \$0.0001 per share ("Class A Common Stock"), of the SPAC are registered pursuant to Section 12(b) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and are listed for trading on NYSE under the symbol "KCAC". Except as disclosed in the SPAC's filings with the Commission, there is no suit, action, proceeding or investigation pending or, to the knowledge of the SPAC, threatened against the SPAC by NYSE or the Commission, respectively, to prohibit or terminate the listing of the SPAC's Class A Common Stock on NYSE or to deregister the Class A Common Stock under the Exchange Act. Other than in connection with the Transaction Closing, the SPAC has taken no action that is designed to terminate the registration of the Class A Common Stock under the Exchange Act.
- g. A copy of each form, report, statement, schedule, prospectus, proxy, registration statement and other document, if any, filed by the SPAC with the Commission since its initial registration of the Class A Common Stock under the Exchange Act (the "SEC Documents") is available to the undersigned via the Commission's EDGAR system. None of the SEC Documents contained, when filed or, if amended (such amendment to be deemed to supersede the applicable prior filings), as of the date of such amendment with respect to those disclosures that are amended, any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading; provided, however, that with respect to the information about the SPAC's affiliates contained in the proxy statement (the "Proxy Statement") contained in the Form F-4 registration statement regarding the Transaction or an SEC Document, the representation and warranty in this sentence is made to the SPAC's knowledge. The SPAC has timely filed each report, statement, schedule, prospectus, and registration statement that the SPAC was required to file with the Commission since its initial registration of the Class A Common Stock under the Exchange Act. The financial statements of the SPAC included in the SEC Documents comply in all material respects with applicable accounting requirements and the rules and regulations of the Commission with respect thereto as in effect at the time of filing and fairly present in all material respects the financial position of the SPAC as of and for the dates thereof and the results of operations and cash flows for the periods then ended, subject, in the case of unaudited statements, to normal, year-end audit adjustments. There are no material outstanding or unresolved comments in comment letters from the staff of the Division of Corporation Finance (the "Staff") of the Commission with respect to any of the SEC Documents. Notwithstanding the foregoing, any restatement, revision or other modification of the SEC's "Staff Statement on Accounting and Reporting Considerations for Warrants Issued by Special Purpose Acquisition Companies" on April 12, 2021 (the "SEC Statement") will not be deemed "material" for purposes of this Section 5(i).g.

h. Except for such matters as have not had and would not be reasonably likely to have, individually or in the aggregate, a SPAC Material Adverse Effect, there is no (i) action, suit, claim or other proceeding, in each case by or before any governmental authority pending, or, to the knowledge of the SPAC, threatened against the SPAC or (ii) judgment, decree, injunction, ruling or order of any governmental entity or arbitrator outstanding against the SPAC.

i. As of the date of this Subscription Agreement and as of immediately prior to the Subscription Closing, the authorized capital stock of the SPAC consists of 100,000,000 shares of Class A Common Stock, 10,000,000 shares of Class B common stock, par value \$0,0001 per share ("Class B Common Stock" and together with the Class A Common Stock, "Common Stock"), and 1,000,000 shares of the SPAC's preferred stock, par value \$0.0001 per share ("Preferred Stock"). As of the date of this Subscription Agreement: (i) 23,000,000 shares of Class A Common Stock (including [] such shares which have been properly tendered for redemption), 5,750,000 shares of Class B Common Stock and no shares of Preferred Stock are issued and outstanding; (ii) 5,750,000 public warrants and 8,800,000 private placement warrants (collectively, the "Warrants"), each exercisable to purchase one share of Class A Common Stock at \$11.50 per share, are issued and outstanding; and (iii) no shares of Common Stock are subject to issuance upon exercise of outstanding options. No Warrants are exercisable on or prior to the Subscription Closing. All (i) issued and outstanding Common Stock has been duly authorized and validly issued, is fully paid and non-assessable and is not subject to preemptive rights and (ii) outstanding Warrants have been duly authorized and validly issued, are fully paid and are not subject to preemptive rights. As of the date of this Subscription Agreement, except as set forth above and pursuant to the Transaction Agreement, there are no outstanding options, warrants or other rights to subscribe for, purchase or acquire from the SPAC any Common Stock or other equity interests in the SPAC (collectively, "SPAC Equity Interests") or securities convertible into or exchangeable or exercisable for SPAC Equity Interests. There are no securities or instruments issued by or to which the SPAC is a party containing antidilution or similar provisions that will be triggered by the issuance of (i) the Shares pursuant to this Subscription Agreement, or (ii) the Shares to be issued pursuant to any Other Subscription Agreement. As of the date of this Subscription Agreement, the SPAC has no subsidiaries and does not own, directly or indirectly, interests or investments (whether equity or debt) in any person, whether incorporated or unincorporated.

j. Other than the Other Subscription Agreements and the PIPE Subscription Agreements, the SPAC has not entered into any side letter or similar agreement with any Other Subscriber in connection with such Other Subscriber's direct or indirect investment in the Company or with or any other investor, and such Other Subscription Agreements have not been amended in any material respect following the date of this Subscription Agreement and reflect the same Per Share Price and terms that are no more favorable to any such Other Subscriber thereunder than the terms of this Subscription Agreement. The SPAC shall not release any Other Subscriber (or any of its affiliates) under any Other Subscription Agreement from any of its material

obligations thereunder or any other agreements (including side letters or similar agreements in respect thereof) with any Other Subscriber (or any of its affiliates) under any Other Subscription Agreement unless it offers a similar release to the undersigned with respect to any similar obligation it has hereunder. The SPAC has not agreed and will not agree to issue any warrants to any person in connection with the Transaction other than to exchange warrants outstanding on the date hereof for new warrants qualifying for classification as equity instruments (rather than liabilities).

- k. Neither the SPAC, nor any person acting on its behalf has, directly or indirectly, made any offers or sales of any SPAC or Company security or solicited any offers to buy any security, under circumstances that would adversely affect reliance by the SPAC or the Company on Rule 4(a)(2) under the Securities Act for the exemption from registration for the transactions contemplated hereby or would require registration of the Shares under the Securities Act.
- 1. Neither the SPAC nor any person acting on its behalf has conducted any general solicitation or general advertising (as those terms are used in Regulation D of the Securities Act) in connection with the offer or sale of any of the Shares, and assuming the accuracy of the representations and warranties of the undersigned herein and the representations and warranties of the Other Subscribers in the Other Subscription Agreements, the Shares are not being offered in a manner involving a public offering under, or in a distribution in violation of, the Securities Act, or any state securities laws.
- m. The SPAC is not, and immediately after receipt of payment for the Shares will not be, an "investment company" within the meaning of the Investment Company Act of 1940, as amended.
- n. Neither the SPAC, nor any person acting on its behalf has entered into any agreement or arrangement entitling any agent, broker, investment banker, financial advisor or other person to any broker's or finder's fee or any other commission or similar fee in connection with the transactions contemplated by this Subscription Agreement for which the undersigned could become liable. Other than the Placement Agents (as defined below), no person has been or will be paid (directly or indirectly) remuneration for solicitation of purchasers in connection with the sale of any Shares.
- o. The SPAC and, to the knowledge of the SPAC, the officers, directors, employees, and agents of the SPAC, in each case, acting on behalf of the SPAC, have been in compliance in all material respects with all applicable Anti-Corruption Laws (as herein defined), (ii) the SPAC has not been convicted of violating any Anti-Corruption Laws or, to the knowledge of the SPAC, subjected to any investigation by a governmental authority for violation of any applicable Anti-Corruption Laws, (iii) the SPAC has not conducted or initiated any internal investigation or made a voluntary, directed, or involuntary disclosure to any governmental authority regarding any alleged act or omission arising under or relating to any noncompliance with any Anti-Corruption Laws and (iv) the SPAC has not received any written notice or citation from a governmental authority for any actual or potential noncompliance with any applicable Anti-Corruption Laws. As used in this Subscription Agreement, "Anti-Corruption Laws" means any applicable laws relating to corruption and bribery, including the U.S. Foreign Corrupt Practices Act of 1977 (as amended), the UK Bribery Act 2010, and any similar law that prohibits bribery or corruption.

- (ii) Company Representations and Warranties. The Company represents and warrants to the undersigned that:
- a. The Company has been duly organized and is validly existing under the laws of the Netherlands, with corporate power and authority to own, lease and operate its properties and conduct its business as presently conducted and to enter into, deliver and perform its obligations under this Subscription Agreement.
- b. The Shares have been duly authorized and, when issued and delivered to the undersigned against full payment therefor in accordance with the terms of this Subscription Agreement, the Shares will be validly issued, fully paid and non-assessable, free and clear of any liens, charges or encumbrances (other than restrictions under applicable securities laws) and will not have been issued in violation of or subject to any preemptive or similar rights created under the Company's organizational documents (as in effect immediately prior to the Transaction Closing) or under the laws of the Netherlands, or otherwise.
- c. This Subscription Agreement and the Transaction Agreement have been duly authorized, executed and delivered by the Company and are enforceable in accordance with their respective terms, except as may be limited or otherwise affected by (i) bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other laws relating to or affecting the rights of creditors generally, and (ii) principles of equity, whether considered at law or equity.
- d. The issuance of the Shares and the compliance by the Company with all of the provisions of this Subscription Agreement and the consummation of the transactions herein will not (i) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge or encumbrance upon any of the property or assets of the Company pursuant to the terms of any indenture, mortgage, deed of trust, loan agreement, lease, license or other agreement or instrument to which the Company is a party or by which the Company is bound or to which any of the property or assets of the Company are subject, which would have a material adverse effect on the business, properties, financial condition, stockholders' equity or results of operations of the Company (a "Company Material Adverse Effect") or materially affect the validity of the Shares or the legal authority of the Company to comply in all material respects with the terms of this Subscription Agreement; (ii) result in any violation of the provisions of the organizational documents of the Company; or (iii) result in any violation of any statute or any judgment, order, rule or regulation of any court or governmental agency or body, domestic or foreign, having jurisdiction over the Company of its properties that would have a Company Material Adverse Effect or materially affect the validity of the Shares or the legal authority of the Company to comply with this Subscription Agreement.

- e. The Company is not required to obtain any consent, waiver, authorization or order of, give any notice to, or make any filing or registration with, any court or other federal, state, local or other governmental authority, self-regulatory organization or other person in connection with the execution, delivery and performance of this Subscription Agreement (including, without limitation, the issuance of the Shares), other than (i) filings with the Commission, (ii) filings required by applicable state securities laws, (iii) filings required to consummate the Transaction as provided under the definitive documents relating to the Transaction, (iv) filings required by NYSE or Nasdaq in connection with the listing of the Shares, and (v) where the failure of which to obtain would not reasonably be expected to have a Company Material Adverse Effect or have a material adverse effect on the Company's ability to consummate the transactions contemplated hereby, including the issuance and sale of the Shares.
- f. The Company has not received any written communication from a governmental entity that alleges that the Company is not in compliance with or is in default or violation of any applicable law, except where such non-compliance, default or violation would not be reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- g. As of the Transaction Closing, the Ordinary Shares will be registered pursuant to Section 12(b) of the Exchange Act and will be listed for trading on the NYSE or Nasdaq. There is no suit, action, proceeding or investigation pending or, to the knowledge of the Company, threatened against the Company by the NYSE, Nasdaq or the Commission with respect to any intention by such entity to deregister the Ordinary Shares or prohibit or terminate the listing of Ordinary Shares on the NYSE or Nasdaq.
- h. Assuming the accuracy of the undersigned's representations and warranties set forth in Section 6 of this Subscription Agreement, no registration under the Securities Act is required for the offer, issuance and sale of the Shares by the Company to the undersigned or to any Other Subscriber pursuant to the Other Subscription Agreements. The Shares offered hereby and pursuant to each Other Subscription Agreement (i) were not offered by any form of general solicitation or general advertising (within the meaning of Regulation D of the Securities Act) and (ii) are not being offered in a manner involving a public offering under, or in a distribution in violation of, the Securities Act, or any state securities laws.
- i. Except for such matters as have not had and would not be reasonably likely to have, individually or in the aggregate, a Company Material Adverse Effect, there is no (i) action, suit, claim or other proceeding, in each case by or before any governmental authority pending, or, to the knowledge of the Company, threatened against the Company or (ii) judgment, decree, injunction, ruling or order of any governmental entity or arbitrator outstanding against the Company.

- j. As of the date of this Subscription Agreement, the issued share capital of the Company consists of ten (10) Class A ordinary shares of the Company with a nominal value of 0.12 per share. As of the date of this Subscription Agreement, other than pursuant to (i) the Other Subscription Agreements, (ii) the Transaction Agreement and the (iii) PIPE Subscription Agreements, there are no outstanding options, warrants or other rights to subscribe for, purchase or acquire from the Company any Common Stock or other equity interests in the Company (collectively, "Company Equity Interests") or securities convertible into or exchangeable or exercisable for Company Equity Interests. Other than Merger Sub, as of the date of this Subscription Agreement, the Company has no subsidiaries and does not own, directly or indirectly, interests or investments (whether equity or debt) in any person, whether incorporated or unincorporated. There are no stockholder agreements, voting trusts or other agreements or understandings to which the Company is a party or by which it is bound relating to the voting of any securities of the Company other than as set forth in the SEC Documents and as contemplated by the Transaction Agreement.
- k. Other than the Other Subscription Agreements and the PIPE Subscription Agreements, the Company has not entered into any side letter or similar agreement with any Other Subscriber in connection with such Other Subscriber's direct or indirect investment in the Company or with or any other investor, and such Other Subscription Agreements have not been amended following the date of this Subscription Agreement and reflect the same Per Share Price and terms that are no more favorable to any such Other Subscriber thereunder than the terms of this Subscription Agreement. The Company shall not release any Other Subscriber (or any of its affiliates) under any Other Subscription Agreement from any of its material obligations thereunder or any other agreements (including side letters or similar agreements in respect thereof) with any Other Subscriber (or any of its affiliates) under any Other Subscription Agreement unless it offers a similar release to the undersigned with respect to any similar obligations it has hereunder. Other than pursuant to the Transaction Agreement, the Company has not agreed and will not agree to issue any warrants to any person in connection with the Transaction.
- 1. Neither the Company, nor any person acting on its behalf has, directly or indirectly, made any offers or sales of any SPAC or Company security or solicited any offers to buy any security, under circumstances that would adversely affect reliance by the Company or the SPAC on Rule 4(a)(2) under the Securities Act for the exemption from registration for the transactions contemplated hereby or would require registration of the Shares under the Securities Act.
- m. Neither the Company nor any person acting on its behalf has conducted any general solicitation or general advertising (as those terms are used in Regulation D of the Securities Act) in connection with the offer or sale of any of the Shares, and assuming the accuracy of the representations and warranties of the undersigned herein and the representations and warranties of the Other Subscribers in the Other Subscription Agreements, the Shares are not being offered in a manner involving a public offering under, or in a distribution in violation of, the Securities Act, or any state securities laws.
- n. The Company is not, and immediately after receipt of payment for the Shares will not be, an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

- o. Neither the Company, nor any person acting on its behalf has entered into any agreement or arrangement entitling any agent, broker, investment banker, financial advisor or other person to any broker's or finder's fee or any other commission or similar fee in connection with the transactions contemplated by this Subscription Agreement for which the undersigned could become liable. Other than the Placement Agents, no person has been or will be paid (directly or indirectly) remuneration for solicitation of purchasers in connection with the sale of any Shares.
- p. The Company, and, to the knowledge of the Company, the officers, directors, employees, and agents of the Company, in each case, acting on behalf of the Company, have been in compliance in all material respects with all applicable Anti-Corruption Laws, (ii) the Company has not been convicted of violating any Anti-Corruption Laws or, to the knowledge of the Company, subjected to any investigation by a governmental authority for violation of any applicable Anti-Corruption Laws, (iii) the Company has not conducted or initiated any internal investigation or made a voluntary, directed, or involuntary disclosure to any governmental authority regarding any alleged act or omission arising under or relating to any noncompliance with any Anti-Corruption Laws and (iv) the Company has not received any written notice or citation from a governmental authority for any actual or potential noncompliance with any applicable Anti-Corruption Laws.
- 6. <u>Subscriber Representations and Warranties</u>. Each of the undersigned, severally and not jointly, represents and warrants to the SPAC and the Company that:
- a. The undersigned is (i) a "qualified institutional buyer" (as defined under the Securities Act) or (ii) an institutional "accredited investor" (within the meaning of Rule 501(a) under the Securities Act), in each case, satisfying the requirements set forth on Schedule A, and is acquiring the Shares only for its own account and not for the account of others, and not on behalf of any other account or person or with a view to, or for offer or issuance in connection with, any distribution thereof in violation of the Securities Act (and shall provide the requested information on Schedule A following the signature page hereto). Accordingly, the undersigned understands that the offering of the Shares meets the exemptions from filing under FINRA Rule 5123(b)(1)(C) or (J). The undersigned is not an entity formed for the specific purpose of acquiring the Shares.
- b. The undersigned (i) is an institutional account as defined in FINRA Rule 4512(c), (ii) is a sophisticated investor, experienced in investing in private equity transactions and capable of evaluating investment risks independently, both in general and with regard to all transactions and investment strategies involving a security or securities and (iii) has exercised independent judgment in evaluating its participation in the subscription for and acceptance of the Shares. Accordingly, the undersigned understands that the offering of Shares to the undersigned hereunder meets (x) the exemptions from filing under FINRA Rule 5123(b)(1)(A) and (y) the institutional customer exemption under FINRA Rule 2111(b).

- c. The undersigned understands that the Shares are being offered in a transaction not involving any public offering within the meaning of the Securities Act and that the Shares have not been registered under the Securities Act. The undersigned understands that the Shares may not be resold, transferred, pledged or otherwise disposed of by the undersigned absent an effective registration statement under the Securities Act except (i) to the Company, (ii) to non-U.S. persons pursuant to offers and sales that occur outside the United States within the meaning of Regulation S under the Securities Act or (iii) pursuant to another applicable exemption from the registration requirements of the Securities Act, and in each of cases (i) and (iii) in accordance with any applicable securities laws of the states and other jurisdictions of the United States, and that any certificates or book-entry positions representing the Shares shall contain a legend to such effect. The undersigned acknowledges that the Shares will not initially be eligible for resale pursuant to Rule 144A promulgated under the Securities Act. The undersigned understands and agrees that the Shares will be subject to the foregoing transfer restrictions and, as a result of these transfer restrictions, the undersigned may not be able to readily resell the Shares and may be required to bear the financial risk of an investment in the Shares for an indefinite period of time. The undersigned understands that it has been advised to consult legal counsel prior to making any offer, resale, pledge or transfer of any of the Shares.
- d. The undersigned understands and agrees that the undersigned is subscribing and accepting Shares directly issued from the Company. The undersigned further acknowledges that there have been no representations, warranties, covenants and agreements made to the undersigned by the SPAC or the Company, or any of their officers or directors, expressly or by implication, other than those representations, warranties, covenants and agreements included in this Subscription Agreement.
- e. Either (i) the undersigned is not a Benefit Plan Investor as contemplated by the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), or (ii) the undersigned's subscription and acceptance and holding of the Shares will not constitute or result in a non-exempt prohibited transaction under Section 406 of ERISA, Section 4975 of the Internal Revenue Code of 1986, as amended, or any applicable similar law.
- f. The undersigned acknowledges and agrees that the undersigned has received, and has had an adequate opportunity to review, such financial and other information as the undersigned deems necessary in order to make an investment decision with respect to the Shares, the SPAC, the Company and (the business of) Wallbox and made its own assessment and is satisfied concerning the relevant tax and other economic considerations relevant to the undersigned's investment in the Shares. Without limiting the generality of the foregoing, the undersigned acknowledges that it has reviewed the documents provided to the undersigned by the SPAC and the Company. The undersigned represents and agrees that the undersigned and the undersigned's professional advisor(s), if any, have had the opportunity to ask such questions, receive such answers and obtain such information as the undersigned and such undersigned's professional advisor(s), if any, have deemed necessary to make an investment decision with respect to the Shares.
- g. The undersigned became aware of this offering of the Shares solely by means of direct contact between the undersigned and the SPAC, the Company or a representative of the SPAC or the Company, and the Shares were offered to the undersigned solely by direct contact between the undersigned and the SPAC, the Company or a representative of the SPAC or the Company. The undersigned did not become aware of this offering of the Shares, nor were the

Shares offered to the undersigned, by any other means. The undersigned acknowledges that the Company represents and warrants that the Shares (i) were not offered by any form of general solicitation or general advertising and (ii) to the knowledge of the undersigned, the Shares are not being offered in a manner involving a public offering under, or in a distribution in violation of, the Securities Act, or any state securities laws.

- h. The undersigned acknowledges that it is aware that there are substantial risks incident to the issuance and ownership of the Shares. The undersigned is able to fend for itself in the transactions contemplated herein, has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of an investment in the Shares and has the ability to bear the economic risks of such investment in the Shares and can afford a complete loss of such investment. The undersigned has sought such accounting, legal and tax advice as the undersigned has considered necessary to make an informed investment decision.
- i. In making its decision to subscribe for and accept the Shares, the undersigned has relied solely upon independent investigation made by the undersigned and the representations, warranties, covenants and agreements contained herein. Without limiting the generality of the foregoing, the undersigned has not relied on any statements or other information provided by the Placement Agents (as defined below) or any of their respective affiliates or their respective control persons, officers, directors or employees concerning the SPAC, the Company, Wallbox or the Shares or the offer and issuance of the Shares.
- j. The undersigned understands and agrees that no federal or state agency has passed upon or endorsed the merits of the offering of the Shares or made any findings or determination as to the fairness of this investment.
- k. The undersigned has been duly formed or incorporated and is validly existing in good standing under the laws of its jurisdiction of incorporation or formation.
- 1. The execution, delivery and performance by the undersigned of this Subscription Agreement are within the powers of the undersigned, have been duly authorized and will not constitute or result in a breach or default under or conflict with any order, ruling or regulation of any court or other tribunal or of any governmental commission or agency, or any agreement or other undertaking, to which the undersigned is a party or by which the undersigned is bound, which, in each case, would reasonably be expected to have a material adverse effect on the legal authority of the undersigned to enter into and timely perform its obligations under this Subscription Agreement, and, if the undersigned is not an individual, will not violate any provisions of the undersigned's charter documents, including, without limitation, its incorporation or formation papers, bylaws, indenture of trust or partnership or operating agreement, as may be applicable. The signature on this Subscription Agreement is genuine, and the signatory has been duly authorized to execute the same, and this Subscription Agreement constitutes a legal, valid and binding obligation of the undersigned, enforceable against the undersigned in accordance with its terms.

- m. Neither the due diligence investigation conducted by the undersigned in connection with making its decision to acquire the Shares nor any representations and warranties made by the undersigned herein shall modify, amend or affect the undersigned's right to rely on the truth, accuracy and completeness of the SPAC's and the Company's representations and warranties contained herein.
- n. The undersigned is not (i) a person or entity named on the List of Specially Designated Nationals and Blocked Persons administered by the U.S. Treasury Department's Office of Foreign Assets Control ("OFAC") or in any Executive Order issued by the President of the United States and administered by OFAC ("OFAC List"), or a person or entity prohibited by any OFAC sanctions program, (ii) a Designated National as defined in the Cuban Assets Control Regulations, 31 C.F.R. Part 515, or (iii) a non-U.S. shell bank or providing banking services indirectly to a non-U.S. shell bank (collectively, a "Prohibited Investor"). The undersigned agrees to provide law enforcement agencies, if requested thereby, such records as required by applicable law, provided, however, that the undersigned is permitted to do so under applicable law. If the undersigned is a financial institution subject to the Bank Secrecy Act (31 U.S.C. Section 5311 et seq.) (the "BSA"), as amended by the USA PATRIOT Act of 2001 (the "PATRIOT Act"), and its implementing regulations (collectively, the "BSA/PATRIOT Act"), the undersigned maintains policies and procedures reasonably designed to comply with applicable obligations under the BSA/PATRIOT Act. To the extent required, the undersigned maintains policies and procedures reasonably designed (a) for the screening of its investors against the OFAC sanctions programs, including the OFAC List, and (b) to ensure that the funds held by the undersigned and used to issue the Shares were legally derived.
- o. To the undersigned's knowledge, no disclosure or offering document has been prepared by UBS Securities LLC or Barclays Capital Inc. (together, the "Placement Agents") or any of their respective affiliates in connection with the offer and sale of the Shares.
- p. The Placement Agents and their respective directors, officers, employees, representatives and controlling persons have made no independent investigation with respect to the SPAC, the Company or the Shares or the accuracy, completeness or adequacy of any information supplied to the undersigned by the SPAC or the Company.
- q. As of the date of this Subscription Agreement the undersigned does not have, and during the thirty (30) day period immediately prior to the date of this Subscription Agreement the undersigned has not entered into, any "put equivalent position" as such term is defined in Rule 16a-1 under the Exchange Act or Short Sale positions with respect to the securities of the Company or the SPAC. For purposes of this Subscription Agreement, "Short Sales" shall include, without limitation, all "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Exchange Act, and all types of direct and indirect stock pledges (other than pledges in the ordinary course of business as part of prime brokerage arrangements), forward sale contracts, options, puts, calls, swaps and similar arrangements (including on a total return basis), and sales and other transactions through non-U.S. broker dealers or foreign regulated brokers. Notwithstanding the foregoing, in case the undersigned is a multi-managed investment vehicle whereby separate portfolio managers manage separate portions of such Subscriber's assets, the representation set forth above in this paragraph shall only apply with respect to the portion of assets managed by the portfolio manager that made the investment decision to purchase the Shares covered by this Subscription Agreement.

- r. In connection with the issue and purchase of the Shares, the Placement Agents have not acted as the undersigned's financial advisor or fiduciary or as an underwriter, initial purchaser, dealer or in any other such capacity. The Placement Agents shall not, nor shall any of their respective affiliates or their respective control persons, officers, directors or employees, be liable to the undersigned for any action heretofore or hereafter taken or omitted to be taken by any of them in connection with the undersigned's subscription and acceptance of the Shares. The Subscriber agrees and acknowledges that, and unconditionally waives any conflicts of interest with respect to the fact that, (i) the Placement Agents are acting as the Company's placement agents in connection with the transactions contemplated by this Subscription Agreement, (ii) Barclays Capital Inc. is acting as advisor to Wallbox in connection with the Transaction, (iii) UBS Securities LLC is acting as advisor to the SPAC in connection with the Transaction and (iv) neither Placement Agent is and neither has acted as the Subscriber's financial advisor or fiduciary.
- s. Neither Placement Agent has made, nor will it make, any representation or warranty, whether express or implied, of any kind or character, nor has it provided any advice or recommendation to the undersigned in connection with the transactions contemplated hereby. The Placement Agents will have no responsibility with respect to (A) any representations, warranties or agreements made by any person or entity under or in connection with the transactions contemplated hereby or any of the documents furnished pursuant thereto or in connection therewith, or the execution, legality, validity or enforceability (with respect to any person) of any thereof, or (B) the financial condition, business, or any other matter concerning the Company or the transactions contemplated hereby.
- t. If the undersigned is a resident of Canada, the undersigned hereby declares, represents, warrants and agrees as set forth in the attached Schedule B.
- u. The Subscriber acknowledges and agrees that the SPAC continues to review the SEC Statement and its implications, including on the financial statements and other information included in its SEC Documents, and any restatement, revision or other modification of the SEC Documents relating to or arising from such review, any subsequent related agreements or other guidance from the Staff of the SEC shall be deemed not material for purposes of Section 5(i).g.
- 7. Additional Subscriber Agreement. The undersigned hereby agrees that, from the date of this Subscription Agreement and until the Subscription Closing, no person or entity, while acting in connection with this Transaction and on behalf of the undersigned or any of its controlled affiliates or pursuant to any understanding in connection with this Transaction with the undersigned or any of its controlled affiliates, will engage in any Short Sales with respect to securities of the Company or the SPAC. For purposes hereof, "Short Sales" shall include, without limitation, all "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Exchange Act. Solely for purposes of this paragraph, subject to the undersigned's compliance with its obligations under the U.S. federal securities laws and the undersigned's internal policies, the

restrictions set forth above in this paragraph shall not apply to any employees, subsidiaries, desks, groups, or affiliates of the undersigned that are effectively walled off by appropriate "fire wall" information barriers approved by the undersigned's legal or compliance department. Notwithstanding the foregoing, if the undersigned is a multi-managed investment vehicle whereby separate portfolio managers manage separate portions of such Subscriber's assets, this Section 7 shall only apply with respect to the portion of assets managed by the portfolio manager that made the investment decision to purchase the Shares covered by this Subscription Agreement.

8. Registration Rights.

a. In the event that the Shares are not registered in connection with the consummation of the Transaction, the Company agrees that, within thirty (30) calendar days after the consummation of the Transaction (the "Filing Deadline"), the Company will file with the Commission (at the Company's sole cost and expense) a registration statement (the "Registration Statement") registering the resale of the Shares, and the Company shall use its commercially reasonable efforts to have the Registration Statement declared effective as soon as practicable after the filing thereof, but no later than the earlier of (i) 60 calendar days (or 120 calendar days if the Commission notifies the Company that it will "review" the Registration Statement) following the Filing Deadline, and (ii) five (5) business days after the Company is notified (orally or in writing, whichever is earlier) by the Commission that the Registration Statement will not be "reviewed" or will not be subject to further review (such earlier date, the "Effectiveness Date"); provided, however, that the Company's obligations to include the Shares in the Registration Statement are contingent upon the undersigned furnishing in writing to the Company such information regarding the undersigned, the securities of the Company held by the undersigned and the intended method of disposition of the Shares as shall be reasonably requested in writing by the Company to effect the registration of the Shares, and the undersigned shall execute such documents in connection with such registration as the Company may reasonably request that are customary of a selling stockholder in similar situations, including providing that the Company shall be entitled to postpone and suspend the use of the Registration Statement as permitted hereunder; provided, further, however, that the undersigned shall not in connection with the foregoing be required to execute any lock-up or similar agreement or otherwise be subject to any contractual restriction on the ability to transfer the Shares. With respect to the information to be provided by the undersigned pursuant to this Section 8, the Company shall request such information at least ten (10) business days prior to the anticipated initial filing date of the Registration Statement. The Company will provide a draft of the Registration Statement to the undersigned for review at least two (2) business days in advance of its anticipated initial filing date. Notwithstanding the foregoing, if the Commission prevents the Company from including any or all of the shares proposed to be registered under the Registration Statement due to limitations on the use of Rule 415 of the Securities Act for the resale of the Shares by the applicable stockholders or otherwise, such Registration Statement shall register for resale such number of Shares which is equal to the maximum number of Shares as is permitted by the Commission. In such event, the number of Shares to be registered for each selling stockholder named in the Registration Statement shall be reduced pro rata among all such selling stockholders, and as promptly as practicable after being permitted to register additional Shares under Rule 415 under the Securities Act, the Company shall file a new Registration Statement to register such Shares not included in the initial Registration

Statement and cause such Registration Statement to become effective as promptly as practicable consistent with the terms of this Section 8. In no event shall the undersigned be identified as a statutory underwriter in the Registration Statement unless in response to a comment or request from the staff of the Commission or another regulatory agency; provided, however, that if the Commission requests that the undersigned be identified as a statutory underwriter in the Registration Statement, the undersigned will have an opportunity to withdraw from the Registration Statement. The Company will use its commercially reasonable efforts to maintain the continuous effectiveness of the Registration Statement until the earliest of (i) the date on which the Shares subscribed for by the undersigned hereunder may be resold without volume or manner of sale limitations pursuant to Rule 144 promulgated under the Securities Act and without the requirement for the Company to be in compliance with the current public information required under Rule 144(c)(2) (or Rule 144(i)(2), if applicable), (ii) the date on which all Shares subscribed for by the undersigned hereunder have actually been sold and (iii) the date which is three (3) years after the initial Registration Statement filed hereunder is declared effective (the "Effectiveness Period"). For as long as the Registration Statement shall remain effective pursuant to the immediately preceding sentence, the Company will use commercially reasonable efforts to file all reports, and provide all customary and reasonable cooperation, necessary to enable the undersigned to resell the Shares pursuant to the Registration Statement or Rule 144 of the Securities Act (when Rule 144 of the Securities Act becomes available to the undersigned), as applicable, qualify the Shares for listing on NYSE, Nasdaq or other applicable stock exchange on which the Ordinary Shares are then listed, and update or amend the Registration Statement as necessary to include the Shares. For purposes of clarification, any failure by the Company to file the Registration Statement by the Filing Deadline or to effect such Registration Statement by the Effectiveness Date shall not otherwise relieve the Company of its obligations to file or effect the Registration Statement set forth in this Section 8. The undersigned shall not be entitled to use the Registration Statement for an underwritten offering of Shares and notwithstanding anything to the contrary in this Subscription Agreement, the Company shall not have any obligation to prepare any prospectus supplement, participate in any due diligence, execute any agreements or certificates or deliver legal opinions or obtain comfort letters in connection with any sales of the Shares under the Registration Statement. For purposes of this Section 8, "Shares" shall mean, as of any date of determination, the Shares acquired by the undersigned pursuant to this Subscription Agreement and any other equity security issued or issuable with respect to such Shares by way of stock split, dividend, distribution, recapitalization, merger, exchange, replacement or similar event, and "undersigned" shall include any affiliate of the undersigned to which the rights under this Section 8 have been duly assigned.

b. In the case of the registration, qualification, exemption or compliance effected by the Company pursuant to this Subscription Agreement, the Company shall inform the undersigned as to the status of such registration, qualification, exemption and compliance. At its expense the Company shall:

i. advise the undersigned within two (2) business days:

- (1) when a Registration Statement or any amendment thereto has been filed with the Commission and when such Registration Statement or any post-effective amendment thereto has become effective;
- (2) of any request by the Commission for amendments or supplements to any Registration Statement or the prospectus included therein or for additional information;
- (3) of the issuance by the Commission of any stop order suspending the effectiveness of any Registration Statement or the initiation of any proceedings for such purpose;
- (4) of the receipt by the Company of any notification with respect to the suspension of the qualification of the Shares included therein for sale in any jurisdiction or the initiation or threatening of any proceeding for such purpose; and
- (5) subject to the provisions in this Subscription Agreement, of the occurrence of any event that requires the making of any changes in any Registration Statement or prospectus so that, as of such date, the statements therein are not misleading and do not omit to state a material fact required to be stated therein or necessary to make the statements therein (in the case of a prospectus, in the light of the circumstances under which they were made) not misleading.

Notwithstanding anything to the contrary set forth herein, the Company shall not, when so advising the undersigned of such events, provide the undersigned with any material, nonpublic information regarding the Company other than to the extent that providing notice to the undersigned of the occurrence of the events listed in (1) through (5) above constitutes material, nonpublic information regarding the Company and the undersigned is notified that such events are material, nonpublic information at the time of notification;

- use commercially reasonable efforts to obtain the withdrawal of any order suspending the effectiveness of any Registration Statement as soon as reasonably practicable;
- iii. upon the occurrence of any event contemplated in Section 8.b.i(5) above, except for such times as the Company is permitted hereunder to suspend, and has suspended, the use of a prospectus forming part of a Registration Statement, the Company shall use its commercially reasonable efforts to as soon as reasonably practicable prepare a post-effective amendment to such Registration Statement or a supplement to the related prospectus, or file any other required document so that, as thereafter delivered to purchasers of the Shares included therein, such prospectus will not include any untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading;

- iv. use its commercially reasonable efforts to cause all Shares to be listed on each securities exchange or market, if any, on which the Ordinary Shares have been listed;
- v. use its commercially reasonable efforts to file all reports and other materials required to be filed by the Exchange Act until the expiry of the Effectiveness Period; and
- vi. if in the opinion of counsel to the Company, it is then permissible to remove the restrictive legend from the Shares pursuant to Rule 144 under the Securities Act, then at Subscriber's request, the Company will request its transfer agent to remove the legend set forth in Section 6.c. above. In the event that the undersigned and its broker(s) provide any certifications requested by the Company or its counsel, the Company shall use its commercially reasonable efforts to have the legend removal referenced above apply to all shares held by the Subscriber in a single transaction.
- c. Notwithstanding anything to the contrary in this Subscription Agreement, the Company shall be entitled to delay or postpone the effectiveness of the Registration Statement, and from time to time to require any Subscriber not to sell under the Registration Statement or to suspend the effectiveness thereof, if the negotiation or consummation of a transaction by the Company or its subsidiaries is pending or an event has occurred, which negotiation, consummation or event, the Company's board of directors reasonably believes, upon the advice of legal counsel, would require additional disclosure by the Company in the Registration Statement of material information that the Company has a bona fide business purpose for keeping confidential and the non-disclosure of which in the Registration Statement would be expected, in the reasonable determination of the Company's board of directors, upon the advice of legal counsel, to cause the Registration Statement to fail to comply with applicable disclosure requirements (each such circumstance, a "Suspension Event"); provided, however, that (x) the Company may not delay or suspend the Registration Statement on more than two (2) occasions, for more than sixty (60) consecutive calendar days, or for more than ninety (90) total calendar days, in each case during any twelve-month period and (y) the Company shall use commercially reasonable efforts to make such registration statement available for the sale by the undersigned of such securities as soon as practicable thereafter. Upon receipt of any written notice from the Company of the happening of any Suspension Event (which notice shall not contain material non-public information) during the period that the Registration Statement is effective or if as a result of a Suspension Event the Registration Statement or related prospectus contains any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made (in the case of the prospectus) not misleading, each of the undersigned agrees that (i) it will immediately discontinue offers and sales of the Shares under the Registration Statement (excluding, for the avoidance of doubt, sales conducted pursuant to Rule 144) until the undersigned receives copies of a supplemental or amended prospectus (which the Company agrees to promptly prepare) that corrects the misstatement(s) or omission(s) referred to above and receives notice that any post-effective

amendment has become effective or unless otherwise notified by the Company that it may resume such offers and sales, and (ii) it will maintain the confidentiality of any information included in such written notice delivered by the Company except (A) for disclosure to the undersigned's employees, agents and professional advisers who need to know such information and are obligated to keep it confidential, and (B) as otherwise required by law or subpoena. Notwithstanding anything to the contrary, after the Effectiveness Date, the Company shall cause its transfer agent to deliver unlegended Shares to a transferee of the undersigned in connection with any sale of Shares with respect to which the undersigned has entered into a contract for sale prior to the undersigned's receipt of the notice of a Suspension Event and which has not yet settled. If so directed by the Company, each of the undersigned will deliver to the Company or, in the undersigned's sole discretion destroy, all copies of the prospectus covering the Shares in its possession; provided, however, that this obligation to deliver or destroy all copies of the prospectus covering the Shares shall not apply (i) to the extent the undersigned is required to retain a copy of such prospectus (a) in order to comply with applicable legal, regulatory, self-regulatory or professional requirements or (b) in accordance with a bona fide pre-existing document retention policy or (ii) to copies stored electronically on archival servers as a result of automatic data back-up.

- d. Subscriber may deliver written notice (including via email) (an "Opt-Out Notice") to the Issuer requesting that Subscriber not receive notices from the Issuer otherwise required by this Section 8; provided, however, that Subscriber may later revoke any such Opt-Out Notice in writing. Following receipt of an Opt-Out Notice from Subscriber (unless and until subsequently revoked), (i) neither the SPAC nor the Company shall deliver any such notices to Subscriber and Subscriber shall no longer be entitled to the rights associated with any such notice and (ii) each time prior to Subscriber's intended use of an effective registration statement, Subscriber will notify the SPAC or the Company, as applicable, in writing at least two (2) business days in advance of such intended use, and if a notice of a Suspension Event was previously delivered (or would have been delivered but for the provisions of this Section 8.d) and the related suspension period remains in effect, the SPAC or the Company, as applicable will so notify Subscriber, within one (1) business day of Subscriber's notification, by delivering to Subscriber a copy of such notice of Suspension Event that would have been provided, and thereafter will provide Subscriber with the related notice of the conclusion of such Suspension Event immediately upon its availability, and Subscriber shall comply with any restrictions on using such Registration Statement during such Suspension Event.
- e. The Company shall, notwithstanding any termination of this Subscription Agreement, indemnify, defend and hold harmless the undersigned (to the extent a seller under the Registration Statement), the officers, directors, employees, investment advisers and agents of each of them, and each person who controls the undersigned (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) to the fullest extent permitted by applicable law, from and against any and all losses, claims, damages, liabilities, costs (including, without limitation, reasonable attorneys' fees) and expenses (collectively, "Losses"), as incurred, that arise out of or are based upon (i) any untrue or alleged untrue statement of a material fact contained in the Registration Statement, any prospectus included in the Registration Statement or any form of prospectus or in any amendment or supplement thereto or in any preliminary prospectus, or arising

out of or relating to any omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein (in the case of any prospectus or form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading, or (ii) any violation or alleged violation by the Company of the Securities Act, Exchange Act or any state securities law or any rule or regulation thereunder, in connection with the performance of its obligations under this Section 8, except to the extent, but only to the extent, that such untrue statements, alleged untrue statements, omissions or alleged omissions are based upon information regarding such Subscriber furnished in writing to the Company by such Subscriber expressly for use therein or such Subscriber has omitted a material fact from such information provided, however, that the Company shall not be liable for any Losses to the extent they arise out of or are based upon a violation which occurs (A) in reliance upon and in conformity with written information furnished by a Subscriber, (B) in connection with any failure of such person to deliver or cause to be delivered a prospectus made available by the Company in a timely manner (to the extent a prospectus was required to be delivered by Subscriber under applicable law), (C) as a result of offers or sales effected by or on behalf of any person by means of a free writing prospectus (as defined in Rule 405 of the Securities Act) that was not authorized in writing by the Company, or (D) in connection with any offers or sales effected by or on behalf of a Subscriber in violation of Section 8 hereof. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of an indemnified party and shall survive the transfer of the Shares by such Subscriber.

f. Each of the undersigned shall, severally and not jointly, indemnify and hold harmless the Company, its directors, officers, agents and employees, and each person who controls the Company (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act), to the fullest extent permitted by applicable law, from and against all Losses, as incurred, arising out of or are based upon any untrue or alleged untrue statement of a material fact contained in any Registration Statement, any prospectus included in the Registration Statement or any form of prospectus, or in any amendment or supplement thereto or in any preliminary prospectus, or arising out of or relating to any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of any prospectus, or any form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading to the extent, but only to the extent, that such untrue statements or omissions are based upon information regarding such Subscriber furnished in writing to the Company by such Subscriber expressly for use therein. In no event shall the liability of any of the undersigned be greater in amount than the dollar amount of the net proceeds received by the undersigned upon the sale of the Shares giving rise to such indemnification obligation. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of an indemnified party and shall survive the transfer of the Shares by such Subscriber.

g. Any person entitled to indemnification pursuant to this Section 8 shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not prejudiced the indemnifying party) and (ii) permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying

party shall not be subject to any liability for any settlement made by the indemnified party without its consent (which consent shall not be unreasonably withheld, conditioned or delayed). An indemnifying party who elects not to assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of legal counsel to any indemnified party a conflict of interest exists between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement for which indemnification could be sought hereunder which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

h. If the indemnification provided under this Section 8 from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any Losses, in lieu of indemnifying the indemnified party shall contribute to the amount paid or payable by the indemnified party as a result of such Losses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in this Section 8, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this Section 8 from any person who was not guilty of such fraudulent misrepresentation. Each indemnifying party's obligation to make a contribution pursuant to this Section 8.h. shall be individual, not joint and several, and in no event shall the liability of the Subscriber hereunder be greater in amount than the dollar amount of the net proceeds received by the Subscriber upon the sale of the Shares giving rise to such indemnification or contribution obligation.

9. <u>Termination</u>. This Subscription Agreement shall terminate and be void and of no further force or effect, and all rights and obligations of the parties hereunder shall terminate without any further liability on the part of any party in respect thereof, upon the earliest to occur of (a) such time as the Company notifies the undersigned or publicly discloses that the parties do not intend to consummate the Transaction, (b) such date and time as the Transaction Agreement is terminated in accordance with its terms without the Transaction being consummated, (c) upon the mutual written agreement of each of the parties hereto to terminate this Subscription Agreement, (d) if any of the conditions to the Subscription Closing set forth in <u>Section 3</u> of this Subscription Agreement are not satisfied or waived on or prior to the Subscription Closing and, as a result

thereof, the transactions contemplated by this Subscription Agreement are not consummated at the Subscription Closing, or (e) if the consummation of the Transaction shall not have occurred by the 270th day after the date this Subscription Agreement is accepted by the SPAC and the Company (and if such 270th day shall not be a business day, then the next following business day); <u>provided</u>, that nothing herein will relieve any party from liability for any willful breach hereof prior to the time of termination, and each party will be entitled to any remedies at law or in equity to recover losses, liabilities or damages arising from such breach. The Company shall promptly notify the undersigned of the termination of the Transaction Agreement after the termination of such agreement.

10. Trust Account Waiver. The undersigned acknowledges that the SPAC is a blank check company with the powers and privileges to effect a merger, asset acquisition, reorganization or similar business combination involving the SPAC and one or more businesses or assets. The undersigned further acknowledges that, as described in the SPAC's prospectus relating to its initial public offering dated February 25, 2021 (the "Prospectus") available at www.sec.gov, substantially all of the SPAC's assets consist of the cash proceeds of the SPAC's initial public offering and private placements of its securities, and substantially all of those proceeds have been deposited in a trust account (the "Trust Account") for the benefit of the SPAC, its public stockholders and the underwriters of the SPAC's initial public offering. For and in consideration of the SPAC entering into this Subscription Agreement, the receipt and sufficiency of which are hereby acknowledged, the undersigned hereby irrevocably waives any and all right, title and interest, or any claim of any kind it has or may have in the future, in or to any monies held in the Trust Account, and agrees not to seek recourse against the Trust Account, in each case, as a result of, or arising out of, this Subscription Agreement; provided, that nothing in this Section 10 shall be deemed to limit the undersigned's right, title, interest or claim to the Trust Account by virtue of the undersigned's record or beneficial ownership of shares of Class A Common Stock, if any.

11. Miscellaneous.

a. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in person, by e-mail or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses or e-mail addresses (or at such other address or email address for a party as shall be specified in a notice given in accordance with this <u>Section 11.a</u>):

If to the SPAC, to it at:

Kensington Capital Acquisition Corp. II 1400 Old Country Road, Suite 301 Westbury, NY 11590 Attention: Justin Mirro Email: justin@kensington-cap.com with a copy (which shall not constitute notice) to:

Hughes Hubbard & Reed LLP One Battery Park Plaza

New York, NY 10004

Attention: Charles A. Samuelson

Email: chuck.samuelson@hugheshubbard.com

If to the Company, to it at:

Wallbox B.V. Carrer del Foc, 68 Barcelona, Spain 08038

Attention: Enric Asuncion Sousa Email: enric@wallbox.com

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 811 Main Street Suite 3700 Houston, TX 77002 United States

Attention: Ryan Maierson Email: Ryan.Maierson@lw.com

Plaza de la Independencia 6

28001 Madrid

Spain

Attention: José Antonio Sánchez Email: Jose.Sanchez@lw.com

and

Hughes Hubbard & Reed LLP One Battery Park Plaza New York, NY 10004

Attention: Charles A. Samuelson

Email: chuck.samuelson@hugheshubbard.com

If to the undersigned, to the address or email address set forth for the undersigned on the signature page hereof.

b. All the agreements, representations and warranties made by each party hereto in this Subscription Agreement shall survive the Subscription Closing.

- c. If any term or other provision of this Subscription Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Subscription Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Subscription Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.
- d. This Subscription Agreement constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes all prior agreements and undertakings, both written and oral, among the parties, or any of them, with respect to the subject matter hereof. This Subscription Agreement shall not be assigned (whether pursuant to a merger, by operation of law or otherwise), by any party without the prior express written consent of the other parties hereto except that (i) this Subscription Agreement and any of the Subscriber's rights and obligations hereunder may be assigned to any fund or account managed by the same investment manager as the Subscriber or by an "affiliate" (as defined in Rule 12b-2 under the Exchange Act) of such investment manager without the prior consent of the SPAC or the Company and (ii) the Subscriber's rights under Section 8 are deemed to be assigned to an assignee or transferee of the Shares, provided, that such assignee will be deemed to have made each of the representations, warranties and covenants of the Subscriber set forth in Section 6 as of the date of such assignment and as of the Subscription Closing, and no such assignment by the Subscriber will relieve the Subscriber of its obligations under this Subscription Agreement and the Subscriber will remain secondarily liable under this Subscription Agreement for the obligations of the assignee hereunder.
- e. This Subscription Agreement shall be binding upon and inure solely to the benefit of each party hereto, and except as otherwise expressly set forth in subsection (q) of this Section 11, nothing in this Subscription Agreement, express or implied, is intended to or shall confer upon any other person any right, benefit or remedy of any nature whatsoever under or by reason of this Subscription Agreement.
- f. This Subscription Agreement shall be governed by, and construed in accordance with, the laws of the State of New York applicable to contracts executed in and to be performed in that State. All legal actions and proceedings arising out of or relating to this Subscription Agreement shall be heard and determined exclusively in any Supreme Court of the State of New York; provided, however, that if jurisdiction is not then available in the Supreme Court of the State of New York, then any such legal action may be brought in any federal court located in the State of New York or any other New York state court. The parties hereto hereby (a) irrevocably submit to the exclusive jurisdiction of the aforesaid courts for themselves and with respect to their respective properties for the purpose of any action arising out of or relating to this Subscription Agreement brought by any party hereto, and (b) agree not to commence any action relating thereto except in the courts described above in New York, other than actions in any court of competent jurisdiction to enforce any judgment, decree or award rendered by any such court in

New York as described herein. Each of the parties further agrees that notice as provided herein shall constitute sufficient service of process and the parties further waive any argument that such service is insufficient. Each of the parties hereby irrevocably and unconditionally waives, and agrees not to assert, by way of motion or as a defense, counterclaim or otherwise, in any action arising out of or relating to this Subscription Agreement or the transactions contemplated hereby, (a) any claim that it is not personally subject to the jurisdiction of the courts in New York as described herein for any reason, (b) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (c) that (i) the action in any such court is brought in an inconvenient forum, (ii) the venue of such action is improper or (iii) this Subscription Agreement, or the subject matter hereof, may not be enforced in or by such courts.

- g. EACH OF THE PARTIES HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS SUBSCRIPTION AGREEMENT. EACH OF THE PARTIES HERETO (I) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THAT FOREGOING WAIVER AND (II) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS SUBSCRIPTION AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY, AS APPLICABLE, BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 11.G.
- h. The descriptive headings contained in this Subscription Agreement are included for convenience of reference only and shall not affect in any way the meaning or interpretation of this Subscription Agreement.
- i. This Subscription Agreement may be executed and delivered (including by facsimile or portable document format (pdf) transmission) in counterparts, and by the different parties hereto in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.
- j. The parties hereto agree that irreparable damage would occur in the event any provision of this Subscription Agreement was not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof, in addition to any other remedy at law or in equity. Each of the parties hereby further waives (a) any defense in any action for specific performance that a remedy at law would be adequate and (b) any requirement under any law to post security or a bond as a prerequisite to obtaining equitable relief.
- k. Except as otherwise provided herein, all costs and expenses incurred in connection with this Subscription Agreement and the transactions contemplated hereby shall be paid by the party incurring such costs and expenses, whether or not the transactions contemplated hereby are consummated.

- 1. This Subscription Agreement may be amended in writing by the parties hereto at any time prior to the Subscription Closing. This Subscription Agreement may not be amended except by an instrument in writing signed by each of the parties hereto.
- m. At any time, the Company may (a) extend the time for the performance of any obligation or other act of the undersigned, (b) waive any inaccuracy in the representations and warranties of the undersigned contained herein or in any document delivered by the undersigned pursuant hereto and (c) waive compliance with any agreement of the undersigned or any condition to its own obligations contained herein. At any time, the undersigned may (a) extend the time for the performance of any obligation or other act of the Company, (b) waive any inaccuracy in the representations and warranties of the Company contained herein or in any document delivered by the Company pursuant hereto and (c) waive compliance with any agreement of the Company or any condition to its own obligations contained herein. Any such extension or waiver shall be valid if set forth in an instrument in writing signed by the party or parties to be bound thereby.
- n. The language used in this Subscription Agreement shall be deemed to be the language chosen by the parties to express their mutual intent and no rule of strict construction shall be applied against any party.
- o. Notwithstanding anything in this Subscription Agreement to the contrary, each party hereto acknowledges and agrees that the undersigned, without the prior written consent of the SPAC and the Company, will not publicly make reference to the SPAC, the Company or any of their affiliates, and the SPAC and the Company, without the prior written consent of the undersigned, will not, and will cause their representatives, including the Placement Agents and their respective representatives, not to, publicly make reference to the undersigned or its affiliates, in each case (i) in connection with the Transaction or this Subscription Agreement (provided, that the undersigned may disclose its entry into this Subscription Agreement and the Subscription Price) or (ii) in any press release, promotional materials, media, or similar circumstances, or filings with the Commission or any regulatory agency or trading agency except, in each case, as required by law or regulation or at the request of the Staff of the Commission or regulatory agency or under the regulations of NYSE, including, in the case of the SPAC and the Company (a) as required by the federal securities law in connection with the Registration Statement, and (b) the filing of this Subscription Agreement (or a form of this Subscription Agreement) with the Commission. Furthermore, the undersigned hereby consents to the disclosure with the Commission or any other securities authorities, of the undersigned's identity and beneficial ownership of the Shares and the nature of the undersigned's commitments, arrangements and understandings under and relating to this Subscription Agreement. The undersigned will promptly provide any information reasonably requested by the SPAC or the Company for any regulatory application or filing made or approval sought in connection with the Transactions (including filings with the Commission), in each case to the extent within the undersigned's possession and control or otherwise readily available to the undersigned.

- p. The Company may request from the undersigned such additional information as the Company may deem necessary to evaluate the eligibility of the undersigned to acquire the Shares, and the undersigned shall provide such information as may reasonably be requested, to the extent readily available and to the extent consistent with its internal policies and procedures.
- q. The undersigned acknowledges that the SPAC, the Company and the Placement Agents (pursuant to the ultimate sentence of this paragraph) will rely on the undersigned's acknowledgments, understandings, agreements, representations and warranties contained in this Subscription Agreement. The SPAC and the Company each acknowledges that the undersigned will rely on the SPAC's and the Company's acknowledgments, understandings, agreements, representations and warranties contained in this Subscription Agreement. Prior to the Subscription Closing, each of the parties agrees to promptly notify the other parties if (i) any of such party's acknowledgments, understandings, agreements, representations and warranties (other than any such representations and warranties that are qualified by materiality) made herein are no longer accurate in any material respect or (ii) any of such party's representations and warranties made herein that are qualified by materiality are no longer accurate in any respect. Each of the parties agrees that the purchase by the undersigned of Shares from the Company and the issuance of Shares by the Company to the undersigned will constitute a reaffirmation of the acknowledgments, understandings, agreements, representations and warranties herein (as modified by any such notice) by such party as of the time of such subscription and/or purchase. The undersigned further acknowledges and agrees that the Placement Agents are third-party beneficiaries of the representations and warranties of the undersigned contained in Sections 6.a, 6.b, 6.c, 6.f, 6.g, 6.h, 6.i., 6.o, 6.p, 6.q, 6.r, 6.s, 6.t and 6.u of this Subscription Agreement.
- r. Each of the SPAC, the Company and the undersigned is entitled to rely upon this Subscription Agreement and is irrevocably authorized to produce this Subscription Agreement or a copy hereof when required by law, regulatory authority, NYSE or Nasdaq, as applicable, to do so in any administrative or legal proceeding or official inquiry with respect to the matters covered hereby.
- 12. Separate Obligations. The obligations of the Subscriber under this Subscription Agreement are several and not joint with the obligations of any Other Subscriber or any other investor under the Other Subscription Agreements, and the Subscriber shall not be responsible in any way for the performance of the obligations of any Other Subscriber under the Other Subscription Agreements. The decision of the Subscriber to purchase the Shares pursuant to this Subscription Agreement has been made by the Subscriber independently of any Other Subscriber or any other investor and independently of any information, materials, statements or opinions as to the business, affairs, operations, assets, properties, liabilities, results of operations, condition (financial or otherwise) or prospects of the SPAC, the Company or Wallbox or any of their subsidiaries which may have been made or given by any Other Subscriber or other investor or by any agent or employee of any Other Subscriber or other investor, and neither the Subscriber nor

any of its agents or employees shall have any liability to any Other Subscriber or other investor (or any other person) relating to or arising from any such information, materials, statements or opinions. Nothing contained herein or in any Other Subscription Agreement, and no action taken by the Subscriber or any Other Subscriber or other investor pursuant hereto or thereto, shall be deemed to constitute the Subscriber, on the one hand, and any Other Subscriber or other investor, on the other hand, as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the Subscriber and any Other Subscriber or other investor are in any way acting in concert or as a group with respect to such obligations or the transactions contemplated by this Subscription Agreement and the Other Subscription Agreements; provided, that it is acknowledged that the Subscriber may be under common management with one or more Other Subscribers. Subscriber acknowledges that no Other Subscriber has acted as agent for the Subscriber in connection with making its investment hereunder and no Other Subscriber will be acting as agent of the Subscriber in connection with monitoring its investment in the Shares or enforcing its rights under this Subscription Agreement. Subscriber shall be entitled to independently protect and enforce its rights, including without limitation the rights arising out of this Subscription Agreement, and it shall not be necessary for any Other Subscriber or investor to be joined as an additional party in any proceeding for such purpose.

[SIGNATURE PAGES FOLLOW]

Name of Investor:	State/Country of Formation or Domicile:
By:	
Name in which shares are to be registered (if different):	Date:, 2021
Investor's EIN:	
Business Address-Street:	Mailing Address-Street (if different):
City, State, Zip:	City, State, Zip:
Attn:	Attn:
Telephone No.:	Telephone No.:
Facsimile No.:	Facsimile No.:
Number of Shares subscribed for:	
Aggregate Subscription Amount: \$	Price Per Share: \$10.00
V 4 0 1 1 2 4 1 1 1 4 6 CH	

IN WITNESS WHEREOF, the undersigned has executed or caused this Subscription Agreement to be executed by its duly authorized

representative as of the date set forth below.

You must pay the Subscription Amount by wire transfer of United States dollars in immediately available funds to the account specified by the Company and separately communicated to you.

IN WITNESS WHEREOF, each of Kensington Capital Acquisition Corp. II and Wallbox B.V. has accepted this Subscription Agreement as of

the date set forth below.

SCHEDULE A ELIGIBILITY REPRESENTATIONS OF THE INVESTOR

QUALIFIED INSTITUTIONAL BUYER STATUS (Please check the applicable subparagraphs):

В.

1.	☐ We are a	"qualified institutional buyer" (as defined in Rule 144A under the Securities Act).		
		ACCREDITED INVESTOR STATUS oplicable subparagraphs):		
1.	1. We are an "accredited investor" (within the meaning of Rule 501(a) under the Securities Act) for one or more of the following re(Please check the applicable subparagraphs):			
		Any bank, registered broker or dealer, insurance company, registered investment company, business development company, small business investment company, private business development company, or rural business investment company;		
		Any investment adviser registered pursuant to section 203 of the Investment Advisers Act or registered pursuant to the laws of a state;		
		Any investment adviser relying on the exemption from registering with the Commission under section 203(l) or (m) of the Investment Advisers Act;		
		Any plan established and maintained by a state, its political subdivisions, or any agency or instrumentality of a state or its political subdivisions, for the benefit of its employees, if such plan has total assets in excess of \$5,000,000;		
		Any employee benefit plan within the meaning of Title I of the Employee Retirement Income Security Act of 1974 ("ERISA"), if (i) the investment decision is made by a plan fiduciary, as defined in section 3(21) of ERISA, which is either a bank, a savings and loan association, an insurance company, or a registered investment adviser, (ii) the employee benefit plan has total assets in excess of \$5,000,000 or, (iii) such plan is a self-directed plan, with investment decisions made solely by persons that are "accredited investors";		
		Any (i) corporation, limited liability company or partnership, (ii) Massachusetts or similar business trust, or (iii) organization described in section 501(c)(3) of the Internal Revenue Code, in each case that was not formed for the specific purpose of acquiring the securities offered and that has total assets in excess of \$5,000,000;		

Schedule A

	Any trust, with total assets in excess of \$5,000,000, not formed for the specific purpose of acquiring the securities offered, whose subscription/purchase is directed by a sophisticated person as described in Section 230.506(b)(2)(ii) of Regulation D under the Securities Act;			
		other than an entity described in the categories of "accredited investors" above, not formed for the specific acquiring the securities offered, owning investments in excess of \$5,000,000;		
	Any "family office," as defined under the Investment Advisers Act that satisfies all of the following conditions: (i) with assets under management in excess of \$5,000,000, (ii) that is not formed for the specific purpose of acquiring the securities offered, and (iii) whose prospective investment is directed by a person who has such knowledge and experience in financial and business matters that such family office is capable of evaluating the merits and risks of the prospective investment;			
	Any "family client," as defined under the Investment Advisers Act, of a family office meeting the requirements in the previous paragraph and whose prospective investment in the issuer is directed by such family office pursuant to the previous paragraph and that is an institution; or			
☐ We are an entity in which all of the equity owners are accredited investors (Please check the applicable)		entity in which all of the equity owners are accredited investors (Please check the applicable subparagraphs):		
		Any director, executive officer, or general partner of the issuer of the securities being offered or sold, or any director, executive officer, or general partner of a general partner of that issuer;		
		Any natural person whose individual net worth, or joint net worth with that person's spouse or spousal equivalent, exceeds \$1,000,000. For purposes of calculating a natural person's net worth: (a) the person's primary residence shall not be included as an asset; (b) indebtedness that is secured by the person's primary residence, up to the estimated fair market value of the primary residence at the time of the sale of securities,		

Schedule A

			shall not be included as a liability (except that if the amount of such indebtedness outstanding at the time of sale of securities exceeds the amount outstanding 60 days before such time, other than as a result of the acquisition of the primary residence, the amount of such excess shall be included as a liability); and (c) indebtedness that is secured by the person's primary residence in excess of the estimated fair market value of the primary residence at the time of the sale of securities shall be included as a liability;
			Any natural person who had an individual income in excess of \$200,000 in each of the two most recent years or joint income with that person's spouse or spousal equivalent in excess of \$300,000 in each of those years and has a reasonable expectation of reaching the same income level in the current year;
			Any natural person holding in good standing one or more professional certifications or designations or credentials from an accredited educational institution that the Commission has designated as qualifying an individual for accredited investor status; or
			Any natural person who is a "knowledgeable employee," as defined in the Investment Company Act, of the issuer of the securities being offered or sold where the issuer would be an investment company, as defined in section 3 of such act, but for the exclusion provided by either section $3(c)(1)$ or section $3(c)(7)$ of such act.
AFFI	LIATE STATUS		
(Pleas	se check the applica	ble bo	x)
THE	INVESTOR:		
	is:		
	is not:		
an "a	ffiliate" (as defined	in Rul	e 144 under the Securities Act) of the Company or acting on behalf of an affiliate of the Company.
This page should be completed by the Investor.			
			Schedule A

C.

of

Name of Investor:	State/Country of Formation or Domicile:
By:	
Name:	
Title:	

Schedule A

SCHEDULE B ELIGIBILITY REPRESENTATIONS OF THE INVESTOR (Canadian Investors Only)

- 1. We hereby declare, represent and warrant that:
 - (a) we are subscribing for the Shares as principal for our own account, or are deemed to be subscribing for the Shares as principal for our own account in accordance with applicable Canadian securities laws, and not as agent for the benefit of another investor;
 - (b) we are residents in or subject to the laws of one of the provinces or territories of Canada;
 - (c) we are entitled under applicable securities laws to subscribe and accept the Shares without the benefit of a prospectus qualified under such securities laws and, without limiting the generality of the foregoing, are both:
 - a. an "accredited investor" as defined in section 1.1 of National Instrument 45-106 *Prospectus Exemptions* ("NI 45-106") or section 73.3(2) of the *Securities Act* (Ontario) by virtue of satisfying the indicated criterion in Section 11 below, and we are not a person created or used solely to purchase or hold securities as an "accredited investor" as described in paragraph (m) of the definition of "accredited investor" in section 1.1 of NI 45-106; and
 - b. a "permitted client" as defined in section 1.1 of National Instrument 31-103 Registration Requirements, Exemptions and Ongoing Registrant Obligations ("NI 31-103") by virtue of satisfying the indicated criterion in Section 12 below.
 - (d) we have received, reviewed and understood, this Subscription Agreement and certain disclosure materials relating to the placing of Shares in Canada and, are basing our investment decision solely on this Subscription and the materials provided by the Company and not on any other information concerning the Company or the offering of the Shares;
 - (e) the acquisition of Shares does not and will not contravene any applicable Canadian securities laws, rules or policies of the jurisdiction in which we are resident and does not trigger (i) any obligation to prepare and file a prospectus or similar document or (ii) any registration or other similar obligation on the part of any person;
 - (f) we will execute and deliver within the applicable time periods all documentation as may be required by applicable Canadian securities laws to permit the subscription for the Shares on the terms set forth herein and, if required by applicable Canadian securities laws, will execute, deliver and file or assist the Company in obtaining and filing such reports, undertakings and other documents relating to the subscription for the Shares as may be required by any applicable Canadian securities laws, securities regulator, stock exchange or other regulatory authority; and

- (g) neither we nor any party on whose behalf we are acting has been established, formed or incorporated solely to acquire or permit the subscription of Shares without a prospectus in reliance on an exemption from the prospectus requirements of applicable Canadian securities laws.
- 2. We are aware of the characteristics of the Shares, the risks relating to an investment therein and agree that we must bear the economic risk of its investment in the Shares. We understand that we will not be able to resell the Shares under applicable Canadian securities laws except in accordance with limited exemptions and compliance with other requirements of applicable law, and we (and not the Company) are responsible for compliance with applicable resale restrictions or hold periods and will comply with all relevant Canadian securities laws in connection with any resale of the Shares.
- 3. We hereby undertake to notify the Company immediately of any change to any declaration, representation, warranty or other information relating to us set forth herein which takes place prior to the closing of the subscription and acceptance of the Shares applied for hereby.
- 4. We understand and acknowledge that (i) the Company is not a reporting issuer in any province or territory in Canada and its securities are not listed on any stock exchange in Canada and there is currently no public market for the Shares in Canada; and (ii) the Company currently has no intention of becoming a reporting issuer in Canada and the Company is not obligated to file and has no present intention of filing a prospectus with any securities regulatory authority in Canada to qualify the resale of the Shares to the public, or listing the Company's securities on any stock exchange in Canada and thus the applicable restricted period or hold period may not commence and the Shares may be subject to an unlimited hold period or restricted period in Canada and in that case may only be sold pursuant to limited exemptions under applicable securities legislation.
- 5. We confirm we have reviewed applicable resale restrictions under relevant Canadian legislation and regulations.
- 6. It is acknowledged that we should consult our own legal and tax advisors with respect to the tax consequences of an investment in the Shares in our particular circumstances and with respect to the eligibility of the Shares for investment by us and resale restrictions under relevant Canadian legislation and regulations, and that we have not relied on the Company or on the contents of the disclosure materials provided by the Company, for any legal, tax or financial advice.

- 7. If we are a resident of Quebec, we acknowledge that it is our express wish that all documents evidencing or relating in any way to the sale of the Shares be drawn in the English language only. Si nous sommes résidents de la province de Québec, nous reconnaissons par les présentes que c'est notre volonté expresse que tous les documents faisant foi ou se rapportant de quelque manière à la vente des engagements soient rédigés en anglais seulement.
- 8. We understand and acknowledge that we are making the representations, warranties and agreements contained herein with the intent that they may be relied upon by the Company and the agents in determining our eligibility to subscribe for the Shares, including the availability of exemptions from the prospectus requirements of applicable Canadian securities laws in connection with the issuance of the Shares.
- 9. We consent to the collection, use and disclosure of certain personal information for the purposes of meeting legal, regulatory, self-regulatory, security and audit requirements (including any applicable tax, securities, money laundering or anti-terrorism legislation, rules or regulations) and as otherwise permitted or required by law, which disclosures may include disclosures to tax, securities or other regulatory or self-regulatory authorities in Canada and/or in foreign jurisdictions, if applicable, in connection with the regulatory oversight mandate of such authorities.
- 10. If we are an individual resident in Canada, we acknowledge that: (A) the Company or the agents may be required to provide personal information pertaining to us as required to be disclosed in Schedule I of Form 45-106F1 Report of Exempt Distribution ("Form 45-106F1") under NI 45-106 (including its name, email address, address, telephone number and the aggregate Subscription Price paid by the subscriber) ("personal information") to the securities regulatory authority or regulator in the local jurisdiction (the "Regulator"); (B) the personal information is being collected indirectly by the Regulator under the authority granted to it in securities legislation; and (C) the personal information is being collected for the purposes of the administration and enforcement of the securities legislation; and by subscribing/purchasing the securities, we shall be deemed to have authorized such indirect collection of personal information by the Regulator. Questions about the indirect collection of information should be directed to the Regulator in the local jurisdiction, using the contact information set out below:
 - (a) in Alberta, the Alberta Securities Commission, Suite 600, 250-5th Street SW, Calgary, Alberta T2P 0R4, Telephone: (403) 297-6454, toll free in Canada: 1-877-355-0585;
 - (b) in British Columbia, the British Columbia Securities Commission, P.O. Box 10142, Pacific Centre, 701 West Georgia Street, Vancouver, British Columbia V7Y 1L2, Inquiries: (604) 899-6581, toll free in Canada: 1-800-373-6393, Email: inquiries:@bcsc.bc.ca;
 - (c) in Manitoba, The Manitoba Securities Commission, 500–400 St. Mary Avenue, Winnipeg, Manitoba R3C 4K5, Telephone: (204) 945-2548, toll free in Manitoba 1-800-655-5244;

- (d) in New Brunswick, Financial and Consumer Services Commission (New Brunswick), 85 Charlotte Street, Suite 300, Saint John, New Brunswick E2L 2J2, Telephone: (506) 658-3060, toll free in Canada: 1-866-933-2222, Email: info@fcnb.ca;
- (e) in Newfoundland and Labrador, Government of Newfoundland and Labrador, Financial Services Regulation Division, P.O. Box 8700, Confederation Building, 2nd Floor, West Block, Prince Philip Drive, St. John's, Newfoundland and Labrador, A1B 4J6, Attention: Director of Securities, Telephone: (709) 729-4189,
- (f) in the Northwest Territories, the Government of the Northwest Territories, Office of the Superintendent of Securities, P.O. Box 1320, Yellowknife, Northwest Territories X1A 2L9, Attention: Deputy Superintendent, Legal & Enforcement, Telephone: (867) 920-8984;
- (g) in Nova Scotia, the Nova Scotia Securities Commission, Suite 400, 5251 Duke Street, Duke Tower, P.O. Box 458, Halifax, Nova Scotia B3J 2P8, Telephone: (902) 424-7768;
- (h) in Nunavut, Government of Nunavut, Department of Justice, Legal Registries Division, P.O. Box 1000, Station 570, 1st Floor, Brown Building, Iqaluit, Nunavut X0A 0H0, Telephone: (867) 975-6590;
- (i) in Ontario, the Inquiries Officer at the Ontario Securities Commission, 20 Queen Street West, 22nd Floor, Toronto, Ontario M5H 3S8, Telephone: (416) 593-8314, toll free in Canada: 1-877-785-1555, Email: exemptmarketfilings@osc.gov.on.ca;
- in Prince Edward Island, the Prince Edward Island Securities Office, 95 Rochford Street, 4th Floor Shaw Building, P.O. Box 2000, Charlottetown, Prince Edward Island C1A 7N8, Telephone: (902) 368-4569;
- (k) in Québec, the Autorité des marchés financiers, 800, Square Victoria, 22e étage, C.P. 246, Tour de la Bourse, Montréal, Québec H4Z 1G3, Telephone: (514) 395-0337 or 1-877-525-0337, Email:financementdessocietes@lautorite.qc.ca (For corporate finance issuers), fonds dinvestissement@lautorite.qc.ca (For investment fund issuers);
- in Saskatchewan, the Financial and Consumer Affairs Authority of Saskatchewan, Suite 601–1919 Saskatchewan Drive, Regina, Saskatchewan S4P 4H2, Telephone: (306) 787-5879; and
- (m) in Yukon, Government of Yukon, Department of Community Services, Law Centre, 3rd Floor, 2130 Second Avenue, Whitehorse, Yukon Y1A 5H6, Telephone: (867) 667-5314.

11.	We hereby represent, warrant, covenant and certify that we are, or any party on whose behalf we are acting is, an "accredited investor" as defined in NI 45-106 or section 73.3(1) of the <i>Securities Act</i> (Ontario) by virtue of satisfying the indicated criterion below:		
	Please	check the category that applies:	
		a Canadian financial institution or a Schedule III bank of the Bank Act (Canada),	
		the Business Development Bank of Canada incorporated under the Business Development Bank of Canada Act (Canada),	
		a subsidiary of any person or company referred to in paragraphs (a) or (b) if the person or company owns all of the voting securities of the subsidiary, except the voting securities required by law to be owned by directors of that subsidiary,	
		a person or company registered under the securities legislation of a province or territory of Canada as an adviser or dealer, except as otherwise prescribed by the regulations,	
		[omitted]	
	(e.1)	[omitted]	
		the Government of Canada, the government of a province or territory of Canada, or any Crown corporation, agency or wholly- owned entity of the Government of Canada or of the government of a province or territory of Canada,	
		a municipality, public board or commission in Canada and a metropolitan community, school board, the Comité de gestion de la taxe scolaire de l' île de Montréal or an intermunicipal management board in Québec,	
		any national, federal, state, provincial, territorial or municipal government of or in any foreign jurisdiction, or any agency of that government,	
	(i)	a pension fund that is regulated by either the Office of the Superintendent of Financial Institutions (Canada) or a pension commission or similar regulatory authority of a province or territory of Canada,	
		[omitted]	
	(j.1)	an individual who beneficially owns financial assets having an aggregate realizable value that before taxes, but net of any related liabilities, exceeds CAD\$5,000,000,	
		[omitted]	
		[omitted]	

a person, other than an individual or investment fund, that has net assets of at least \$5,000,000 as shown on its most recently prepared financial statements,
an investment fund that distributes or has distributed its securities only to a person that is or was an accredited investor at the time of the distribution, a person that acquires or acquired securities in the circumstances referred to in sections 2.10 of NI 45-106 [Minimum amount investment], or 2.19 of NI 45-106 [Additional investment in investment funds], or a person described in paragraph (i) or (ii) that acquires or acquired securities under section 2.18 of NI 45-106 [Investment fund reinvestment],
an investment fund that distributes or has distributed securities under a prospectus in a jurisdiction of Canada for which the regulator or, in Québec, the securities regulatory authority, has issued a receipt,
a trust company or trust corporation registered or authorized to carry on business under the <i>Trust and Loan Companies Act</i> (Canada) or under comparable legislation in a jurisdiction of Canada or a foreign jurisdiction, acting on behalf of a fully managed account managed by the trust company or trust corporation, as the case may be,
a person acting on behalf of a fully managed account managed by that person, if that person is registered or authorized to carry on business as an adviser or the equivalent under the securities legislation of a jurisdiction of Canada or a foreign jurisdiction,
a registered charity under the <i>Income Tax Act</i> (Canada) that, in regard to the trade, has obtained advice from an eligibility adviser or an adviser registered under the securities legislation of the jurisdiction of the registered charity to give advice on the securities being traded,
an entity organized in a foreign jurisdiction that is analogous to any of the entities referred to in paragraphs (a) through (d) or paragraph (i) in form and function,
a person in respect of which all of the owners of interests, direct, indirect or beneficial, except the voting securities required by law to be owned by directors, are persons that are accredited investors,
an investment fund that is advised by a person registered as an adviser or a person that is exempt from registration as an adviser,

. A "fully managed account" means an account of a client for which a person makes the investment decisions if that person has full discretion to trade in securities for the account without requiring the client's express consent to a transaction.

		a person that is recognized or designated by the Commission as an accredited investor,
		a trust established by an accredited investor for the benefit of the accredited investor's family members of which a majority of the trustees are accredited investors and all of the beneficiaries are the accredited investor's spouse, a former spouse of the accredited investor or a parent, grandparent, brother, sister, child or grandchild of that accredited investor, of that accredited investor's spouse or of that accredited investor's former spouse.
12.		ereby represent, warrant, covenant and certify that we are, or any party on whose behalf we are acting is, a "permitted client" by virtue of the on indicated below,
	Pleas	e check the category that applies:
	(a)	a Canadian financial institution or a Schedule III bank;
	(b)	the Business Development Bank of Canada incorporated under the Business Development Bank of Canada Act (Canada);
	(c)	a subsidiary of any person or company referred to in paragraph (a) or (b), if the person or company owns all of the voting securities of the subsidiary, except the voting securities required by law to be owned by directors of the subsidiary;
	(d)	a person or company registered under the securities legislation of a jurisdiction of Canada as an adviser, investment dealer, mutual fund dealer or exempt market dealer;
	(e)	a pension fund that is regulated by either the Office of the Superintendent of Financial Institutions or a pension commission or similar regulatory authority of a jurisdiction of Canada or a wholly-owned subsidiary of such a pension fund;
	(f)	an entity organized in a foreign jurisdiction that is analogous to any of the entities referred to in paragraphs (a) through (e);
	(g)	the Government of Canada or a jurisdiction of Canada, or any Crown corporation, agency or wholly-owned entity of the Government of Canada or a jurisdiction of Canada;
	(h)	any national, federal, state, provincial, territorial or municipal government of or in any foreign jurisdiction, or any agency of that government;
	(i)	a municipality, public board or commission in Canada and a metropolitan community, school board, the Comité de gestion de la taxe scolaire de l' île de Montréal or an intermunicipal management board in Quebec;
	(j)	a trust company or trust corporation registered or authorized to carry on business under the Trust and Loan Companies Act (Canada) or under comparable legislation in a jurisdiction of Canada or a foreign jurisdiction, acting on behalf of a managed account managed by the trust company or trust corporation, as the case may be;

(k)	a person or company acting on behalf of a managed account managed by person or company, if the person or company is registered or authorized to carry on business as an adviser or the equivalent under the securities legislation of a jurisdiction of Canada or a foreign jurisdiction;
(1)	an investment fund if one or both of the following apply:
	(i) the fund is managed by a person or company registered as an investment fund manager under the securities legislation of a jurisdiction of Canada;
	(ii) the fund is advised by a person or company authorized to act as an adviser under the securities legislation of a jurisdiction of Canada;
(m)	in respect of a dealer, a registered charity under the <i>Income Tax Act</i> (Canada) that obtains advice on the securities to be traded from an eligibility adviser or an adviser registered under the securities legislation of the jurisdiction of the registered charity;
(n)	in respect of an adviser, a registered charity under the Income Tax Act (Canada) that is advised by an eligibility adviser or an adviser registered under the securities legislation of the jurisdiction of the registered charity;
(o)	a registered charity under the Income Tax Act (Canada) that obtains advice on the securities to be traded from an eligibility adviser or an adviser registered under the securities legislation of the jurisdiction of the registered charity;
(p)	an individual who beneficially owns financial assets having an aggregate realizable value that, before taxes but net of any related liabilities, exceeds \$5 million;
(q)	a person or company that is entirely owned by an individual or individuals referred to in paragraph (o), who holds the beneficial ownership interest in the person or company directly or through a trust, the trustee of which is a trust company or trust corporation registered or authorized to carry on business under the <i>Trust and Loan Companies Act</i> (Canada) or under comparable legislation in a jurisdiction of Canada or a foreign jurisdiction;
(r)	a person or company, other than an individual or an investment fund, that has net assets of at least Cad\$25,000,000 as shown on its most recently prepared financial statements; or
(s)	a person or company that distributes securities of its own issue in Canada only to persons or companies referred to in paragraphs (a) through (r).

WALLBOX N.V. 2021 INCENTIVE AWARD PLAN

ARTICLE I. PURPOSE

The Plan's purpose is to enhance the Company's ability to attract, retain and motivate persons who make (or are expected to make) important contributions to the Company by providing these individuals with equity ownership opportunities. Capitalized terms used in the Plan are defined in Article XI.

ARTICLE II. ELIGIBILITY

Service Providers are eligible to be granted Awards under the Plan, subject to the limitations described herein.

ARTICLE III. ADMINISTRATION AND DELEGATION

- 3.1 <u>Administration</u>. The Plan is administered by the Administrator. The Administrator has authority to determine which Service Providers receive Awards, grant Awards and set Award terms and conditions, subject to the conditions and limitations in the Plan. The Administrator also has the authority to take all actions and make all determinations under the Plan, to interpret the Plan and Award Agreements and to adopt, amend and repeal Plan administrative rules, guidelines and practices as it deems advisable. The Administrator may correct defects and ambiguities, supply omissions and reconcile inconsistencies in the Plan or any Award as it deems necessary or appropriate to administer the Plan and any Awards. The Administrator's determinations under the Plan are in its sole discretion and will be final and binding on all persons having or claiming any interest in the Plan or any Award. The Administrator may institute and determine the terms and conditions of an Exchange Program.
- 3.2 <u>Appointment of Committees</u>. To the extent Applicable Laws permit, the Board may delegate any or all of its powers under the Plan to one or more Committees or officers of the Company or any of its Subsidiaries. The Board may abolish any Committee or re-vest in itself any previously delegated authority at any time.

ARTICLE IV. STOCK AVAILABLE FOR AWARDS

- 4.1 <u>Number of Shares</u>. Subject to adjustment under Article VII and the terms of this Article IV, Awards may be made under the Plan covering up to the Overall Share Limit. Shares issued under the Plan may consist of newly issued Shares, Shares purchased on the open market and/or treasury Shares.
- 4.2 <u>Share Recycling</u>. If all or any part of an Award expires, lapses or is terminated, exchanged for cash, surrendered to an Exchange Program, repurchased, canceled without having been fully exercised or forfeited, in any case, in a manner that results in the Company acquiring Shares at a price not greater than the price (as adjusted to reflect any Equity Restructuring) paid by the Participant for such Shares or not issuing any Shares covered by the Award, the unused Shares covered by the Award will, as applicable, become or again be available for Award grants under the Plan. Further, Shares delivered (either by actual

delivery or attestation) to the Company by a Participant to satisfy the applicable exercise or purchase price of an Award and/or to satisfy any applicable tax withholding obligation (including Shares retained by the Company from the Award being exercised or purchased and/or creating the tax obligation) will, as applicable, become or again be available for Award grants under the Plan. The payment of Dividend Equivalents in cash in conjunction with any outstanding Awards shall not count against the Overall Share Limit.

- 4.3 <u>Incentive Stock Option Limitations</u>. Notwithstanding anything to the contrary herein, no more than 17,040,419 Shares may be issued pursuant to the exercise of Incentive Stock Options.
- 4.4 Substitute Awards. In connection with an entity's merger or consolidation with the Company or the Company's acquisition of an entity's property or stock, the Administrator may grant Awards in substitution for any options or other stock or stock-based awards granted before such merger or consolidation by such entity or its affiliate in accordance with Applicable Laws. Substitute Awards may be granted on such terms as the Administrator deems appropriate, notwithstanding limitations on Awards in the Plan. Substitute Awards will not count against the Overall Share Limit (nor shall Shares subject to a Substitute Award be added to the Shares available for Awards under the Plan as provided above), except that Shares acquired by exercise of substitute Incentive Stock Options will count against the maximum number of Shares that may be issued pursuant to the exercise of Incentive Stock Options under the Plan. Additionally, in the event that a company acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines has shares available under a pre-existing plan approved by stockholders and not adopted in contemplation of such acquisition or combination, the shares available for grant pursuant to the terms of such pre-existing plan (as adjusted, to the extent appropriate, using the exchange ratio or other adjustment or valuation ratio or formula used in such acquisition or combination to determine the consideration payable to the holders of common stock of the entities party to such acquisition or combination) may be used for Awards under the Plan and shall not reduce the Shares authorized for grant under the Plan (and Shares subject to such Awards shall not be added to the Shares available for Awards under the Plan as provided above); provided that Awards using such available shares shall not be made after the date awards or grants could have been made under the terms of the pre-existing plan, absent the acquisition or combination, and shall only be made to individuals who were not Employ
- 4.5 Non-Employee Director Compensation. Notwithstanding any provision to the contrary in the Plan, the Administrator may establish compensation for non-employee Directors from time to time, subject to the limitations in the Plan and Dutch law, as applicable, with due observance of the Company's remuneration policy in existence at such time. The Administrator will from time to time determine the terms, conditions and amounts of all such non-employee Director compensation in its discretion and pursuant to the exercise of its business judgment, taking into account such factors, circumstances and considerations as it shall deem relevant from time to time, provided that the sum of any cash compensation, or other compensation, and the value (determined as of the grant date in accordance with Financial Accounting Standards Board Accounting Standards Codification Topic 718, or any successor thereto) of Awards granted to a non-employee Director as compensation for services as a non-employee Director during any fiscal year of the Company may not exceed \$1,000,000. The Administrator may make exceptions to these limits for individual non-employee Directors in extraordinary circumstances, as the Administrator may determine in its discretion, provided that the non-employee Director receiving such additional compensation may not participate in the decision to award such compensation or in other contemporaneous compensation decisions involving non-employee Directors.

ARTICLE V. STOCK OPTIONS AND STOCK APPRECIATION RIGHTS

- 5.1 General. The Administrator may grant Options or Stock Appreciation Rights to Service Providers subject to the limitations in the Plan, including any limitations in the Plan that apply to Incentive Stock Options. The Administrator will determine the number of Shares covered by each Option and Stock Appreciation Right, the exercise price of each Option and Stock Appreciation Right and the conditions and limitations applicable to the exercise of each Option and Stock Appreciation Right. A Stock Appreciation Right will entitle the Participant (or other person entitled to exercise the Stock Appreciation Right) to receive from the Company upon exercise of the exercisable portion of the Stock Appreciation Right an amount determined by multiplying the excess, if any, of the Fair Market Value of one Share on the date of exercise over the exercise price per Share of the Stock Appreciation Right by the number of Shares with respect to which the Stock Appreciation Right is exercised, subject to any limitations of the Plan or that the Administrator may impose and payable in cash, Shares valued at such Fair Market Value or a combination of the two as the Administrator may determine or provide in the Award Agreement.
- 5.2 Exercise Price. The Administrator will establish each Option's and Stock Appreciation Right's exercise price and specify the exercise price in the Award Agreement. Unless otherwise determined by the Administrator, the exercise price will not be less than 100% of the Fair Market Value on the grant date of the Option or Stock Appreciation Right.
- 5.3 <u>Duration</u>. Each Option or Stock Appreciation Right will be exercisable at such times and as specified in the Award Agreement, provided that, unless otherwise determined by the Administrator in accordance with Applicable Laws, the term of an Option or Stock Appreciation Right will not exceed ten years. Notwithstanding the foregoing, if the Participant, prior to the end of the term of an Option or Stock Appreciation Right, violates the non-competition, non-solicitation, confidentiality or other similar restrictive covenant provisions of any employment contract, confidentiality and nondisclosure agreement or other agreement between the Participant and the Company or any of its Subsidiaries, the right of the Participant and the Participant's transferees to exercise any Option or Stock Appreciation Right issued to the Participant shall terminate immediately upon such violation unless the Company otherwise determines.
- 5.4 Exercise. Options and Stock Appreciation Rights may be exercised by delivering to the Company a written notice of exercise, in a form the Administrator approves (which may be electronic), signed by the person authorized to exercise the Option or Stock Appreciation Right, together with, as applicable, payment in full (i) as specified in Section 5.5 for the number of Shares for which the Award is exercised and (ii) as specified in Section 9.5 for any applicable taxes. Option or Stock Appreciation Right cannot be exercised for a fraction of a Share.
- 5.5 <u>Payment Upon Exercise</u>. Subject to Section 10.8, any Company insider trading policy (including blackout periods) and Applicable Laws, the exercise price of an Option must be paid by:
- (a) cash, wire transfer of immediately available funds or by check payable to the order of the Company, provided that the Company may limit the use of one of the foregoing payment forms if one or more of the payment forms below is permitted;
- (b) if there is a public market for Shares at the time of exercise, unless the Company otherwise determines, (A) delivery (including telephonically to the extent permitted by the Company) of an irrevocable and unconditional undertaking by a broker acceptable to the Company to deliver promptly to the Company sufficient funds to pay the exercise price, or (B) the Participant's delivery to the Company of a copy of irrevocable and unconditional instructions to a broker acceptable to the Company to deliver promptly to the Company cash or a check sufficient to pay the exercise price; provided that such amount is paid to the Company at such time as may be required by the Administrator;

- (c) to the extent permitted by the Administrator, delivery (either by actual delivery or attestation) of Shares owned by the Participant valued at their Fair Market Value;
- (d) to the extent permitted by the Administrator, surrendering Shares then issuable upon the Option's exercise valued at their Fair Market Value on the exercise date;
- (e) to the extent permitted by the Administrator, delivery of any other property that the Administrator determines is good and valuable consideration; or
 - (f) to the extent permitted by the Company, any combination of the above payment forms approved by the Administrator.

ARTICLE VI. RESTRICTED STOCK; RESTRICTED STOCK UNITS

6.1 General. The Administrator may subject to limitations under Dutch law grant Restricted Stock or a security with similar terms, or the right to purchase Restricted Stock, to any Service Provider, subject to the Company's right to repurchase all or part of such shares at their issue price or other stated or formula price from the Participant (or to require forfeiture of such shares) if conditions the Administrator specifies in the Award Agreement are not satisfied before the end of the applicable restriction period or periods that the Administrator establishes for such Award. In addition, the Administrator may grant to Service Providers Restricted Stock Units, which may be subject to vesting and forfeiture conditions during the applicable restriction period or periods, as set forth in an Award Agreement. The Administrator will determine and set forth in the Award Agreement the terms and conditions for each Restricted Stock and Restricted Stock Unit Award, subject to the conditions and limitations contained in the Plan.

6.2 Restricted Stock.

- (a) <u>Dividends</u>. Participants holding shares of Restricted Stock will be entitled to all ordinary cash dividends paid with respect to such Shares, unless the Administrator provides otherwise in the Award Agreement. In addition, unless the Administrator provides otherwise, if any dividends or distributions are paid in Shares, or consist of a dividend or distribution to holders of Common Stock of property other than an ordinary cash dividend, the Shares or other property will be subject to the same restrictions on transferability and forfeitability as the shares of Restricted Stock with respect to which they were paid.
- (b) <u>Stock Certificates</u>. The Company may require that the Participant deposit in escrow with the Company (or its designee) any stock certificates issued in respect of shares of Restricted Stock, together with a stock power endorsed in blank.
- (c) <u>Section 83(b) Election</u>. If a Participant makes an election under Section 83(b) of the Code to be taxed with respect to the Restricted Stock as of the date of transfer of the Restricted Stock rather than as of the date or dates upon which such Participant would otherwise be taxable under Section 83(a) of the Code, such Participant shall be required to deliver a copy of such election to the Company promptly after filing such election with the Internal Revenue Service along with proof of the timely filing thereof.

6.3 Restricted Stock Units.

- (a) <u>Settlement</u>. The Administrator may provide that settlement of Restricted Stock Units will occur upon or as soon as reasonably practicable after the Restricted Stock Units vest or will instead be deferred, on a mandatory basis or at the Participant's election, in a manner intended to comply with Section 409A.
- (b) <u>Stockholder Rights</u>. A Participant will have no rights of a stockholder with respect to Shares subject to any Restricted Stock Unit unless and until the Shares are delivered in settlement of the Restricted Stock Unit.
- (c) <u>Dividend Equivalents</u>. If the Administrator provides, a grant of Restricted Stock Units may provide a Participant with the right to receive Dividend Equivalents. Dividend Equivalents may be paid currently or credited to an account for the Participant, settled in cash or Shares and subject to the same restrictions on transferability and forfeitability as the Restricted Stock Units with respect to which the Dividend Equivalents are granted and subject to other terms and conditions as set forth in the Award Agreement.

ARTICLE VII. OTHER STOCK OR CASH BASED AWARDS

7.1 Other Stock or Cash Based Awards may be granted to Participants, including Awards entitling Participants to receive Shares to be delivered in the future and including annual or other periodic or long-term cash bonus awards (whether based on specified Performance Criteria or otherwise), in each case subject to any conditions and limitations in the Plan. Such Other Stock or Cash Based Awards will also be available as a payment form in the settlement of other Awards, as standalone payments and as payment in lieu of compensation to which a Participant is otherwise entitled. Other Stock or Cash Based Awards may be paid in Shares, cash or other property, as the Administrator determines. Subject to the provisions of the Plan, the Administrator will determine the terms and conditions of each Other Stock or Cash Based Award, including any purchase price, performance goal (which may be based on the Performance Criteria), transfer restrictions, and vesting conditions, which will be set forth in the applicable Award Agreement.

ARTICLE VIII. ADJUSTMENTS FOR CHANGES IN COMMON STOCK AND CERTAIN OTHER EVENTS

- 8.1 Equity Restructuring(a). In connection with any Equity Restructuring, notwithstanding anything to the contrary in this Article VII, the Administrator will equitably adjust each outstanding Award as it deems appropriate to reflect the Equity Restructuring, which may include adjusting the number and type of securities subject to each outstanding Award and/or the Award's exercise price or grant price (if applicable), granting new Awards to Participants, and making a cash payment to Participants. The adjustments provided under this Section 8.1 will be nondiscretionary and final and binding on the affected Participant and the Company; provided that the Administrator will determine whether an adjustment is equitable.
- 8.2 Corporate Transactions. In the event of any dividend or other distribution (whether in the form of cash, Common Stock, other securities, or other property), reorganization, merger, consolidation, combination, amalgamation, repurchase, recapitalization, liquidation, dissolution, or sale, transfer, exchange or other disposition of all or substantially all of the assets of the Company, or sale or exchange of Common Stock or other securities of the Company, Change in Control, issuance of warrants or other rights to purchase Common Stock or other securities of the Company, other similar corporate transaction or event, other unusual or nonrecurring transaction or event affecting the Company or its financial

statements or any change in any Applicable Laws or accounting principles, the Administrator, on such terms and conditions as it deems appropriate, either by the terms of the Award or by action taken prior to the occurrence of such transaction or event (except that action to give effect to a change in Applicable Law or accounting principles may be made within a reasonable period of time after such change) and either automatically or upon the Participant's request, is hereby authorized to take any one or more of the following actions whenever the Administrator determines that such action is appropriate in order to (x) prevent dilution or enlargement of the benefits or potential benefits intended by the Company to be made available under the Plan or with respect to any Award granted or issued under the Plan, (y) to facilitate such transaction or event or (z) give effect to such changes in Applicable Laws or accounting principles:

- (a) To provide for the cancellation of any such Award in exchange for either an amount of cash or other property with a value equal to the amount that could have been obtained upon the exercise or settlement of the vested portion of such Award or realization of the Participant's rights under the vested portion of such Award, as applicable; provided that, if the amount that could have been obtained upon the exercise or settlement of the vested portion of such Award or realization of the Participant's rights, in any case, is equal to or less than zero, then the Award may be terminated without payment;
- (b) To provide that such Award shall vest and, to the extent applicable, be exercisable as to all shares covered thereby, notwithstanding anything to the contrary in the Plan or the provisions of such Award;
- (c) To provide that such Award be assumed by the successor or survivor corporation, or a parent or subsidiary thereof, or shall be substituted for by awards covering the stock of the successor or survivor corporation, or a parent or subsidiary thereof, with appropriate adjustments as to the number and kind of shares and/or applicable exercise or purchase price, in all cases, as determined by the Administrator;
- (d) To make adjustments in the number and type of shares of Common Stock (or other securities or property) subject to outstanding Awards and/or with respect to which Awards may be granted under the Plan (including, but not limited to, adjustments of the limitations in Article IV hereof on the maximum number and kind of shares which may be issued) and/or in the terms and conditions of (including the grant or exercise price), and the criteria included in, outstanding Awards;
 - (e) To replace such Award with other rights or property selected by the Administrator; and/or
 - (f) To provide that the Award will terminate and cannot vest, be exercised or become payable after the applicable event.
- 8.3 <u>Administrative Stand Still</u>. In the event of any pending stock dividend, stock split, combination or exchange of shares, merger, consolidation or other distribution (other than normal cash dividends) of Company assets to stockholders, or any other extraordinary transaction or change affecting the Shares or the share price of Common Stock, including any Equity Restructuring or any securities offering or other similar transaction, for administrative convenience, the Administrator may refuse to permit the exercise of any Award for up to sixty days before or after such transaction.
- 8.4 <u>General</u>. Except as expressly provided in the Plan or the Administrator's action under the Plan, no Participant will have any rights due to any subdivision or consolidation of Shares of any class, dividend payment, increase or decrease in the number of Shares of any class or dissolution, liquidation, merger, or consolidation of the Company or other corporation. Except as expressly provided with respect to an Equity Restructuring under Section 8.1 above or the Administrator's action under the Plan, no

issuance by the Company of Shares of any class, or securities convertible into Shares of any class, will affect, and no adjustment will be made regarding, the number of Shares subject to an Award or the Award's grant or exercise price. The existence of the Plan, any Award Agreements and the Awards granted hereunder will not affect or restrict in any way the Company's right or power to make or authorize (i) any adjustment, recapitalization, reorganization or other change in the Company's capital structure or its business, (ii) any merger, consolidation dissolution or liquidation of the Company or sale of Company assets or (iii) any sale or issuance of securities, including securities with rights superior to those of the Shares or securities convertible into or exchangeable for Shares. The Administrator may treat Participants and Awards (or portions thereof) differently under this Article VIII.

ARTICLE IX. GENERAL PROVISIONS APPLICABLE TO AWARDS

- 9.1 <u>Transferability</u>. Except as the Administrator may determine or provide in an Award Agreement or otherwise for Awards other than Incentive Stock Options, Awards may not be sold, assigned, transferred, pledged or otherwise encumbered, either voluntarily or by operation of law, except by will or the laws of descent and distribution, or, subject to the Administrator's consent, pursuant to a domestic relations order, and, during the life of the Participant, will be exercisable only by the Participant. References to a Participant, to the extent relevant in the context, will include references to a Participant's authorized transferee that the Administrator specifically approves.
- 9.2 <u>Documentation</u>. Each Award will be evidenced in an Award Agreement, which may be written or electronic, as the Administrator determines. Each Award may contain terms and conditions in addition to those set forth in the Plan.
- 9.3 <u>Discretion</u>. Except as the Plan otherwise provides, each Award may be made alone or in addition or in relation to any other Award. The terms of each Award to a Participant need not be identical, and the Administrator need not treat Participants or Awards (or portions thereof) uniformly.
- 9.4 <u>Termination of Service</u>; <u>Change in Status</u>. The Administrator will determine, in its sole discretion, the effect of all matters and questions relating to any Termination of Service, including, without limitation, whether a Termination of Service has occurred, whether a Termination of Service resulted from a discharge for Cause and all questions of whether a particular leave of absence constitutes a Termination of Service or any other change or purported change in a Participant's Service Provider status affects an Award and the extent to which, and the period during which, the Participant, the Participant's legal representative, conservator, guardian or Designated Beneficiary may exercise rights under the Award, if applicable.
- 9.5 Withholding. Each Participant must pay the Company, or make provision satisfactory to the Administrator for payment of, any taxes required by law to be withheld in connection with such Participant's Awards by the date of the event creating the tax liability. The Company may deduct an amount sufficient to satisfy such tax obligations based on the applicable statutory withholding rates (or such other rate as may be determined by the Company after considering any accounting consequences or costs) from any payment of any kind otherwise due to a Participant. Subject to Section 10.8 and any Company insider trading policy (including blackout periods), Participants may satisfy such tax obligations (i) in cash, by wire transfer of immediately available funds, by check made payable to the order of the Company, provided that the Company may limit the use of one of the foregoing payment forms if one or more of the payment forms below is permitted, (ii) to the extent permitted by the Administrator, in whole or in part by delivery of Shares, including Shares retained from the Award creating the tax obligation, valued at their Fair Market Value, (iii) if there is a public market for Shares at the time the tax obligations are satisfied, unless the Company otherwise determines, (A) delivery (including telephonically to the extent permitted by the

Company) of an irrevocable and unconditional undertaking by a broker acceptable to the Company to deliver promptly to the Company sufficient funds to satisfy the tax obligations, or (B) delivery by the Participant to the Company of a copy of irrevocable and unconditional instructions to a broker acceptable to the Company to deliver promptly to the Company cash or a check sufficient to satisfy the tax withholding; provided that such amount is paid to the Company at such time as may be required by the Administrator, or (iv) to the extent permitted by the Company, any combination of the foregoing payment forms approved by the Administrator. If any tax withholding obligation will be satisfied under clause (ii) of the immediately preceding sentence by the Company's retention of Shares from the Award creating the tax obligation and there is a public market for Shares at the time the tax obligation is satisfied, the Company may elect to instruct any brokerage firm determined acceptable to the Company for such purpose to sell on the applicable Participant's behalf some or all of the Shares retained and to remit the proceeds of the sale to the Company or its designee, and each Participant's acceptance of an Award under the Plan will constitute the Participant's authorization to the Company and instruction and authorization to such brokerage firm to complete the transactions described in this sentence.

- 9.6 Amendment of Award; Repricing. The Administrator may amend, modify or terminate any outstanding Award, including by substituting another Award of the same or a different type, changing the exercise or settlement date, and converting an Incentive Stock Option to a Non-Qualified Stock Option. The Participant's consent to such action will be required unless (i) the action, taking into account any related action, does not materially and adversely affect the Participant's rights under the Award, or (ii) the change is permitted under Article VIII or pursuant to Section 10.6.
- 9.7 Conditions on Delivery of Stock. The Company will not be obligated to deliver any Shares under the Plan or remove restrictions from Shares previously delivered under the Plan until (i) all Award conditions have been met or removed to the Company's satisfaction, (ii) as determined by the Company, all other legal matters regarding the issuance and delivery of such Shares have been satisfied, including any applicable securities laws and stock exchange or stock market rules and regulations, and (iii) the Participant has executed and delivered to the Company such representations or agreements as the Administrator deems necessary or appropriate to satisfy any Applicable Laws. The Company's inability to obtain authority from any regulatory body having jurisdiction, which the Administrator determines is necessary to the lawful issuance and sale of any securities, will relieve the Company of any liability for failing to issue or sell such Shares as to which such requisite authority has not been obtained.
- 9.8 <u>Acceleration</u>. The Administrator may at any time provide that any Award will become immediately vested and fully or partially exercisable, free of some or all restrictions or conditions, or otherwise fully or partially realizable.
- 9.9 Additional Terms of Incentive Stock Options. The Administrator may grant Incentive Stock Options only to employees of the Company, any of its present or future parent or subsidiary corporations, as defined in Sections 424(e) or (f) of the Code, respectively, and any other entities the employees of which are eligible to receive Incentive Stock Options under the Code. If an Incentive Stock Option is granted to a Greater Than 10% Stockholder, the exercise price will not be less than 110% of the Fair Market Value on the Option's grant date, and the term of the Option will not exceed five years. All Incentive Stock Options will be subject to and construed consistently with Section 422 of the Code. By accepting an Incentive Stock Option, the Participant agrees if requested by the Company to give prompt notice to the Company of dispositions or other transfers (other than in connection with a Change in Control) of Shares acquired under the Option made within (i) two years from the grant date of the Option or (ii) one year after the transfer of such Shares to the Participant, specifying the date of the disposition or other transfer and the amount the Participant realized, in cash, other property, assumption of indebtedness or other consideration, in such disposition or other transfer. Neither the Company nor the Administrator will be liable to a Participant, or any other party, if an Incentive Stock Option fails or ceases to qualify as an

"incentive stock option" under Section 422 of the Code. Any Incentive Stock Option or portion thereof that fails to qualify as an "incentive stock option" under Section 422 of the Code for any reason, including becoming exercisable with respect to Shares having a fair market value exceeding the \$100,000 limitation under Treasury Regulation Section 1.422-4, will be a Non-Qualified Stock Option.

ARTICLE X. MISCELLANEOUS

- 10.1 No Right to Employment or Other Status. No person will have any claim or right to be granted an Award, and the grant of an Award will not be construed as giving a Participant the right to continued employment or any other relationship with the Company. The Company expressly reserves the right at any time to dismiss or otherwise terminate its relationship with a Participant free from any liability or claim under the Plan or any Award, except as expressly provided in an Award Agreement.
- 10.2 No Rights as Stockholder; Certificates. Subject to the Award Agreement, no Participant or Designated Beneficiary will have any rights as a stockholder with respect to any Shares to be distributed under an Award until becoming the record holder of such Shares. Notwithstanding any other provision of the Plan, unless the Administrator otherwise determines or Applicable Laws require, the Company will not be required to deliver to any Participant certificates evidencing Shares issued in connection with any Award and instead such Shares may be recorded in the books of the Company (or, as applicable, its transfer agent or stock plan administrator). The Company may place legends on stock certificates issued under the Plan that the Administrator deems necessary or appropriate to comply with Applicable Laws.
- 10.3 Effective Date and Term of Plan. The Plan will become effective on the Effective Date and, unless earlier terminated by the Board, will remain in effect until the earlier of (i) the earliest date as of which all Awards granted under the Plan have been satisfied in full or terminated and no Shares approved for issuance under the Plan remain available to be granted under new Awards or (ii) the tenth anniversary of the Effective Date, but Awards previously granted may extend beyond that date in accordance with the Plan. If the Plan is not approved by the Company's stockholders, the Plan will not become effective and no Awards will be granted under the Plan.
- 10.4 Amendment of Plan. The Administrator may amend, suspend or terminate the Plan at any time; provided that no amendment, other than an increase to the Overall Share Limit, may materially and adversely affect any Award outstanding at the time of such amendment without the affected Participant's consent. No Awards may be granted under the Plan during any suspension period or after Plan termination. Awards outstanding at the time of any Plan suspension or termination will continue to be governed by the Plan and the Award Agreement, as in effect before such suspension or termination. The Board will obtain stockholder approval of any Plan amendment to the extent necessary to comply with Applicable Laws.
- 10.5 <u>Provisions for Foreign Participants</u>. The Administrator may modify Awards granted to Participants who are foreign nationals or employed outside the United States or establish subplans or procedures under the Plan to address differences in laws, rules, regulations or customs of such foreign jurisdictions with respect to tax, securities, currency, employee benefit or other matters.

10.6 Section 409A.

(a) General. The Company intends that all Awards be structured to comply with, or be exempt from, Section 409A, such that no adverse tax consequences, interest, or penalties under Section 409A apply. Notwithstanding anything in the Plan or any Award Agreement to the contrary, the Administrator may, without a Participant's consent, amend this Plan or Awards, adopt policies and procedures, or take any other actions (including amendments, policies, procedures and retroactive actions)

as are necessary or appropriate to preserve the intended tax treatment of Awards, including any such actions intended to (A) exempt this Plan or any Award from Section 409A, or (B) comply with Section 409A, including regulations, guidance, compliance programs and other interpretative authority that may be issued after an Award's grant date. The Company makes no representations or warranties as to an Award's tax treatment under Section 409A or otherwise. The Company will have no obligation under this Section 10.6 or otherwise to avoid the taxes, penalties or interest under Section 409A with respect to any Award and will have no liability to any Participant or any other person if any Award, compensation or other benefits under the Plan are determined to constitute noncompliant "nonqualified deferred compensation" subject to taxes, penalties or interest under Section 409A.

- (b) <u>Separation from Service</u>. If an Award constitutes "nonqualified deferred compensation" under Section 409A, any payment or settlement of such Award upon a termination of a Participant's Service Provider relationship will, to the extent necessary to avoid taxes under Section 409A, be made only upon the Participant's "separation from service" (within the meaning of Section 409A), whether such "separation from service" occurs upon or after the Termination of Service of a Participant. For purposes of this Plan or any Award Agreement relating to any such payments or benefits, references to a "termination," "termination of employment," Termination of Service or like terms means a "separation from service."
- (c) <u>Payments to Specified Employees</u>. Notwithstanding any contrary provision in the Plan or any Award Agreement, any payment(s) of "nonqualified deferred compensation" required to be made under an Award to a "specified employee" (as defined under Section 409A and as the Administrator determines) due to his or her "separation from service" will, to the extent necessary to avoid taxes under Section 409A(a)(2)(B)(i) of the Code, be delayed for the six-month period immediately following such "separation from service" (or, if earlier, until the specified employee's death) and will instead be paid (as set forth in the Award Agreement) on the day immediately following such six-month period or as soon as administratively practicable thereafter (without interest). Any payments of "nonqualified deferred compensation" under such Award payable more than six months following the Participant's "separation from service" will be paid at the time or times the payments are otherwise scheduled to be made.
- 10.7 <u>Limitations on Liability</u>. Notwithstanding any other provisions of the Plan, no individual acting as a director, officer, other employee or agent of the Company or any Subsidiary will be liable to any Participant, former Participant, spouse, beneficiary, or any other person for any claim, loss, liability, or expense incurred in connection with the Plan or any Award, and such individual will not be personally liable with respect to the Plan because of any contract or other instrument executed in his or her capacity as an Administrator, director, officer, other employee or agent of the Company or any Subsidiary. The Company will indemnify and hold harmless each director, officer, other employee and agent of the Company or any Subsidiary that has been or will be granted or delegated any duty or power relating to the Plan's administration or interpretation, against any cost or expense (including attorneys' fees) or liability (including any sum paid in settlement of a claim with the Administrator's approval) arising from any act or omission concerning this Plan unless arising from such person's own fraud or bad faith.
- 10.8 <u>Lock-Up Period</u>. The Company may, at the request of any underwriter representative or otherwise, in connection with registering the offering of any Company securities under the Securities Act, prohibit Participants from, directly or indirectly, selling or otherwise transferring any Shares or other Company securities during a period of up to one hundred eighty days following the effective date of a Company registration statement filed under the Securities Act, or such longer period as determined by the underwriter.

10.9 Data Privacy. As a condition for receiving any Award, each Participant explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of personal data as described in this section by and among the Company and its Subsidiaries and affiliates exclusively for implementing, administering and managing the Participant's participation in the Plan. The Company and its Subsidiaries and affiliates may hold certain personal information about a Participant, including the Participant's name, address and telephone number; birthdate; social security, insurance number or other identification number; salary; nationality; job title(s); any Shares held in the Company or its Subsidiaries and affiliates; and Award details, to implement, manage and administer the Plan and Awards (the "Data"). The Company and its Subsidiaries and affiliates may transfer the Data amongst themselves as necessary to implement, administer and manage a Participant's participation in the Plan, and the Company and its Subsidiaries and affiliates may transfer the Data to third parties assisting the Company with Plan implementation, administration and management. These recipients may be located in the Participant's country, or elsewhere, and the Participant's country may have different data privacy laws and protections than the recipients' country. By accepting an Award, each Participant authorizes such recipients to receive, possess, use, retain and transfer the Data, in electronic or other form, to implement, administer and manage the Participant's participation in the Plan, including any required Data transfer to a broker or other third party with whom the Company or the Participant may elect to deposit any Shares. The Data related to a Participant will be held only as long as necessary to implement, administer, and manage the Participant's participation in the Plan. A Participant may, at any time, view the Data that the Company holds regarding such Participant, request additional information about the storage and processing of the Data regarding such Participant, recommend any necessary corrections to the Data regarding the Participant or refuse or withdraw the consents in this Section 10.9 in writing, without cost, by contacting the local human resources representative. The Company may cancel Participant's ability to participate in the Plan and, in the Administrator's discretion, the Participant may forfeit any outstanding Awards if the Participant refuses or withdraws the consents in this Section 10.9. For more information on the consequences of refusing or withdrawing consent, Participants may contact their local human resources representative.

- 10.10 <u>Severability</u>. If any portion of the Plan or any action taken under it is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Plan, and the Plan will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- 10.11 <u>Governing Documents</u>. If any contradiction occurs between the Plan and any Award Agreement or other written agreement between a Participant and the Company (or any Subsidiary) that the Administrator has approved, the Plan will govern, unless it is expressly specified in such Award Agreement or other written document that a specific provision of the Plan will not apply.
- 10.12 Governing Law. The Plan and all Awards will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.
- 10.13 <u>Claw-back Provisions</u>. All Awards (including any proceeds, gains or other economic benefit the Participant actually or constructively receives upon receipt or exercise of any Award or the receipt or resale of any Shares underlying the Award) will be subject to any Company claw-back policy, including any claw-back policy adopted to comply with Applicable Laws (including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder) as set forth in such claw-back policy or the Award Agreement.
- 10.14 <u>Titles and Headings</u>. The titles and headings in the Plan are for convenience of reference only and, if any conflict, the Plan's text, rather than such titles or headings, will control.

- 10.15 <u>Conformity to Securities Laws</u>. Participant acknowledges that the Plan is intended to conform to the extent necessary with Applicable Laws. Notwithstanding anything herein to the contrary, the Plan and all Awards will be administered only in conformance with Applicable Laws. To the extent Applicable Laws permit, the Plan and all Award Agreements will be deemed amended as necessary to conform to Applicable Laws.
- 10.16 <u>Relationship to Other Benefits</u>. No payment under the Plan will be taken into account in determining any benefits under any pension, retirement, savings, profit sharing, group insurance, welfare or other benefit plan of the Company or any Subsidiary except as expressly provided in writing in such other plan or an agreement thereunder.
- 10.17 <u>Broker-Assisted Sales</u>. In the event of a broker-assisted sale of Shares in connection with the payment of amounts owed by a Participant under or with respect to the Plan or Awards, including amounts to be paid under the final sentence of Section 9.5: (a) any Shares to be sold through the broker-assisted sale will be sold on the day the payment first becomes due, or as soon thereafter as practicable; (b) such Shares may be sold as part of a block trade with other Participants in the Plan in which all participants receive an average price; (c) the applicable Participant will be responsible for all broker's fees and other costs of sale, and by accepting an Award, each Participant agrees to indemnify and hold the Company harmless from any losses, costs, damages, or expenses relating to any such sale; (d) to the extent the Company or its designee receives proceeds of such sale that exceed the amount owed, the Company will pay such excess in cash to the applicable Participant as soon as reasonably practicable; (e) the Company and its designees are under no obligation to arrange for such sale at any particular price; and (f) in the event the proceeds of such sale are insufficient to satisfy the Participant's applicable obligation, the Participant may be required to pay immediately upon demand to the Company or its designee an amount in cash sufficient to satisfy any remaining portion of the Participant's obligation.

ARTICLE XI. DEFINITIONS

As used in the Plan, the following words and phrases will have the following meanings:

- 11.1 "Administrator" means the Board or a Committee to the extent that the Board's powers or authority under the Plan have been delegated to such Committee.
- 11.2 "Applicable Laws" means the requirements relating to the administration of equity incentive plans under U.S. federal and state securities, tax and other applicable laws, rules and regulations, the applicable rules of any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws and rules of any foreign country or other jurisdiction where Awards are granted.
- 11.3 "Award" means, individually or collectively, a grant under the Plan of Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units or Other Stock or Cash Based Awards.
- 11.4 "Award Agreement" means a written agreement evidencing an Award, which may be electronic, that contains such terms and conditions as the Administrator determines, consistent with and subject to the terms and conditions of the Plan.
 - 11.5 "Board" means the Board of Directors of the Company.
- 11.6 "Cause" means (i) if a Participant is a party to a written employment, severance or consulting agreement with the Company or any of its Subsidiaries or an Award Agreement in which the term "cause" is defined (a "Relevant Agreement"), "Cause" as defined in the Relevant Agreement, and (ii) if no Relevant Agreement exists, (A) a material failure by the Participant to perform the Participant's

responsibilities or duties to the Company or any of its Subsidiaries under any agreement with the Company or any of its Subsidiaries or those other responsibilities or duties as reasonably requested from time to time by the Board, after written demand for performance has been given by the Board that identifies how the Participant has not performed his or her responsibilities or duties and such failure, if susceptible of cure, has not been cured for a period of thirty (30) days after the Participant receives notice from the Board; (B) the Participant's engagement in illegal conduct or gross misconduct that the Company in good faith believes has materially harmed or is reasonably likely to materially harm the standing and reputation of the Company or any of its Subsidiaries; (C) the Participant's commission or conviction of, or plea of guilty or nolo contendere to, a felony, a crime involving moral turpitude or any other act or omission that the Company in good faith believes has materially harmed or is reasonably likely to materially harm the standing and reputation of the Company or any of its Subsidiaries; (D) a material breach of the Participant's duty of loyalty to the Company or any of its Subsidiaries or the Participant's subsidiaries or the Participant's subsidiaries or any of its Subsidiaries written code of conduct and business ethics, in either case, that the Company in good faith believes has materially harmed or is reasonably likely to materially harm the standing and reputation of the Company or any of its Subsidiaries or the Participant's breach of any of the provision of this Agreement, or the Participant's material breach of any other material written agreement between the Participant and the Company or any of its Subsidiaries; (E) dishonesty that the Company in good faith believes has materially harmed or is reasonably likely to materially harm the Company or any of its Subsidiaries; or (G) excessive and unreasonable absences from your duties for any reason or as a result of the Participant's Di

11.7 "Change in Control" means and includes each of the following:

- (a) A transaction or series of transactions (other than an offering of Common Stock to the general public through a registration statement filed with the Securities and Exchange Commission or a transaction or series of transactions that meets the requirements of clauses (i) and (ii) of subsection (c) below) whereby any "person" or related "group" of "persons" (as such terms are used in Sections 13(d) and 14(d)(2) of the Exchange Act) (other than the Company, any of its Subsidiaries, an employee benefit plan maintained by the Company or any of its Subsidiaries or a "person" that, prior to such transaction, directly or indirectly controls, is controlled by, or is under common control with, the Company) directly or indirectly acquires beneficial ownership (within the meaning of Rule 13d-3 under the Exchange Act) of securities of the Company possessing more than 50 % of the total combined voting power of the Company's securities outstanding immediately after such acquisition; or
- (b) During any period of two consecutive years, individuals who, at the beginning of such period, constitute the Board together with any new Director(s) (other than a Director designated by a person who shall have entered into an agreement with the Company to effect a transaction described in subsections (a) or (c) whose election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds of the Directors then still in office) who either were Directors at the beginning of the two-year period or whose election or nomination for election was previously so approved, cease for any reason to constitute a majority thereof; or
- (c) The consummation by the Company (whether directly involving the Company or indirectly involving the Company through one or more intermediaries) of (x) a merger, consolidation, reorganization, or business combination or (y) a sale or other disposition of all or substantially all of the Company's assets in any single transaction or series of related transactions or (z) the acquisition of assets or stock of another entity, in each case other than a transaction:

(i) which results in the Company's voting securities outstanding immediately before the transaction continuing to represent (either by remaining outstanding or by being converted into voting securities of the Company or the person that, as a result of the transaction, controls, directly or indirectly, the Company or owns, directly or indirectly, all or substantially all of the Company's assets or otherwise succeeds to the business of the Company (the Company or such person, the "Successor Entity")) directly or indirectly, at least a majority of the combined voting power of the Successor Entity's outstanding voting securities immediately after the transaction, and

(ii) after which no person or group beneficially owns voting securities representing 50% or more of the combined voting power of the Successor Entity; provided, however, that no person or group shall be treated for purposes of this clause (ii) as beneficially owning 50% or more of the combined voting power of the Successor Entity solely as a result of the voting power held in the Company prior to the consummation of the transaction.

Notwithstanding the foregoing, if a Change in Control constitutes a payment event with respect to any Award (or portion of any Award) that provides for the deferral of compensation that is subject to Section 409A, to the extent required to avoid the imposition of additional taxes under Section 409A, the transaction or event described in subsection (a), (b) or (c) with respect to such Award (or portion thereof) shall only constitute a Change in Control for purposes of the payment timing of such Award if such transaction also constitutes a "change in control event," as defined in Treasury Regulation Section 1.409A-3(i)(5).

The Administrator shall have full and final authority, which shall be exercised in its discretion, to determine conclusively whether a Change in Control has occurred pursuant to the above definition, the date of the occurrence of such Change in Control and any incidental matters relating thereto; provided that any exercise of authority in conjunction with a determination of whether a Change in Control is a "change in control event" as defined in Treasury Regulation Section 1.409A-3(i)(5) shall be consistent with such regulation.

- 11.8 "Code" means the Internal Revenue Code of 1986, as amended, and the regulations issued thereunder.
- 11.9 "Committee" means one or more committees or subcommittees of the Board, which may include one or more Company directors or executive officers, to the extent Applicable Laws permit. To the extent required to comply with the provisions of Rule 16b-3, it is intended that each member of the Committee will be, at the time the Committee takes any action with respect to an Award that is subject to Rule 16b-3, a "non-employee director" within the meaning of Rule 16b-3; however, a Committee member's failure to qualify as a "non-employee director" within the meaning of Rule 16b-3 will not invalidate any Award granted by the Committee that is otherwise validly granted under the Plan.
 - 11.10 "Common Stock" means the Class A common stock of the Company.
- 11.11 "Company" means Wallbox N.V., a public company with limited liability incorporated under the laws of the Netherlands, registered with the Dutch trade register under number 83012559.
- 11.12 "Consultant" means any person, including any adviser, engaged by the Company or its parent or Subsidiary to render services to such entity if the consultant or adviser: (i) renders bona fide services to the Company; (ii) renders services not in connection with the offer or sale of securities in a capital-raising transaction and does not directly or indirectly promote or maintain a market for the Company's securities; and (iii) is a natural person.
- 11.13 "Designated Beneficiary" means the beneficiary or beneficiaries the Participant designates, in a manner the Administrator determines, to receive amounts due or exercise the Participant's rights if the Participant dies or becomes incapacitated. Without a Participant's effective designation, "Designated Beneficiary" will mean the Participant's estate.

- 11.14 "Director" means a Board member.
- 11.15 "Disability" means Employee's inability to perform the essential functions of his job, with or without accommodation, as a result of any mental or physical disability or incapacity for an extended period but not less than sixty (60) business days in any consecutive 6 month period, as determined in the sole discretion of Board.
- 11.16 "Dividend Equivalents" means a right granted to a Participant under the Plan to receive the equivalent value (in cash or Shares) of dividends paid on Shares.
- 11.17 "*Effective Date*" means the date on which the transactions contemplated by that certain Business Combination Agreement, by and among the Company, Kensington Capital Acquisition Corp. II, and Orion Merger Sub Corp., dated as of June 9, 2021, as may be amended from time to time, *provided* that the Board has adopted the Plan prior to or on such date, subject to approval of the Plan by the Company's stockholders.
 - 11.18 "Employee" means any employee of the Company or its Subsidiaries.
- 11.19 "Equity Restructuring" means a nonreciprocal transaction between the Company and its stockholders, such as a stock dividend, stock split, spin-off or recapitalization through a large, nonrecurring cash dividend, that affects the number or kind of Shares (or other Company securities) or the share price of Common Stock (or other Company securities) and causes a change in the per share value of the Common Stock underlying outstanding Awards.
 - 11.20 "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- 11.21 "Exchange Program" shall mean a program under which (i) outstanding Awards are surrendered or cancelled in exchange for Awards of the same type (which may have higher or lower exercise prices and different terms), Awards of a different type, and/or cash, (ii) Holders would have the opportunity to transfer any outstanding Awards to a financial institution or other person or entity selected by the Administrator, and/or (iii) the exercise price of an outstanding Award is reduced or increased. The Administrator will determine the terms and conditions of any Exchange Program in its sole discretion.
- 11.22 "Fair Market Value" means, as of any date, the value of Common Stock determined as follows: (i) if the Common Stock is listed on any established stock exchange, its Fair Market Value will be the closing sales price for such Common Stock as quoted on such exchange for such date, or if no sale occurred on such date, the last day preceding such date during which a sale occurred, as reported in The Wall Street Journal or another source the Administrator deems reliable; (ii) if the Common Stock is not traded on a stock exchange but is quoted on a national market or other quotation system, the closing sales price on such date, or if no sales occurred on such date, then on the last date preceding such date during which a sale occurred, as reported in The Wall Street Journal or another source the Administrator deems reliable; or (iii) in any case the Administrator may determine the Fair Market Value in its discretion.
- 11.23 "Greater Than 10% Stockholder" means an individual then owning (within the meaning of Section 424(d) of the Code) more than 10% of the total combined voting power of all classes of stock of the Company or its parent or subsidiary corporation, as defined in Section 424(e) and (f) of the Code, respectively.

- 11.24 "Incentive Stock Option" means an Option intended to qualify as an "incentive stock option" as defined in Section 422 of the Code.
- 11.25 "Non-Qualified Stock Option" means an Option not intended or not qualifying as an Incentive Stock Option.
- 11.26 "Option" means an option to purchase Shares.
- 11.27 "Other Stock or Cash Based Awards" means cash awards, awards of Shares, and other awards valued wholly or partially by referring to, or are otherwise based on, Shares or other property.
- 11.28 "Overall Share Limit" means the sum of (i) 17,040,419 Shares; and (ii) an annual increase on the first day of each calendar year beginning January 1, 2022 and ending on and including January 1, 2031, equal to the lesser of (A) 2.5% of the aggregate number of shares of Common Stock outstanding on the final day of the immediately preceding calendar year and (B) such smaller number of Shares as is determined by the Board.
 - 11.29 "Participant" means a Service Provider who has been granted an Award.
- 11.30 "Performance Criteria" mean the criteria (and adjustments) that the Administrator may select for an Award to establish performance goals for a performance period, which may include the following: net earnings or losses (either before or after one or more of interest, taxes, depreciation, amortization, and non-cash equity-based compensation expense); gross or net sales or revenue or sales or revenue growth; net income (either before or after taxes) or adjusted net income; profits (including but not limited to gross profits, net profits, profit growth, net operation profit or economic profit), profit return ratios or operating margin; budget or operating earnings (either before or after taxes or before or after allocation of corporate overhead and bonus); cash flow (including operating cash flow and free cash flow or cash flow return on capital); return on assets; return on capital or invested capital; cost of capital; return on stockholders' equity; total stockholder return; return on sales; costs, reductions in costs and cost control measures; expenses; working capital; earnings or loss per share; adjusted earnings or loss per share; price per share or dividends per share (or appreciation in or maintenance of such price or dividends); regulatory achievements or compliance; implementation, completion or attainment of objectives relating to research, development, regulatory, commercial, or strategic milestones or developments; market share; economic value or economic value added models; division, group or corporate financial goals; customer satisfaction/growth; customer service; employee satisfaction; recruitment and maintenance of personnel; human resources management; supervision of litigation and other legal matters; strategic partnerships and transactions; financial ratios (including those measuring liquidity, activity, profitability or leverage); debt levels or reductions; sales-related goals; financing and other capital raising transactions; cash on hand; acquisition activity; investment sourcing activity; marketing initiatives; and other measures of performance selected by the Board or Committee whether or not listed herein, any of which may be measured in absolute terms or as compared to any incremental increase or decrease. Such performance goals also may be based solely by reference to the Company's performance or the performance of a Subsidiary, division, business segment or business unit of the Company or a Subsidiary, or based upon performance relative to performance of other companies or upon comparisons of any of the indicators of performance relative to performance of other companies. The Committee may provide for exclusion of the impact of an event or occurrence which the Committee determines should appropriately be excluded, including (a) restructurings, discontinued operations, extraordinary items, and other unusual, infrequently occurring or non-recurring charges or events, (b) asset write-downs, (c) litigation or claim judgments or settlements, (d) acquisitions or divestitures, (e) reorganization or change in the corporate structure or capital structure of the Company, (f) an event either not directly related to the operations of the Company, Subsidiary, division, business segment or business unit or not within the reasonable control of management, (g) foreign exchange gains

and losses, (h) a change in the fiscal year of the Company, (i) the refinancing or repurchase of bank loans or debt securities, (j) unbudgeted capital expenditures, (k) the issuance or repurchase of equity securities and other changes in the number of outstanding shares, (l) conversion of some or all of convertible securities to Common Stock, (m) any business interruption event (n) the cumulative effects of tax or accounting changes in accordance with U.S. generally accepted accounting principles, or (o) the effect of changes in other laws or regulatory rules affecting reported results.

- 11.31 "Plan" means this 2021 Incentive Award Plan.
- 11.32 "Restricted Stock" means Shares awarded to a Participant under Article VI subject to certain vesting conditions and other restrictions.
- 11.33 "Restricted Stock Unit" means an unfunded, unsecured right to receive, on the applicable settlement date, one Share or an amount in cash or other consideration determined by the Administrator to be of equal value as of such settlement date, subject to certain vesting conditions and other restrictions.
 - 11.34 "Rule 16b-3" means Rule 16b-3 promulgated under the Exchange Act.
- 11.35 "Section 409A" means Section 409A of the Code and all regulations, guidance, compliance programs and other interpretative authority thereunder.
 - 11.36 "Securities Act" means the Securities Act of 1933, as amended.
 - 11.37 "Service Provider" means an Employee, Consultant or Director.
 - 11.38 "Shares" means shares of Common Stock.
 - 11.39 "Stock Appreciation Right" means a stock appreciation right granted under Article V.
- 11.40 "Subsidiary" means any entity (other than the Company), whether domestic or foreign, in an unbroken chain of entities beginning with the Company if each of the entities other than the last entity in the unbroken chain beneficially owns, at the time of the determination, securities or interests representing at least 50% of the total combined voting power of all classes of securities or interests in one of the other entities in such chain.
- 11.41 "Substitute Awards" shall mean Awards granted or Shares issued by the Company in assumption of, or in substitution or exchange for, awards previously granted, or the right or obligation to make future awards, in each case by a company acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines.
 - 11.42 "Termination of Service" means the date the Participant ceases to be a Service Provider.

* * * * *

WALLBOX N.V. 2021 EMPLOYEE STOCK PURCHASE PLAN

ARTICLE I. PURPOSE

The purpose of this Plan is to assist Eligible Employees of the Company and its Designated Subsidiaries in acquiring a stock ownership interest in the Company.

The Plan consists of two components: (i) the Section 423 Component and (ii) the Non-Section 423 Component. The Section 423 Component is intended to qualify as an "employee stock purchase plan" under Section 423 of the Code and shall be administered, interpreted and construed in a manner consistent with the requirements of Section 423 of the Code. The Non-Section 423 Component authorizes the grant of rights which need not qualify as rights granted pursuant to an "employee stock purchase plan" under Section 423 of the Code. Rights granted under the Non-Section 423 Component shall be granted pursuant to separate Offerings containing such sub-plans, appendices, rules or procedures as may be adopted by the Administrator and designed to achieve tax, securities laws or other objectives for Eligible Employees and Designated Subsidiaries but shall not be intended to qualify as an "employee stock purchase plan" under Section 423 of the Code. Except as otherwise determined by the Administrator or provided herein, the Non-Section 423 Component will operate and be administered in the same manner as the Section 423 Component. Offerings intended to be made under the Non-Section 423 Component will be designated as such by the Administrator at or prior to the time of such Offering.

For purposes of this Plan, the Administrator may designate separate Offerings under the Plan in which Eligible Employees will participate. The terms of these Offerings need not be identical, even if the dates of the applicable Offering Period(s) in each such Offering are identical, provided that the terms of participation are the same within each separate Offering under the Section 423 Component (as determined under Section 423 of the Code). Solely by way of example and without limiting the foregoing, the Company could, but shall not be required to, provide for simultaneous Offerings under the Section 423 Component and the Non-Section 423 Component of the Plan.

ARTICLE II. DEFINITIONS AND CONSTRUCTION

Wherever the following terms are used in the Plan they shall have the meanings specified below, unless the context clearly indicates otherwise.

- 2.1 "Administrator" means the entity that conducts the general administration of the Plan as provided in Article XI.
- 2.2 "Agent" means the brokerage firm, bank or other financial institution, entity or person(s), if any, engaged, retained, appointed or authorized to act as the agent of the Company or an Employee with regard to the Plan.
- 2.3 "Applicable Law" means the requirements relating to the administration of equity incentive plans under U.S. federal and state securities, tax and other applicable laws, rules and regulations, the applicable rules of any stock exchange or quotation system on which Shares are listed or quoted and the applicable laws and rules of any foreign country or other jurisdiction where rights under this Plan are granted.

- 2.4 "Articles of Association" means the Company's articles of association, as amended from time to time.
- 2.5 "Board" means the Board of Directors of the Company.
- 2.6 "Code" means the U.S. Internal Revenue Code of 1986, as amended, and the regulations issued thereunder.
- 2.7 "Common Stock" means common stock of the Company and such other securities of the Company that may be substituted therefore.
- 2.8 "Company" means Wallbox N.V., a public company with limited liability incorporated under the laws of the Netherlands, registered with the Dutch trade register under number 83012559, or any successor.
- 2.9 "Compensation" of an Eligible Employee means, unless otherwise determined by the Administrator, the gross base compensation or wages received by such Eligible Employee as compensation for services to the Company or any Designated Subsidiary, excluding overtime payments, sales commissions, incentive compensation, bonuses, expense reimbursements, income received in connection with any compensatory equity awards, fringe benefits and other special payments.
- 2.10 "Designated Subsidiary" means any Subsidiary designated by the Administrator in accordance with Section, such designation to specify whether such participation is in the Section 423 Component or Non-Section 423 Component. A Designated Subsidiary may participate in either the Section 423 Component or Non-Section 423 Component, but not both; provided that a Subsidiary that, for U.S. tax purposes, is disregarded from the Company or any Subsidiary that participates in the Section 423 Component shall automatically constitute a Designated Subsidiary that participates in the Section 423 Component.
- 2.11 "Effective Date" means the date on which the transactions contemplated by that certain Business Combination Agreement, by and among the Company, Kensington Capital Acquisition Corp. II, and Orion Merger Sub Corp., dated as of June 9, 2021, as may be amended from time to time, provided that the Board has adopted the Plan prior to or on such date, subject to approval of the Plan by the Company's stockholders.

2.12 "Eligible Employee" means:

- (a) an Employee who does not, immediately after any rights under this Plan are granted, own (directly or through attribution) stock possessing 5% or more of the total combined voting power or value of all classes of Shares and other securities of the Company, a Parent or a Subsidiary (as determined under Section 423(b)(3) of the Code). For purposes of the foregoing, the rules of Section 424(d) of the Code with regard to the attribution of stock ownership shall apply in determining the stock ownership of an individual, and stock that an Employee may purchase under outstanding options shall be treated as stock owned by the Employee.
- (b) Notwithstanding the foregoing, the Administrator may provide in an Offering Document that an Employee shall not be eligible to participate in an Offering Period under the Section 423 Component if: (i) such Employee is a highly compensated employee within the meaning of Section 423(b)(4)(D) of the Code; (ii) such Employee has not met a service requirement designated by the Administrator pursuant to Section 423(b)(4)(A) of the Code (which service requirement may not exceed two years); (iii) such Employee's customary employment is for twenty hours per week or less; (iv) such

Employee's customary employment is for less than five months in any calendar year; and/or (v) such Employee is a citizen or resident of a foreign jurisdiction and the grant of a right to purchase Shares under the Plan to such Employee would be prohibited under the laws of such foreign jurisdiction or the grant of a right to purchase Shares under the Plan to such Employee in compliance with the laws of such foreign jurisdiction would cause the Plan to violate the requirements of Section 423 of the Code, as determined by the Administrator in its sole discretion; provided, further, that any exclusion in clauses (i), (ii), (iii), (iv) or (v) shall be applied in an identical manner under each Offering Period to all Employees, in accordance with Treasury Regulation Section 1.423-2(e).

- (c) Further notwithstanding the foregoing, with respect to the Non-Section 423 Component, the first sentence in this definition shall apply in determining who is an "Eligible Employee," except (i) the Administrator may limit eligibility further within the Company or a Designated Subsidiary so as to only designate some Employees of the Company or a Designated Subsidiary as Eligible Employees, and (ii) to the extent the restrictions in the first sentence in this definition are not consistent with applicable local laws, the applicable local laws shall control.
- 2.13 "Employee" means any individual who renders services to the Company or any Designated Subsidiary in the status of an employee, and, with respect to the Section 423 Component, a person who is an employee within the meaning of Section 3401(c) of the Code. For purposes of an individual's participation in, or other rights under the Plan, all determinations by the Company shall be final, binding and conclusive, notwithstanding that any court of law or governmental agency subsequently makes a contrary determination. For purposes of the Plan, the employment relationship shall be treated as continuing intact while the individual is on sick leave or other leave of absence approved by the Company or Designated Subsidiary and meeting the requirements of Treasury Regulation Section 1.421-1(h)(2). Where the period of leave exceeds three (3) months and the individual's right to reemployment is not guaranteed either by statute or by contract, the employment relationship shall be deemed to have terminated on the first day immediately following such three (3)-month period.
 - 2.14 "Enrollment Date" means the first Trading Day of each Offering Period.
- 2.15 "Fair Market Value" means, as of any date, the value of Shares determined as follows: (i) if the Shares are listed on any established stock exchange, its Fair Market Value will be the closing sales price for such Shares as quoted on such exchange for such date, or if no sale occurred on such date, the last day preceding such date during which a sale occurred, as reported in The Wall Street Journal or another source the Administrator deems reliable; (ii) if the Shares are not traded on a stock exchange but are quoted on a national market or other quotation system, the closing sales price on such date, or if no sales occurred on such date, then on the last date preceding such date during which a sale occurred, as reported in The Wall Street Journal or another source the Administrator deems reliable; or (iii) without an established market for the Shares, the Administrator will determine the Fair Market Value in its discretion.
- 2.16 "Non-Section 423 Component" means those Offerings under the Plan, together with the sub-plans, appendices, rules or procedures, if any, adopted by the Administrator as a part of this Plan, in each case, pursuant to which rights to purchase Shares during an Offering Period may be granted to Eligible Employees that need not satisfy the requirements for rights to purchase Shares granted pursuant to an "employee stock purchase plan" that are set forth under Section 423 of the Code.
- 2.17 "Offering" means an offer under the Plan of a right to purchase Shares that may be exercised during an Offering Period as further described in Article IV hereof. Unless otherwise specified by the Administrator, each Offering to the Eligible Employees of the Company or a Designated Subsidiary shall be deemed a separate Offering, even if the dates and other terms of the applicable Offering Periods of each such Offering are identical, and the provisions of the Plan will separately apply to each Offering. To the extent permitted by Treas. Reg. § 1.423-2(a)(1), the terms of each separate Offering under the Section 423 Component need not be identical, provided that the terms of the Section 423 Component and an Offering thereunder together satisfy Treas. Reg. § 1.423-2(a)(2) and (a)(3).

- 2.18 "Offering Document" has the meaning given to such term in Section.
- 2.19 "Offering Period" has the meaning given to such term in Section .
- 2.20 "Parent" means any corporation, other than the Company, in an unbroken chain of corporations ending with the Company if, at the time of the determination, each of the corporations other than the Company owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain.
- 2.21 "Participant" means any Eligible Employee who has executed a subscription agreement and been granted rights to purchase Shares pursuant to the Plan.
- 2.22 "*Payday*" means the regular and recurring established day for payment of Compensation to an Employee of the Company or any Designated Subsidiary.
- 2.23 "Plan" means this 2021 Employee Stock Purchase Plan, including both the Section 423 Component and Non-Section 423 Component and any other sub-plans or appendices hereto, as amended from time to time.
- 2.24 "Purchase Date" means the last Trading Day of each Purchase Period or such other date as determined by the Administrator and set forth in the Offering Document.
- 2.25 "*Purchase Period*" shall refer to one or more periods within an Offering Period, as designated in the applicable Offering Document; provided, however, that, in the event no purchase period is designated by the Administrator in the applicable Offering Document, the purchase period for each Offering Period covered by such Offering Document shall be the same as the applicable Offering Period.
- 2.26 "Purchase Price" means the purchase price designated by the Administrator in the applicable Offering Document (which purchase price, for purposes of the Section 423 Component, shall not be less than 85% of the Fair Market Value of a Share on the Enrollment Date or on the Purchase Date, whichever is lower); provided, however, that, in the event no purchase price is designated by the Administrator in the applicable Offering Document, the purchase price for the Offering Periods covered by such Offering Document shall be 85% of the Fair Market Value of a Share on the Enrollment Date or on the Purchase Date, whichever is lower; provided, further, that the Purchase Price may be adjusted by the Administrator pursuant to Article VIII and shall not be less than the par value of a Share.
- 2.27 "Section 423 Component" means those Offerings under the Plan, together with the sub-plans, appendices, rules or procedures, if any, adopted by the Administrator as a part of this Plan, in each case, pursuant to which rights to purchase Shares during an Offering Period may be granted to Eligible Employees that are intended to satisfy the requirements for rights to purchase Shares granted pursuant to an "employee stock purchase plan" that are set forth under Section 423 of the Code.
 - 2.28 "Securities Act" means the U.S. Securities Act of 1933, as amended.
 - 2.29 "Share" means a share of Common Stock.

- 2.30 "Subsidiary" means any corporation, other than the Company, in an unbroken chain of corporations beginning with the Company if, at the time of the determination, each of the corporations other than the last corporation in an unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain; provided, however, that a limited liability company or partnership may be treated as a Subsidiary to the extent either (a) such entity is treated as a disregarded entity under Treasury Regulation Section 301.7701-3(a) by reason of the Company or any other Subsidiary that is a corporation being the sole owner of such entity, or (b) such entity elects to be classified as a corporation under Treasury Regulation Section 301.7701-3(a) and such entity would otherwise qualify as a Subsidiary. In addition, with respect to the Non-Section 423 Component, Subsidiary shall include any corporate or non-corporate entity in which the Company has a direct or indirect equity interest or significant business relationship.
 - 2.31 "Trading Day" means a day on which national stock exchanges in the United States are open for trading.
 - 2.32 "Treas. Reg." means U.S. Department of the Treasury regulations.

ARTICLE III. SHARES SUBJECT TO THE PLAN

- 3.1 Number of Shares. Subject to Article VIII, the aggregate number of Shares that may be issued pursuant to rights granted under the Plan shall be 8,550,209 Shares. In addition to the foregoing, subject to Article VIII, on the first day of each calendar year beginning on January 1, 2022 and ending on and including January 1, 2031, the number of Shares available for issuance under the Plan shall be increased by that number of Shares equal to the lesser of (a) 1% of the aggregate number of shares of Common Stock of the Company outstanding on the final day of the immediately preceding calendar year and (b) such smaller number of Shares as determined by the Board subject to the amount of the Company's authorized share capital under the Articles of Association. If any right granted under the Plan shall for any reason terminate without having been exercised, the Shares not purchased under such right shall again become available for issuance under the Plan. Notwithstanding anything in this Section to the contrary, the number of Shares that may be issued or transferred pursuant to the rights granted under the Section 423 Component of the Plan shall not exceed an aggregate of 25,000 Shares, subject to Article VIII.
- 3.2 <u>Shares Distributed</u>. Any Shares distributed pursuant to the Plan may consist, of newly issued Shares, treasury Shares and/or Shares purchased on the open market.

ARTICLE IV. OFFERING PERIODS; OFFERING DOCUMENTS; PURCHASE DATES

- 4.1 Offering Periods. The Administrator may from time to time grant or provide for the grant of rights to purchase Shares under the Plan to Eligible Employees during one or more periods (each, an "Offering Period") selected by the Administrator. The terms and conditions applicable to each Offering Period shall be set forth in an "Offering Document" adopted by the Administrator, which Offering Document shall be in such form and shall contain such terms and conditions as the Administrator shall deem appropriate and shall be incorporated by reference into and made part of the Plan and shall be attached hereto as part of the Plan. The provisions of separate Offerings or Offering Periods under the Plan need not be identical.
- 4.2 <u>Offering Documents</u>. Each Offering Document with respect to an Offering Period shall specify (through incorporation of the provisions of this Plan by reference or otherwise):

- (a) the length of the Offering Period, which period shall not exceed twenty-seven months;
- (b) the maximum number of Shares that may be purchased by any Eligible Employee during such Offering Period, which, in the absence of a contrary designation by the Administrator, shall be 20,000 Shares; and
 - (c) such other provisions as the Administrator determines are appropriate, subject to the Plan.

ARTICLE V. ELIGIBILITY AND PARTICIPATION

5.1 <u>Eligibility</u>. Any Eligible Employee who shall be employed by the Company or a Designated Subsidiary on a given Enrollment Date for an Offering Period shall be eligible to participate in the Plan during such Offering Period, subject to the requirements of this Article V and, for the Section 423 Component, the limitations imposed by Section 423(b) of the Code.

5.2 Enrollment in Plan.

- (a) Except as otherwise set forth in an Offering Document or determined by the Administrator, an Eligible Employee may become a Participant in the Plan for an Offering Period by delivering a subscription agreement to the Company by such time prior to the Enrollment Date for such Offering Period (or such other date specified in the Offering Document) designated by the Administrator and in such form as the Company provides.
- (b) Each subscription agreement shall designate either (i) a whole percentage of such Eligible Employee's Compensation (ii) or a fixed dollar amount, in either case, to be withheld by the Company or the Designated Subsidiary employing such Eligible Employee on each Payday during the Offering Period as payroll deductions under the Plan. In either event, the designated percentage or fixed dollar amount may not be less than 1% and may not be more than the maximum percentage specified by the Administrator in the applicable Offering Document (which percentage shall be 20% in the absence of any such designation) as payroll deductions. The payroll deductions made for each Participant shall be credited to an account for such Participant under the Plan and shall be deposited with the general funds of the Company.
- (c) A Participant may increase or decrease the percentage of Compensation or the fixed dollar amount designated in his or her subscription agreement, subject to the limits of this Section, or may suspend his or her payroll deductions, at any time during an Offering Period; provided, however, that the Administrator may limit the number of changes a Participant may make to his or her payroll deduction elections during each Offering Period in the applicable Offering Document (and in the absence of any specific designation by the Administrator, a Participant shall be allowed to decrease (but not increase) his or her payroll deduction elections one time during each Offering Period). Any such change or suspension of payroll deductions shall be effective with the first full payroll period following ten business days after the Company's receipt of the new subscription agreement (or such shorter or longer period as may be specified by the Administrator in the applicable Offering Document). In the event a Participant suspends his or her payroll deductions, such Participant's cumulative payroll deductions prior to the suspension shall remain in his or her account and shall be applied to the purchase of Shares on the next occurring Purchase Date and shall not be paid to such Participant unless he or she withdraws from participation in the Plan pursuant to Article VII.

- (d) Except as otherwise set forth in an Offering Document or determined by the Administrator, a Participant may participate in the Plan only by means of payroll deduction and may not make contributions by lump sum payment for any Offering Period.
- 5.3 Payroll Deductions. Except as otherwise provided in the applicable Offering Document, payroll deductions for a Participant shall commence on the first Payday following the Enrollment Date and shall end on the last Payday in the Offering Period to which the Participant's authorization is applicable, unless sooner terminated by the Participant as provided in Article VII or suspended by the Participant or the Administrator as provided in Section and Section, respectively. Notwithstanding any other provisions of the Plan to the contrary, in non-U.S. jurisdictions where participation in the Plan through payroll deductions is prohibited, the Administrator may provide that an Eligible Employee may elect to participate through contributions to the Participant's account under the Plan in a form acceptable to the Administrator in lieu of or in addition to payroll deductions; provided, however, that, for any Offering under the Section 423 Component, the Administrator shall take into consideration any limitations under Section 423 of the Code when applying an alternative method of contribution.
- 5.4 Effect of Enrollment. A Participant's completion of a subscription agreement will enroll such Participant in the Plan for each subsequent Offering Period on the terms contained therein until the Participant either submits a new subscription agreement, withdraws from participation under the Plan as provided in Article VII or otherwise becomes ineligible to participate in the Plan.
- 5.5 <u>Limitation on Purchase of Shares</u>. An Eligible Employee may be granted rights under the Section 423 Component only if such rights, together with any other rights granted to such Eligible Employee under "employee stock purchase plans" of the Company, any Parent or any Subsidiary, as specified by Section 423(b)(8) of the Code, do not permit such employee's rights to purchase stock of the Company or any Parent or Subsidiary to accrue at a rate that exceeds \$25,000 of the fair market value of such stock (determined as of the first day of the Offering Period during which such rights are granted) for each calendar year in which such rights are outstanding at any time. This limitation shall be applied in accordance with Section 423(b)(8) of the Code.
- 5.6 <u>Suspension of Payroll Deductions</u>. Notwithstanding the foregoing, to the extent necessary to comply with Section 423(b)(8) of the Code and Section (with respect to the Section 423 Component) or the other limitations set forth in this Plan, a Participant's payroll deductions may be suspended by the Administrator at any time during an Offering Period. The balance of the amount credited to the account of each Participant that has not been applied to the purchase of Shares by reason of Section 423(b)(8) of the Code, Section or the other limitations set forth in this Plan shall be paid to such Participant in one lump sum in cash as soon as reasonably practicable after the Purchase Date.
- 5.7 Foreign Employees. In order to facilitate participation in the Plan, the Administrator may provide for such special terms applicable to Participants who are citizens or residents a foreign jurisdiction, or who are employed by a Designated Subsidiary outside of the United States, as the Administrator may consider necessary or appropriate to accommodate differences in local law, tax policy or custom. Except as permitted by Section 423 of the Code, with respect to the Section 423 Component, such special terms may not be more favorable than the terms of rights granted under the Section 423 Component to Eligible Employees who are residents of the United States. Such special terms may be set forth in an addendum to the Plan in the form of an appendix or sub-plan (which appendix or sub-plan may be designed to govern Offerings under the Section 423 Component or the Non-Section 423 Component, as determined by the Administrator). To the extent that the terms and conditions set forth in an appendix or sub-plan conflict with any provisions of the Plan, the provisions of the appendix or sub-plan shall govern. The adoption of any such appendix or sub-plan shall be pursuant to Section 11.2(g). Without limiting the foregoing, the Administrator is specifically authorized to adopt rules and procedures, with respect to Participants who are

foreign nationals or employed in non-U.S. jurisdictions, regarding the exclusion of particular Subsidiaries from participation in the Plan, eligibility to participate, the definition of Compensation, handling of payroll deductions or other contributions by Participants, payment of interest, conversion of local currency, data privacy security, payroll tax, withholding procedures, establishment of bank or trust accounts to hold payroll deductions or contributions.

5.8 <u>Leave of Absence</u>. During leaves of absence approved by the Company meeting the requirements of Treasury Regulation Section 1.421-1(h)(2) under the Code, a Participant may continue participation in the Plan by making cash payments to the Company on his or her normal Payday equal to the Participant's authorized payroll deduction.

ARTICLE VI. GRANT AND EXERCISE OF RIGHTS

- 6.1 <u>Grant of Rights</u>. On the Enrollment Date of each Offering Period, each Eligible Employee participating in such Offering Period shall be granted a right to purchase the maximum number of Shares specified under Section, subject to the limits in Section, and shall have the right to buy, on each Purchase Date during such Offering Period (at the applicable Purchase Price), such number of whole Shares as is determined by dividing (a) such Participant's payroll deductions accumulated prior to such Purchase Date and retained in the Participant's account as of the Purchase Date, by (b) the applicable Purchase Price (rounded down to the nearest Share). The right shall expire on the earliest of: (x) the last Purchase Date of the Offering Period, (y) the last day of the Offering Period, and (z) the date on which the Participant withdraws in accordance with Section 7.1 or Section 7.3.
- 6.2 Exercise of Rights. On each Purchase Date, each Participant's accumulated payroll deductions and any other additional payments specifically provided for in the applicable Offering Document will be applied to the purchase of whole Shares, up to the maximum number of Shares permitted pursuant to the terms of the Plan and the applicable Offering Document, at the Purchase Price. No fractional Shares shall be issued upon the exercise of rights granted under the Plan. Any cash in lieu of fractional Shares remaining after the purchase of whole Shares upon exercise of a purchase right will be credited to a Participant's account and carried forward and applied toward the purchase of whole Shares for the next following Offering Period. Shares issued pursuant to the Plan may be evidenced in such manner as the Administrator may determine and may be issued in certificated form or issued pursuant to book-entry procedures.
- 6.3 Pro Rata Allocation of Shares. If the Administrator determines that, on a given Purchase Date, the number of Shares with respect to which rights are to be exercised may exceed (a) the number of Shares that were available for issuance under the Plan on the Enrollment Date of the applicable Offering Period, or (b) the number of Shares available for issuance under the Plan on such Purchase Date, the Administrator may in its sole discretion provide that the Company shall make a pro rata allocation of the Shares available for purchase on such Enrollment Date or Purchase Date, as applicable, in as uniform a manner as shall be practicable and as it shall determine in its sole discretion to be equitable among all Participants for whom rights to purchase Shares are to be exercised pursuant to this Article VI on such Purchase Date, and shall either (i) continue all Offering Periods then in effect, or (ii) terminate any or all Offering Periods then in effect pursuant to Article IX. The Company may make pro rata allocation of the Shares available on the Enrollment Date of any applicable Offering Period pursuant to the preceding sentence, notwithstanding any authorization of additional Shares for issuance under the Plan by the Company's stockholders subsequent to such Enrollment Date. The balance of the amount credited to the account of each Participant that has not been applied to the purchase of Shares shall be paid to such Participant in one lump sum in cash as soon as reasonably practicable after the Purchase Date or such earlier date as determined by the Administrator.

- 6.4 Withholding. At the time a Participant's rights under the Plan are exercised, in whole or in part, or at the time some or all of the Shares issued under the Plan is disposed of, the Participant must make adequate provision for the Company's federal, state, or other tax withholding obligations, if any, that arise upon the exercise of the right or the disposition of the Shares. At any time, the Company may, but shall not be obligated to, withhold from the Participant's Compensation or Shares received pursuant to the Plan the amount necessary for the Company to meet applicable withholding obligations, including any withholding required to make available to the Company any tax deductions or benefits attributable to sale or early disposition of Shares by the Participant.
- 6.5 Conditions to Issuance of 6.6 Shares. The Company shall not be required to issue or deliver any certificate or certificates for, or make any book entries evidencing, Shares purchased upon the exercise of rights under the Plan prior to fulfillment of all of the following conditions: (a) the admission of such Shares to listing on all stock exchanges, if any, on which the Shares are then listed; (b) the completion of any registration or other qualification of such Shares under any state or federal law or under the rulings or regulations of the Securities and Exchange Commission or any other governmental regulatory body, that the Administrator shall, in its absolute discretion, deem necessary or advisable; (c) the obtaining of any approval or other clearance from any state or federal governmental agency that the Administrator shall, in its absolute discretion, determine to be necessary or advisable; (d) the payment to the Company of all amounts that it is required to withhold under federal, state or local law upon exercise of the rights, if any; and (e) the lapse of such reasonable period of time following the exercise of the rights as the Administrator may from time to time establish for reasons of administrative convenience.

ARTICLE VII. WITHDRAWAL; CESSATION OF ELIGIBILITY

- 7.1 Withdrawal. A Participant may withdraw all but not less than all of the payroll deductions credited to his or her account and not yet used to exercise his or her rights under the Plan at any time by giving written notice to the Company in a form acceptable to the Company no later than one week prior to the end of the Offering Period (or such shorter or longer period as may be specified by the Administrator in the applicable Offering Document). All of the Participant's payroll deductions credited to his or her account during an Offering Period shall be paid to such Participant as soon as reasonably practicable after receipt of notice of withdrawal and such Participant's rights for the Offering Period shall be automatically terminated, and no further payroll deductions for the purchase of Shares shall be made for such Offering Period. If a Participant withdraws from an Offering Period, payroll deductions shall not resume at the beginning of the next Offering Period unless the Participant timely delivers to the Company a new subscription agreement.
- 7.2 <u>Future Participation</u>. A Participant's withdrawal from an Offering Period shall not have any effect upon his or her eligibility to participate in any similar plan that may hereafter be adopted by the Company or a Designated Subsidiary or in subsequent Offering Periods that commence after the termination of the Offering Period from which the Participant withdraws.
- 7.3 Cessation of Eligibility. Upon a Participant's ceasing to be an Eligible Employee for any reason, he or she shall be deemed to have elected to withdraw from the Plan pursuant to this Article VII and the payroll deductions credited to such Participant's account during the Offering Period shall be paid to such Participant or, in the case of his or her death, to the person or persons entitled thereto under Section, as soon as reasonably practicable, and such Participant's rights for the Offering Period shall be automatically terminated. If a Participant transfers employment from the Company or any Designated Subsidiary participating in the Section 423 Component, such transfer shall not be treated as a termination of employment, but the Participant shall immediately cease to participate in the Section 423 Component; however, any

contributions made for the Offering Period in which such transfer occurs shall be transferred to the Non-Section 423 Component, and such Participant shall immediately join the then-current Offering under the Non-Section 423 Component upon the same terms and conditions in effect for the Participant's participation in the Section 423 Component, except for such modifications otherwise applicable for Participants in such Offering. A Participant who transfers employment from any Designated Subsidiary participating in the Non-Section 423 Component to the Company or any Designated Subsidiary participating in the Section 423 Component shall not be treated as terminating the Participant's employment and shall remain a Participant in the Non-Section 423 Component until the earlier of (i) the end of the current Offering Period under the Non-Section 423 Component or (ii) the Enrollment Date of the first Offering Period in which the Participant is eligible to participate following such transfer. Notwithstanding the foregoing, the Administrator may establish different rules to govern transfers of employment between entities participating in the Section 423 Component and the Non-Section 423 Component, consistent with the applicable requirements of Section 423 of the Code.

ARTICLE VIII. ADJUSTMENTS UPON CHANGES IN SHARES

- 8.1 Changes in Capitalization. Subject to Section, in the event that the Administrator determines that any dividend or other distribution (whether in the form of cash, Shares, other securities, or other property), change in control, reorganization, merger, amalgamation, consolidation, combination, repurchase, redemption, recapitalization, liquidation, dissolution, or sale, transfer, exchange or other disposition of all or substantially all of the assets of the Company, or sale or exchange of Shares or other securities of the Company, issuance of warrants or other rights to purchase Shares or other securities of the Company, or other similar corporate transaction or event, as determined by the Administrator, affects the Shares such that an adjustment is determined by the Administrator to be appropriate in order to prevent dilution or enlargement of the benefits or potential benefits intended by the Company to be made available under the Plan or with respect to any outstanding purchase rights under the Plan, the Administrator shall make equitable adjustments, if any, to reflect such change with respect to (a) the aggregate number and type of Shares (or other securities or property) that may be issued under the Plan (including, but not limited to, adjustments of the limitations in Section and the limitations established in each Offering Document pursuant to Section on the maximum number of Shares that may be purchased); (b) the class(es) and number of Shares and price per Share subject to outstanding rights; and (c) the Purchase Price with respect to any outstanding rights.
- 8.2 Other Adjustments. Subject to Section, in the event of any transaction or event described in Section or any unusual or nonrecurring transactions or events affecting the Company, any affiliate of the Company, or the financial statements of the Company or any affiliate (including without limitation, any change in control), or of changes in Applicable Law or accounting principles, the Administrator, in its discretion, and on such terms and conditions as it deems appropriate, is hereby authorized to take any one or more of the following actions whenever the Administrator determines that such action is appropriate in order to prevent the dilution or enlargement of the benefits or potential benefits intended to be made available under the Plan or with respect to any right under the Plan, to facilitate such transactions or events or to give effect to such changes in laws, regulations or principles:
- (a) To provide for either (i) termination of any outstanding right in exchange for an amount of cash, if any, equal to the amount that would have been obtained upon the exercise of such right had such right been currently exercisable or (ii) the replacement of such outstanding right with other rights or property selected by the Administrator in its sole discretion;
- (b) To provide that the outstanding rights under the Plan shall be assumed by the successor or survivor corporation, or a Parent or Subsidiary thereof, or shall be substituted for by similar rights covering the stock of the successor or survivor corporation, or a Parent or Subsidiary thereof, with appropriate adjustments as to the number and kind of shares and prices;

- (c) To make adjustments in the number and type of Shares (or other securities or property) subject to outstanding rights under the Plan and/or in the terms and conditions of outstanding rights and rights that may be granted in the future;
- (d) To provide that Participants' accumulated payroll deductions may be used to purchase Shares prior to the next occurring Purchase Date on such date as the Administrator determines in its sole discretion and the Participants' rights under the ongoing Offering Period(s) shall be terminated; and
 - (e) To provide that all outstanding rights shall terminate without being exercised.
- 8.3 No Adjustment Under Certain Circumstances. Unless determined otherwise by the Administrator, no adjustment or action described in this Article VIII or in any other provision of the Plan shall be authorized to the extent that such adjustment or action would cause the Section 423 Component of the Plan to fail to satisfy the requirements of Section 423 of the Code.
- 8.4 No Other Rights. Except as expressly provided in the Plan, no Participant shall have any rights by reason of any subdivision or consolidation of shares of stock of any class, the payment of any dividend, any increase or decrease in the number of shares of stock of any class or any dissolution, liquidation, merger, or consolidation of the Company or any other corporation. Except as expressly provided in the Plan or pursuant to action of the Administrator under the Plan, no issuance by the Company of shares of stock of any class, or securities convertible into shares of stock of any class, shall affect, and no adjustment by reason thereof shall be made with respect to, the number of Shares subject to outstanding rights under the Plan or the Purchase Price with respect to any outstanding rights.

ARTICLE IX. AMENDMENT. MODIFICATION AND TERMINATION

- 9.1 <u>Amendment, Modification and Termination</u>. The Administrator may amend, suspend or terminate the Plan at any time and from time to time; <u>provided, however</u>, that approval of the Company's stockholders shall be required to amend the Plan to: (a) increase the aggregate number, or change the type, of shares that may be sold pursuant to rights under the Plan under Section (other than an adjustment as provided by Article VIII) or (b) change the corporations or classes of corporations whose employees may be granted rights under the Plan.
- 9.2 Certain Changes to Plan. Without stockholder consent and without regard to whether any Participant rights may be considered to have been adversely affected (and, with respect to the Section 423 Component of the Plan, after taking into account Section 423 of the Code), the Administrator shall be entitled to change or terminate the Offering Periods, add or revise Offering Period share limits, limit the frequency and/or number of changes in the amount withheld from Compensation during an Offering Period, establish the exchange ratio applicable to amounts withheld in a currency other than U.S. dollars, permit payroll withholding in excess of the amount designated by a Participant in order to adjust for delays or mistakes in the Company's s processing of payroll withholding elections, establish reasonable waiting and adjustment periods and/or accounting and crediting procedures to ensure that amounts applied toward the purchase of Shares for each Participant properly correspond with amounts withheld from the Participant's Compensation, and establish such other limitations or procedures as the Administrator determines in its sole discretion to be advisable that are consistent with the Plan.

- 9.3 Actions In the Event of Unfavorable Financial Accounting Consequences. In the event the Administrator determines that the ongoing operation of the Plan may result in unfavorable financial accounting consequences, the Administrator may, in its discretion and, to the extent necessary or desirable, modify or amend the Plan to reduce or eliminate such accounting consequence including, but not limited to:
 - (a) altering the Purchase Price for any Offering Period including an Offering Period underway at the time of the change in Purchase Price;
- (b) shortening any Offering Period so that the Offering Period ends on a new Purchase Date, including an Offering Period underway at the time of the Administrator action; and
 - (c) allocating Shares.

Such modifications or amendments shall not require stockholder approval or the consent of any Participant.

9.4 <u>Payments Upon Termination of Plan</u>. Upon termination of the Plan, the balance in each Participant's Plan account shall be refunded as soon as practicable after such termination, without any interest thereon, or the Offering Period may be shortened so that the purchase of Shares occurs prior to the termination of the Plan.

ARTICLE X. TERM OF PLAN

The Plan shall become effective on the Effective Date. The effectiveness of the Section 423 Component of the Plan shall be subject to approval of the Plan by the Company's stockholders within twelve months following the date the Plan is first approved by the Board. No right may be granted under the Section 423 Component of the Plan prior to such stockholder approval. The Plan shall remain in effect until terminated under Section 9.1. No rights may be granted under the Plan during any period of suspension of the Plan or after termination of the Plan.

ARTICLE XI. ADMINISTRATION

- 11.1 <u>Administrator</u>. Unless otherwise determined by the Board, the Administrator of the Plan shall be the Compensation Committee of the Board (or another committee or a subcommittee of the Board to which the Board delegates administration of the Plan). The Board may at any time vest in the Board any authority or duties for administration of the Plan. The Administrator may delegate administrative tasks under the Plan to the services of an Agent or Employees to assist in the administration of the Plan, including establishing and maintaining an individual securities account under the Plan for each Participant.
- 11.2 <u>Authority of Administrator</u>. The Administrator shall have the power, subject to, and within the limitations of, the express provisions of the Plan:
- (a) To determine when and how rights to purchase Shares shall be granted and the provisions of each offering of such rights (which need not be identical).
- (b) To designate from time to time which Subsidiaries of the Company shall be Designated Subsidiaries, which designation may be made without the approval of the stockholders of the Company.

- (c) To impose a mandatory holding period pursuant to which Employees may not dispose of or transfer Shares purchased under the Plan for a period of time determined by the Administrator in its discretion.
- (d) To construe and interpret the Plan and rights granted under it, and to establish, amend and revoke rules and regulations for its administration. The Administrator, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan, in a manner and to the extent it shall deem necessary or expedient to make the Plan fully effective.
 - (e) To amend, suspend or terminate the Plan as provided in Article IX.
- (f) Generally, to exercise such powers and to perform such acts as the Administrator deems necessary or expedient to promote the best interests of the Company and its Subsidiaries and to carry out the intent that the Plan be treated as an "employee stock purchase plan" within the meaning of Section 423 of the Code for the Section 423 Component.
- (g) The Administrator may adopt sub-plans applicable to particular Designated Subsidiaries or locations, which sub-plans may be designed to be outside the scope of Section 423 of the Code. The rules of such sub-plans may take precedence over other provisions of this Plan, with the exception of Section hereof, but unless otherwise superseded by the terms of such sub-plan, the provisions of this Plan shall govern the operation of such sub-plan.
- 11.3 <u>Decisions Binding</u>. The Administrator's interpretation of the Plan, any rights granted pursuant to the Plan, any subscription agreement and all decisions and determinations by the Administrator with respect to the Plan are final, binding, and conclusive on all parties.

ARTICLE XII. MISCELLANEOUS

- 12.1 <u>Restriction upon Assignment</u>. A right granted under the Plan shall not be transferable other than by will or the applicable laws of descent and distribution, and is exercisable during the Participant's lifetime only by the Participant. Except as provided in Section hereof, a right under the Plan may not be exercised to any extent except by the Participant. The Company shall not recognize and shall be under no duty to recognize any assignment or alienation of the Participant's interest in the Plan, the Participant's rights under the Plan or any rights thereunder.
- 12.2 Rights as a Stockholder. With respect to Shares subject to a right granted under the Plan, a Participant shall not be deemed to be a stockholder of the Company, and the Participant shall not have any of the rights or privileges of a stockholder, until such Shares have been issued to the Participant or his or her nominee following exercise of the Participant's rights under the Plan. No adjustments shall be made for dividends (ordinary or extraordinary, whether in cash securities, or other property) or distribution or other rights for which the record date occurs prior to the date of such issuance, except as otherwise expressly provided herein or as determined by the Administrator.
 - 12.3 Interest. No interest shall accrue on the payroll deductions or contributions of a Participant under the Plan.

12.4 Designation of Beneficiary.

- (a) A Participant may, in the manner determined by the Administrator, file a written designation of a beneficiary who is to receive any Shares and/or cash, if any, from the Participant's account under the Plan in the event of such Participant's death subsequent to a Purchase Date on which the Participant's rights are exercised but prior to delivery to such Participant of such Shares and cash. In addition, a Participant may file a written designation of a beneficiary who is to receive any cash from the Participant's account under the Plan in the event of such Participant's death prior to exercise of the Participant's rights under the Plan. If the Participant is married and resides in a community property state, a designation of a person other than the Participant's spouse as his or her beneficiary shall not be effective without the prior written consent of the Participant's spouse.
- (b) Such designation of beneficiary may be changed by the Participant at any time by written notice to the Company. In the event of the death of a Participant and in the absence of a beneficiary validly designated under the Plan who is living at the time of such Participant's death, the Company shall deliver such Shares and/or cash to the executor or administrator of the estate of the Participant, or if no such executor or administrator has been appointed (to the knowledge of the Company), the Company, in its discretion, may deliver such Shares and/or cash to the spouse or to any one or more dependents or relatives of the Participant, or if no spouse, dependent or relative is known to the Company, then to such other person as the Company may designate.
- 12.5 <u>Notices</u>. All notices or other communications by a Participant to the Company under or in connection with the Plan shall be deemed to have been duly given when received in the form specified by the Company at the location, or by the person, designated by the Company for the receipt thereof.
- 12.6 Equal Rights and Privileges. Subject to Section, all Eligible Employees will have equal rights and privileges under the Section 423 Component so that the Section 423 Component of this Plan qualifies as an "employee stock purchase plan" within the meaning of Section 423 of the Code. Subject to Section, any provision of the Section 423 Component that is inconsistent with Section 423 of the Code will, without further act or amendment by the Company, the Board or the Administrator, be reformed to comply with the equal rights and privileges requirement of Section 423 of the Code. Eligible Employees participating in the Non-Section 423 Component need not have the same rights and privileges as other Eligible Employees participating in the Non-Section 423 Component or as Eligible Employees participating in the Section 423 Component.
- 12.7 <u>Use of Funds</u>. All payroll deductions received or held by the Company under the Plan may be used by the Company for any corporate purpose, and the Company shall not be obligated to segregate such payroll deductions.
- 12.8 No Employment Rights. Nothing in the Plan shall be construed to give any person (including any Eligible Employee or Participant) the right to remain in the employ of the Company or any Parent or Subsidiary or affect the right of the Company or any Parent or Subsidiary to terminate the employment of any person (including any Eligible Employee or Participant) at any time, with or without cause.
- 12.9 Notice of Disposition of Shares. Each Participant shall give prompt notice to the Company of any disposition or other transfer of any Shares purchased upon exercise of a right under the Section 423 Component of the Plan if such disposition or transfer is made: (a) within two years from the Enrollment Date of the Offering Period in which the Shares were purchased or (b) within one year after the Purchase Date on which such Shares were purchased. Such notice shall specify the date of such disposition or other transfer and the amount realized, in cash, other property, assumption of indebtedness or other consideration, by the Participant in such disposition or other transfer.

12.10 <u>Governing Law</u>. The Plan and any agreements hereunder shall be administered, interpreted and enforced in accordance with the laws of the State of Delaware, disregarding any state's choice of law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

12.11 <u>Electronic Forms</u>. To the extent permitted by Applicable Law and in the discretion of the Administrator, an Eligible Employee may submit any form or notice as set forth herein by means of an electronic form approved by the Administrator. Before the commencement of an Offering Period, the Administrator shall prescribe the time limits within which any such electronic form shall be submitted to the Administrator with respect to such Offering Period in order to be a valid election.

* * * * *

KENSINGTON CAPITAL ACQUISITION CORP. II AND WALLBOX ANNOUNCE CLOSING OF BUSINESS COMBINATION; WALLBOX TO TRADE ON NYSE UNDER TICKER "WBX" BEGINNING ON OCTOBER 4

WESTBURY, N.Y. and BARCELONA, Spain, October 1, 2021 - Kensington Capital Acquisition Corp. II (NYSE: KCAC) ("Kensington") and Wall Box Chargers, S.L. ("Wallbox"), a leader in electric vehicle charging and energy management solutions, announced today the completion of its business combination. The business combination was approved by Kensington stockholders on September 30, 2021, by a quorum of 76.5% of the outstanding shareholders and received approval from 94.9% of those votes cast. Beginning on October 4, 2021, the Class A ordinary shares of Wallbox N.V., a newly formed holding company, will trade on the NYSE under the ticker symbol "WBX" and its warrants will trade on the NYSE under the ticker symbol "WBX.WS."

"The completion of our business combination with Kensington greatly accelerates our strategy globally, including the construction of our manufacturing facility in Texas and working capital needed to support sales growth in over 70 countries," said Enric Asunción, CEO and co-founder of Wallbox. "We are pleased to have partnered with Kensington, who out-performed in a challenging SPAC market, and shares our vision of a more energy-sustainable automotive future."

Justin Mirro, Chairman and CEO of Kensington, added, "we are proud to complete our business combination with Wallbox and look forward to the future success of Enric and his team of over 700 associates around the world. To partner with such a visionary company, that is already impacting how people manage their energy consumption today, continues our tradition at Kensington to support those companies that make the world greener, safer and better for the next generation."

"We appreciate the continued support of our investors throughout this process, including Janus Henderson Investors and Kepos Capital. It is great to work with these world-class investment managers that support the Kensington investment thesis and share our vision for the future," added Dan Huber, CFO of Kensington.

The transaction will result in gross proceeds of approximately \$250 million to Wallbox to fund the company as it expands its manufacturing capacity and global sales presence. Hughes Hubbard & Reed LLP served as legal advisor to Kensington and Latham and Watkins LLP served as legal advisor to Wallbox. Houthoff and Loyens & Loeff advised Kensington and Wallbox, respectively, on matters of Dutch law and Cuatrecasas, Gonçalves Pereira, S.L.P. advised Kensington on matters of Spanish law. UBS Securities LLC, Stifel Nicolaus & Company, Incorporated and Robert W. Baird & Co. Incorporated served as financial advisors to Kensington and Barclays Capital Inc. and Drake Star Partners served as financial advisors to Wallbox. UBS Securities LLC and Barclays Capital Inc. served as joint placement agents on the PIPE offering.

About Wallbox

Wallbox is a global technology company, dedicated to changing the way the world uses energy. Wallbox creates advanced electric vehicle charging and energy management systems that redefine users' relationship to the grid. Wallbox goes beyond electric vehicle charging to give users the power to control their consumption, save money, and live more sustainably. Wallbox offers a complete portfolio of charging and energy management solutions for residential, semi-public and public use in more than 80 countries.

Founded in 2015 and headquartered in Barcelona, the company now employs over 700 people in its offices in Europe, Asia, and the Americas.

For additional information, please visit www.wallbox.com.

About Kensington

Kensington Capital Acquisition Corp. II, which traded on the NYSE under the ticker symbol KCAC, was a special purpose acquisition company formed for the purpose of effecting a merger, stock purchase or similar business combination with a business in the automotive and automotive-related sector. The company was sponsored by Kensington Capital Partners ("KCP") and the management team included Justin Mirro, Bob Remenar, Simon Boag and Dan Huber. The company was also supported by a board of independent directors including Tom LaSorda, Nicole Nason, Anders Pettersson, Mitch Quain, Don Runkle and Matt Simoncini. The Kensington team has completed over 70 automotive transactions and has over 300 years of combined experience leading some of the largest automotive companies in the world.

For additional information, please visit www.autospac.com.

Caution About Forward-Looking Statements

The information in this press release includes "forward-looking statements" within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934, as amended. All statements, other than statements of present or historical fact included in this press release, regarding Kensington's business combination with Wallbox, the development and performance of Wallbox's products (including the timeframe for development of such products), the benefits of the transaction and the combined company's future financial performance, as well as the combined company's strategy, future operations, estimated financial position, estimated revenues and losses, projected costs, prospects, plans and objectives of management are forward-looking statements. When used in this press release, the words "are designed to," "could," "will," "may," "believe," "anticipate," "intend," "estimate," "expect," "project," the negative of such terms and other similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain such identifying words. These forward-looking statements are based on management's current expectations and assumptions about future events and are based on currently available information as to the outcome and timing of future events. Except as otherwise required by applicable law, Wallbox disclaims any duty to update any forward-looking statements, all of which are expressly qualified by the statements in this section, to reflect events or circumstances after the date hereof. Wallbox cautions you that these forwardlooking statements are subject to numerous risks and uncertainties, most of which are difficult to predict and many of which are beyond the control of either Kensington or Wallbox. In addition, Wallbox cautions you that the forward-looking statements contained herein are subject to the following uncertainties and risk factors that could affect the combined company's future performance and cause results to differ from the forward-looking statements herein: Wallbox's ability to realize the anticipated benefits of the business combination, which may be affected by, among other things, competition and the ability of Wallbox to grow and manage growth profitably following the business combination; risks relating to the outcome and timing of the Company's development of its charging and energy management technology and related manufacturing processes; intense competition in the electric vehicle charging space; risks related to health pandemics, including the COVID-19 pandemic; the possibility that Wallbox may be adversely affected by other economic, business, and/or competitive factors; the possibility that the expected timeframe for, and other expectations regarding the development and performance of, Wallbox products will differ from current assumptions; the outcome of any legal proceedings that may be instituted against the combined company or others following the consummation of the business combination; the ability to meet stock exchange listing standards following the consummation of the business combination; the risk that the business combination disrupts current plans and operations of the combined company as a result of the consummation of the business combination; costs related to the business combination; and changes in applicable laws or regulations. Should one or more of the risks or uncertainties described in this press release, or should underlying assumptions prove incorrect, actual results and plans could different materially from those expressed in any forward-looking statements. Additional information concerning these and other factors that may impact the operations and projections discussed herein can be found in Kensington's periodic filings with the SEC, and the proxy statement/prospectus of Wallbox B.V. in the registration statement on Form F-4 filed with the SEC. Kensington's and Wallbox B.V.'s SEC filings are available publicly on the SEC's website at www.sec.gov.

Contacts:

For Wallbox

Investors

ICR, Inc.

 $\underline{investors@wallbox.com}$

Media

ICR, Inc.

WallboxPR@icrinc.com

For Kensington

Dan Huber dan@kensington-cap.com 703-674-6514